Equality Commission FOR NORTHERN IRELAND

Disability Code of Practice Employment and Occupation

The Disability Discrimination Act 1995

(as modified by Schedule 8 thereof for application in Northern Ireland)

NIA 167/03

DDA Code of Practice – Employment & Occupation

ALERT – A NOTE ABOUT UPDATES APRIL 2013

This Code of Practice was initially issued by the Equality Commission in June 2005. Since then a major case law decision by the UK's highest court has changed our understanding of some important aspects of the law. There have also been some changes made to the Disability Discrimination Act ("DDA") itself by later statutory amendments. In summary, the main developments were:

Case law decision:

 In 2008, the Judicial Committee of the House of Lords heard the case of Malcolm –v- Lewisham Borough Council. In it, the Law Lords made a ruling that significantly changed our understanding of the concept of disability-related discrimination.

Changes made to the statutory definition of "disability" by:

- Disability Discrimination (NI) Order 2006
- Autism (NI) Act 2011

Purpose of this note

The purpose of this note is to offer some guidance for employers and other readers on how to read and apply the Code now in the light of the said changes. This guidance is set out in the tables on pages 2 to 6.

Reasonable adjustments

A significant feature of the DDA is the *duty to make reasonable adjustments*. The advice provided in the Code of Practice has not been significantly affected by the legal developments that are noted here.



Disability Code of Practice

Employment and Occupation

This document is available in other formats upon request

Laid in draft before the Northern Ireland Assembly by the Office of the First Minister and deputy First Minister under Article 54A(4) of the Disability Discrimination Act 1995 (as modified by Schedule 8 thereof for application in Northern Ireland).

Date 16th May 2005

NIA 167/03

Published by the Equality Commission for Northern Ireland in 2005.

This Code of Practice was developed in partnership with the Disability Rights Commission, Great Britain. The Equality Commission gratefully acknowledges the substantial contribution made by the Disability Rights Commission to this publication.

The Code of Practice and information about the Disability Discrimination Act are available from the Equality Commission in other formats, including large print, Braille and audiotape. The Code can also be downloaded from the Equality Commission's website.

The Equality Commission for Northern Ireland Equality House 7-9 Shaftesbury Square Belfast BT2 7DP

Telephone:028 90 500 600Textphone:028 90 500 589Fax:028 90 248 687E-mail:information@equalityni.orgWebsite:www.equalityni.orgTypetalk

ISBN 1-903941-76-8

The Code of Practice shall come into effect on such day as the Office of the First Minister and Deputy First Minister may by order appoint. Please contact the Equality Commission for further details.

The Code of Practice applies to Northern Ireland.

This Code of Practice deals with the duties under Part 2 (referred to in the legislation as 'Part II') of the Disability Discrimination Act 1995, which is based on the principle that disabled people should not be discriminated against in employment or when seeking employment. It also protects disabled people engaged in a range of occupations outside employment. Employers must comply with the duties set out in Part 2 of the Act, as must others to whom those duties apply.

Except for the armed forces, these duties now apply to all employers no matter how many (or how few) people they employ. The law changed in this regard on 1 October 2004. Other changes in the law which took effect on that date, including the extension of the duties to some people or bodies who are not employers, are summarised in Appendix A to this Code.

This Code replaces the original Code, which was issued in 1996. It takes account of changes to the employment provisions of the Disability Discrimination Act 1995 that reflect the requirements of the European Union's Employment Framework Directive (Council Directive 2000/78/EC of 27 November 2000), which established an anti-discrimination principle of equal treatment in relation to disability (as well as a number of other grounds) across all Member States.

Foreword

Codes of Practice and associated guidance published by the Equality Commission are important in explaining the complexities of the law, and in helping employers and others comply with it. This Code is one of two that we have prepared to interpret the requirements of Part 2 of the Disability Discrimination Act 1995. This Code of Practice explains how disabled people are protected by that Act if they are in employment, if they are seeking employment, or if they are involved in a range of occupations. The other Code relates to discrimination by trade organisations and qualifications bodies.

Although discrimination against disabled people in employment has been outlawed since the Disability Discrimination Act came into force, the law has continued to evolve. Not only have the original provisions of the Act been subject to judicial interpretation, but major changes have been made to the legislation itself. From 1 October 2004 the Act's coverage was extended significantly to take account of the requirements of an EU Council Directive (2000/78/EC), which establishes a general framework for equal treatment, in employment and occupation.

This Code has been prepared with particular regard to the needs of legal advisers when advising their clients, and to assisting courts and industrial tribunals when interpreting the new concepts. It is necessarily comprehensive and detailed and therefore the Equality Commission has also produced a range of other information to help disabled people and employers alike to understand their rights and responsibilities. How to find this information is explained in Appendix C to this Code.

This Code, and associated guidance, is intended to be a resource for employers and others who seek to understand

their duties and responsibilities under the Act and for disabled people who need to know and understand their rights under the disability legislation. The Commission hopes these publications will be widely used and helpful.

Dame Joan Harbison

Chief Commissioner, Equality Commission

ALERT – A NOTE ABOUT UPDATES APRIL 2013

This Code of Practice was initially issued by the Equality Commission in June 2005. Since then a major case law decision by the UK's highest court has changed our understanding of some important aspects of the law. There have also been some changes made to the Disability Discrimination Act ("DDA") itself by later statutory amendments. In summary, the main developments were:

Case law decision:

 In 2008, the Judicial Committee of the House of Lords heard the case of Malcolm –v- Lewisham Borough Council. In it, the Law Lords made a ruling that significantly changed our understanding of the concept of disability-related discrimination.

Changes made to the statutory definition of "disability" by:

- Disability Discrimination (NI) Order 2006
- Autism (NI) Act 2011

Purpose of this note

The purpose of this note is to offer some guidance for employers and other readers on how to read and apply the Code now in the light of the said changes. This guidance is set out in the tables on pages 2 to 6.

Reasonable adjustments

A significant feature of the DDA is the *duty to make reasonable adjustments*. The advice provided in the Code of Practice has not been significantly affected by the legal developments that are noted here.

Code of Practice – Employment & Occupation: UPDATE

The following paragraphs of the current Code of Practice should now be read as follows.

	Paragraph	Notes	
1	Paragraph 2.50 Page 37 (this is linked to Chapter 13 of the Code which deals with the topic In more detail)	Replace original text with the following: "The recommended procedures to follow when dealing with disputes between employers and their employees are described in the Labour Relations Agency's Code of Practice on Disciplinary and Grievance Procedures [April 2011]. Employers and employees alike should have regard to this."	
2	Paragraph 4.14 Page 55	In the first example given on this page, delete the final sentence.	
3	Paragraph 4.19 Page 57	Delete the entire paragraph .	
4	Paragraph 4.22 Page 61	In the first example given on this page, delete the final sentence and insert this text in its place: "(The disabled person would be likely to have a good claim of disability discrimination in relation to the employer's failure to make a reasonable adjustment for her (i.e. the failure to allow her to use the adapted keyboard))."	
5	Paragraphs 4.27 to 4.32 Pages 62 to 65 What is disability- related discrimination	 (a) Keep paragraph 4.27 unchanged. (b) Delete paragraphs 4.28 to 4.31 and the associated examples, and replace them with the following text instead: "4.28 For disability-related discrimination to occur, it is necessary that the employer should know that the disabled person is disabled. 4.29 The comparators to be used when assessing whether direct discrimination or disability-related discrimination has occurred <u>are the same</u>." (c) Insert a single example after paragraph 4.29. It states- "If a disabled person is dismissed for having been off work on 6 months disability-related sick leave, then his/her comparator is any other person who has been off work for the same (or a longer) period but for reasons that are not related to a disability. If that person was <u>not</u> dismissed, then that will raise an inference of disability-related discrimination. It will also raise an inference of direct discrimination." (d) Keep paragraph 4.32 unchanged. 	

	Paragraph	Notes
6	Paragraph 4.36 Page 67	Replace original text with the following:
	Victimisation	"A person does not need to be a disabled person himself (or herself), to be able to submit a valid claim of victimisation."
7	Paragraph 4.37 Pages 67 and 68	The example here explains the relationship between the various forms of disability discrimination. It is still mostly valid, but it needs some amendment. It should be deleted and replaced with the following text-
	The Example	"A person with arthritis applies for a secretarial job in a local business. There is a question on the application form about disability. The applicant has arthritis and states this on the form. The arthritis does not affect the applicant's typing ability. This fact is also stated by the applicant on the form. The employer rejects the application because he, nevertheless, wrongly assumes that the disabled person will not be able to carry out the job due to the arthritis. This is direct discrimination . It cannot be justified.
		In the situation described above, let's change the scenario and instead say that the arthritis does affect the applicant's typing ability and that she discloses this information in the application form. The applicant is called for an interview and is told that a typing test forms part of the selection process. The applicant requests the use of an adapted keyboard in order to take the test, but this is not provided on the day of the interview, and the applicant fails the test as a result. As a consequence of failing the test, the applicant is turned down for the job.
		In such circumstances the employer has a duty to make reasonable adjustments to its selection arrangements. Depending on the circumstances, it may be a reasonable adjustment for the employer to provide the adapted keyboard or allow the applicant to use her own keyboard in order that she is not placed at a substantial disadvantage by the test. If this is the case, then the employer will be unlawfully discriminating against the applicant by failing to make the adjustment .
		The disabled applicant was rejected because she failed the test. She failed the test for a reason related to her arthritis. Thus, she failed the test for a reason related to her disability. Whether she was subjected to disability-related discrimination , however, depends on how the employer treated other non-disabled persons who also failed the test. If these people (or one or two of them) were appointed despite having failed the test, then it is likely that the disabled person was subjected to disability-related discrimination. On the other hand, if those other people were rejected too then there was no disability-related discrimination.
		If evidence of disability-related discrimination does exist, as described above, it will also probably be sufficient to prove a claim of direct discrimination. Thus, these two forms of discrimination are closely linked.
		As a result of this treatment, the applicant makes a claim against the employer under Part 2 of the Act. Some time later, however, the same employer advertises a further secretarial vacancy. The applicant applies again, but the employer rejects the application because the applicant has previously made a claim under the Act. This is victimisation ."

	Paragraph	Notes
8	Paragraph 4.43 Pages 72 and 73	This section explains the burden of proof in relation to the various forms of disability discrimination. The 3 rd bullet-point, which deals with disability-related discrimination , should be deleted and replaced with the following text: "To prove a complaint of disability-related discrimination, the employee must prove facts from which it could be inferred in the absence of an adequate explanation that, for a reason relating to his or her disability, s/he has been treated less favourably than an appropriate comparator has been, or would be treated. If this inference could be drawn, it would simultaneously raise an inference of direct discrimination too."
9	Paragraph 5.12 Page 79	In the first sentence , delete the words "and 4.31".
11	Paragraph 7.10 Page 117	Delete the first sentence of this paragraph and replace it with the following: "An employer is entitled to specify that applicants for a job must have certain qualifications. It is best practice and legally prudent to set criteria that are justified by the needs of the job."
12	Paragraph 7.31 and the examples Pages 130 to 131	 (a) Delete paragraph 7.31 and replace it with the following: "7.31 The Act does not prevent an employer from asking job applicants to attend for a medical examination. But, it is likely to be unlawful to impose the requirement on disabled persons only. It is best practice to impose such requirements only if they are justified by the needs of the job. When they are imposed, it is also best practice only to impose them on those applicants (disabled or not) who have been made conditional offers of employment. 7.31A A response to a medical questionnaire, the results of a medical examination or the opinion of a Medical Advisor should only be one of a number of factors that an employer should consider in reaching his or her final selection decision, and it should not, without justification, be treated in itself as determinative of the matter. 7.31B In particular, in the case of disabled applicants, the employer should also consider what reasonable adjustments, if any, may be required, and should make his selection decision following an assessment of how the applicant would perform in the job if these reasonable adjustments were actually made." (b) Delete examples 1, 2 and 3 on page 131.

	Paragraph	Notes
13	Paragraph 8.5 Page 134	Delete the final sentence of this paragraph.
14	Page 136	Delete the first example on this page.
15	Paragraph 8.8 Page 138	In the third example on this page, delete the end of the final sentence where it states: <i>"would be unlikely to be justified".</i>
		Replace those words with the following: "would probably be a failure to comply with the duty to make reasonable adjustments."
16	Paragraph 8.13 Page 140	Delete the second and third sentences of this paragraph; i.e. between the words: <i>"It is likely to be direct discriminationthe employer can show that it is justified".</i>
17	Paragraph 8.24 Page 149	Delete the second and third sentences of this paragraph; i.e. between the words: <i>"It is likely to be direct discriminationthe employer can show that it is justified".</i>
18	Paragraph 10.4 Page 178	Delete the last sentence of the example (i.e. <i>"This is likely to be unlawful"</i>) and replace it with the following:
		"If the employer has the same worry about other non-disabled employees but nevertheless still invites, or would invite, them to join the scheme, then his refusal to invite the disabled employee would be unlawful direct discrimination."
19	Paragraph 11.20 Pages 195 to 196	Delete the last sentence of this paragraph; i.e. between the words: "However, as noted at paragraphs 11.12disability-related discrimination (see paragraph 11.21)."
20	Paragraphs 12.6 to 12.12	These paragraphs make references to certain building regulations. The following changes should be noted:
	Pages 201 to 202	(a) The Building Regulations (NI) 2000 were revoked on 31 October 2012 and were replaced by the Building Regulations (NI) 2012 [SR 2012 No. 192].
		(b) Part R of the 2000 regulations was replaced by a closely similar set of provisions in Part R of the 2012 regulations .
		(c) Technical Booklet R (2000) was replaced by Technical Booklet R (2012) . The exemption described in paragraph 12.8 will apply to the latest version in the same way as it applied to previous versions.
		(d) The British Standard described in paragraph 12.11, numbered BS 8300:2001 , has been replaced by a newer version; numbered BS 8300:2009+A1:2010 . Both versions have the same title. The guidance given in paragraph 12.11 is otherwise unchanged.

	Paragraph	Notes
21	Chapter 13 Paragraphs 13.2 to 13.12 Pages 213 to 217	These sections discuss the statutory procedures for the informal resolution of disputes. It gives a correct description of the law as it was in 2005, but the legal duty to use internal dispute resolution procedures was significantly changed (but not completely reversed) by the <i>Employment Act (NI) 2011</i> . Readers would be best advised to consult the definitive source of guidance about the new procedures: i.e. the Labour Relations Agency's Code of Practice on Disciplinary and Grievance Procedures [April 2011].
22	Paragraph 13.12 Page 217	Delete this paragraph .
23	Paragraph 13.25 Page 221	There is a reference in this paragraph to the exemption from the anti- discrimination provisions of the DDA in relation to government funded "supported employment" schemes. Add a footnote to the first sentence: "In practice the exemption only applies to one particular scheme that is currently being funded by the government; i.e. the scheme known as <i>Workable (NI)</i> ."
24	Appendix B	In this section, make the following changes-
	Pages 227 to 231	<u>1. When is a person disabled? (page 227)</u> Delete the current paragraph and replace it with the following-
	Meaning of Disability	"A person who contracts cancer, Multiple Sclerosis of HIV Infection, no matter how insubstantial their symptoms currently are, is immediately deemed to be disabled. As to any other person with other impairments, he or she will be deemed to disabled if he/she has a physical or mental impairment, which has a substantial and long-term adverse effect of his or her ability to carry out normal day-to-day activities."
		2. Are all mental impairments covered? (pages 227-228) Delete the second and third sentences of this paragraph; i.e. between the words: "The Act says that a mental illness must be a clinically well-recognised illnessa respected body of medical opinion."
		3. What if the effects come and go? (page 228) In the final sentence, delete the words <i>"it is more probable than not that the effect will recur</i> " and replace with the words <i>"a recurrence could well happen</i> ".
		<u>4. List of capacities (page 229)</u> Add the following bullet-points to the list-
		taking part in normal social interaction;forming social relationships
		5. What about people who know their condition is going to get worse over time? (page 230) In this paragraph, delete the second sentence and in the third sentence , after the words "a progressive condition" add the words "(other than cancer, Multiple Sclerosis or HIV Infection)".

Contents

Page

1.	Introduction	9
	Purpose of Part 2 of the Act	9
	Purpose of the Code	9
	Status of the Code	10
	How to use the Code	11
	Examples in the Code	12
	References to 'employers' in the Code	13
	Other references in the Code	13
	Changes to the legislation	14
	Further information	14
	Public Authorities	15
2.	How can discrimination be avoided?	17
	Introduction	17
	Understanding the social dimension of disability	17
	Recognising the diverse nature of disability	18
	Avoiding making assumptions	18
	Seeking expert advice	19
	Planning ahead	20
	Implementing anti-discriminatory policies and	
	practices	21
	Auditing policies and procedures	27
	Monitoring and Review	30
	Ensuring good practice in recruitment	34
	Resolving disputes	36

3.	The Act's provisions on employment and		
	occupation - an overview	39	
	Introduction	39	
	Who has rights under the Act?	39	

Who has obligations under the Act?	41
What does the Act say about discrimination in rel	lation
to employment and occupation?	44
What else is unlawful under the Act's provisions of	on
employment and occupation?	46
Who is liable for unlawful acts?	46

What is discrimination and harassment?	49
Introduction	49
What does the Act mean by 'direct discrimination'?	50
Failure to make reasonable adjustments - relationsh	nip
to discrimination	62
What is disability-related discrimination?	62
What does the Act say about victimisation?	66
How do the different forms of discrimination compar-	е
in practice?	67
What does the Act say about harassment?	69
What does the Act say about statutory obligations?	70
What evidence is needed to prove that discrimination	า
or harassment has occurred?	71
	Introduction What does the Act mean by 'direct discrimination'? Failure to make reasonable adjustments - relationsh to discrimination What is disability-related discrimination? What does the Act say about victimisation? How do the different forms of discrimination compar- in practice? What does the Act say about harassment? What does the Act say about statutory obligations? What does the Act say about statutory obligations?

5.	What is the duty to make reasonable		
	adjustments?	75	
	Introduction	75	
	When does an employer's duty to make reasonable		
	adjustments arise?	75	
	Which disabled people does the duty protect?	76	
	What are 'provisions, criteria and practices'?	77	
	What is a 'physical feature'?	78	
	What disadvantages give rise to the duty?	79	
	What if the employer does not know that the person		
	is disabled?	79	

Does the duty to make reasonable adjustments apply		
in other situations related to employment and		
occupation? 83	3	
What adjustments might an employer have to make? 83	3	
When is it 'reasonable' for an employer to have to		
make adjustments? 91	I	
Can failure to make a reasonable adjustment ever be		
justified? 102		

6.	Justification	103
	Introduction	103
	When does the Act permit justification?	103
	Can health and safety concerns justify less favoural	ble
	treatment?	106
	Can medical information justify less favourable	
	treatment?	109
7.	Discrimination in the recruitment of employees	113
	Introduction	113
	General considerations	114
	Specifying the job	115
	Advertising the vacancy	118
	Application forms and information	121
	Selection, assessment and interview arrangements	123
	Offers of employment	131
8.	Discrimination against employees	133
	Introduction	122

Introduction	133
Terms and conditions of service	134
Induction, training and development	136
Benefits provided by employers	138
Promotion and transfer	140

	Managing disability or ill health	142
	Termination of employment	149
	After the termination of employment	151
9.	Discrimination in occupation	155
	Introduction	155
	Discrimination against contract workers	155
	Discrimination against office holders	160
	Discrimination against police officers	164
	Discrimination against partners in firms	165
	Discrimination against barristers	167
	Discrimination in relation to practical work experience	169
10.	Discrimination in providing occupational	
	pensions and group insurance services	177
	Introduction	177
	Occupational pension schemes	177
	Group insurance services	183
11.	Discrimination in providing employment	
	services	187
	Introduction	187
	What are employment services?	188
	What is unlawful under the Act?	188
	What amounts to discrimination?	191
	What is the duty to make reasonable adjustments?	192
	Can acts which are potentially discriminatory ever	
	be justified?	195
	Bringing claims for discrimination relating to employ	rment
	services	197

12.	2. Making reasonable adjustments to premises -	
	legal considerations	199
	Introduction	199
	What about the need to obtain statutory consent	
for some building changes? Building Regulations and building design		200
		201
	What if a binding obligation other than a lease	
	prevents a building being altered?	203
What happens if a lease says that certain changes to premises cannot be made?		
		203
	What happens if the lessor has a 'superior' lessor?	204
	How do arrangements for gaining consent work?	205
	When is it unreasonable for a lessor to withhold	
	consent?	207
	What conditions would it be reasonable for a	
	lessor to make when giving consent?	208
	What happens if the lessor refuses consent	
	or attaches conditions to consent?	209
	Comparison with the procedure for obtaining	
	consent under Part 3	210
13.	Other relevant provisions	213
	Resolving disputes	213
	Other provisions	219
	The Equality Commission for Northern Ireland	222
Appendix A: Changes to the Act		
Appendix B: The Meaning of Disability		
Appendix C: Further information		
Index	C	241

Purpose of Part 2 of the Act

- 1.1 The Disability Discrimination Act 1995 (the Act) brought in measures to prevent discrimination against disabled people. Part 2 (referred to in the legislation as 'Part II') of the Act is based on the principle that disabled people should not be discriminated against in employment or when seeking employment. It also protects disabled people engaged in a range of occupations. Employers must comply with the duties set out in Part 2, as must others to whom those duties apply (see paragraphs 3.8 to 3.14).
- 1.2 Except for the armed forces, these duties now apply to all employers no matter how many (or how few) people they employ. The law changed in this regard on 1 October 2004. Other major changes in the law which took effect on that date, including the extension of the duties to some people or bodies who are not employers, are summarised in Appendix A.

Purpose of the Code

1.3 This Code of Practice (the Code) gives practical guidance on how to prevent discrimination against disabled people in employment or when seeking employment. It describes the duties of employers and others in this regard. The Code helps disabled people to understand the law and what they can do if they feel that they have been discriminated against. By encouraging good practice, the Code assists employers to avoid workplace disputes and to work towards the elimination of discrimination against disabled people.

- 1.4 The Code also gives guidance on the law which is intended to help legal advisers when advising their clients, and to assist courts and industrial tribunals when interpreting new legal concepts. The Code explains the operation and effect of technical statutory provisions some of which only came into force on 1 October 2004, and many of which have a complex legal effect. As a consequence of this, the Code is necessarily comprehensive and detailed. However, the Equality Commission also produces a range of other publications about the Act, and about the rights of disabled people under it, which are intended to be of use to a range of audiences and for a variety of purposes. Details of how to obtain these publications are included in Appendix C.
- **s 54A** 1.5 The Equality Commission has prepared and issued the Code under the Act. It applies to Northern Ireland. A similar but separate Code applies to England, Scotland and Wales.

The Code replaces 'the Code of Practice for the elimination of discrimination in the field of employment against disabled persons or persons who have had a disability', issued by the Department of Economic Development in 1996.

Status of the Code

s 54A(8)

1.6 The Code does not impose legal obligations, nor is it an authoritative statement of the law - that is a matter for the courts and industrial tribunals. However, the Code can be used in evidence in legal proceedings under the Act. Courts and industrial tribunals must take into account any part of the Code that appears to them relevant to any question arising in those proceedings. If employers (and others who have duties under the Act's provisions on employment and occupation) follow the guidance in the Code, it may help to avoid an adverse

decision by a court or industrial tribunal in such proceedings.

How to use the Code

- 1.7 This chapter gives an introduction to the Code. Chapter
 2 sets out some general guidance on how to avoid discrimination. Chapter 3 contains an overview of the Act's provisions on employment and occupation, and those provisions are examined in more detail in subsequent chapters.
- 1.8 Chapter 4 details what is meant by discrimination and harassment, and Chapter 5 explains the duty to make reasonable adjustments for disabled people. Chapter 6 examines the relevance of justification under Part 2 of the Act. Chapters 7 and 8 consider the application of these principles in the context of recruitment processes and of subsequent employment, and Chapter 9 examines their application to certain occupations.
- 1.9 **Chapter 10** explains how the Act's provisions on employment and occupation operate in the particular context of occupational pension schemes and group insurance services.
- 1.10 **Chapter 11** describes how the Act deals with discrimination in providing employment services. Such services include vocational guidance and training.
- 1.11 **Chapter 12** looks at particular issues concerning adjustments to premises, and **Chapter 13** deals with various other points and explains what happens if discrimination is alleged.

- 1.12 Appendix A summarises some recent changes in the Act's provisions on employment and occupation.
 Appendix B gives more information on what is meant by 'disability' and by 'disabled person'. Separate statutory guidance relating to the definition of disability has been issued under the Act (see paragraph 3.6). Appendix C lists other sources of relevant information about matters referred to in the Code.
- 1.13 Each chapter of the Code should be viewed as part of an overall explanation of the Act's provisions on employment and occupation and the Regulations made under them. In order to understand the law properly it is necessary to read the Code as a whole. The Code should not be read too narrowly or literally. It is intended to explain the principles of the law, to illustrate how the Act might operate in certain situations and to provide general guidance on good practice. There are some questions which the Code cannot resolve and which must await the authoritative interpretation of the courts and industrial tribunals. The Code is not intended to be a substitute for taking appropriate advice on the legal consequences of particular situations.

Examples in the Code

- 1.14 Examples of good practice and how the Act is likely to work are given in boxes. They are intended simply to illustrate the principles and concepts used in the legislation and should be read in that light. The examples should not be treated as complete or authoritative statements of the law.
- 1.15 As stated in paragraph 1.2, the Act's provisions on employment and occupation now apply to small employers as well as to larger ones. The size of an employer is sometimes relevant to the way in which the

Act applies. Where this is the case, examples in the Code show how the application of the Act could be affected by the size of the employer.

1.16 While the examples refer to particular situations, they should be understood more widely as demonstrating how the law is likely to be applied generally. They can often be used to test how the law might apply in similar circumstances involving different disabilities or situations. Some of the examples are based on real cases which have been decided by the courts. In general, however, the examples attempt to use as many different varieties of disabilities and situations as possible to demonstrate the breadth and scope of the Act. Examples relating to men or women are given for realism and could, of course, apply to people of either gender. For ease of reference, it should be assumed, where appropriate, that the persons referred to in the examples meet the definition of disability under the Act.

References to 'employers' in the Code

1.17 Throughout the Code, references are made to 'employers' for convenience. However, as explained in paragraphs 3.11 and 3.12, the Act's provisions on employment and occupation impose obligations on persons who might not ordinarily be described as employers - such as partners in firms, barristers and people providing practical work experience. References in the Code need to be read in this light.

Other references in the Code

1.18 References to the Act are shown in the margins. For example, s 1(1) means section 1(1) of the Act and Sch 1 means Schedule 1 to the Act. Where reference is made to Regulations, the appropriate Statutory Rule (SR) number is shown in the margin (for example S.R. 2000 No. 389). References to Part 2 or 3 refer to the relevant Part of the Act.

Changes to the legislation

1.19 The Code refers to the Disability Discrimination Act 1995, as amended, as of 1 October 2004. There may be changes to the Act or to other legislation, for example, to the range of people who are considered to be 'disabled' under the Act, which may have an effect on the duties explained in the Code. You will need to ensure that you keep up to date with any developments that affect the Act's provisions. You can get relevant information from the Equality Commission (see below for contact details).

Further information

1.20 Copies of the Act and Regulations made under it can be obtained from The Stationery Office (TSO) - details can be obtained from the Equality Commission, or by visiting TSO's website: www.tso.co.uk. Guidance relating to the definition of disability is also available from TSO.

Further copies of the Code of Practice can be obtained directly from the Equality Commission's website: **www.equalityni.org**. The following Codes of Practice are also available from the Equality Commission:

- Disability Code of Practice: Trade Organisations and Qualifications Bodies;
- Code of Practice: Rights of Access Goods, Facilities, Services and Premises.

1.21 Free information about the Act can be obtained by contacting the Equality Commission:

The Equality Commission for Northern Ireland Equality House 7-9, Shaftesbury Square Belfast BT2 7DP

Telephone:028 90 500 600Textphone:028 90 500 589Fax:028 90 248 687E-mail:information@equalityni.orgWebsite:www.equalityni.org

Typetalk:

The Code and information about the Act are also available in other formats, including large print, Braille and audiotape. The Code can also be downloaded from the Equality Commission's website. If you require further assistance, please contact the Equality Commission.

Public Authorities

- 1.22 In addition to the duties imposed on employers by the Act, designated public authorities in Northern Ireland, which include government departments and local councils, are required by section 75 of the Northern Ireland Act 1998 when carrying out their functions to have due regard to the need to promote equality of opportunity:
 - between persons with a disability and persons without;

- between persons of different religious belief, political opinion, racial group, age, marital status or sexual orientation;
- between men and women generally; and
- between persons with dependants and persons without.

Public authorities must also have regard to the desirability of promoting good relations between persons of different religious belief, political opinion or racial group.

1.23 More detailed advice on these duties and information on preparing and implementing an equality scheme to comply with the legislation can be obtained from the Equality Commission's Statutory Duty Unit.

E-mail:	info.statduty@equalityni.org
Fax:	028 90 315 993
Textphone:	028 90 500 589
Telephone:	028 90 500 600

2 How can discrimination be avoided?

Introduction

2.1 There are various actions which employers can take in order to avoid discriminating against disabled people. By doing so, employers are not only likely to minimise the incidence of expensive and time-consuming litigation, but will also improve their general performance and the quality of their business operations. This chapter sets out some guidance on ways to help ensure that disabled people are not discriminated against.

Understanding the social dimension of disability

2.2 The concept of discrimination in the Act reflects an understanding that it is often environmental factors (such as the structure of a building) or an employer's working practices, rather than limitations arising from a person's disability, which unnecessarily restrict a disabled person's ability to participate fully in society. This principle underpins the duty to make reasonable adjustments described in Chapter 5. Understanding this will assist employers and others to avoid discrimination. It is as important to consider which aspects of employment and occupation create difficulties for a disabled person as it is to understand the particular nature of an individual's disability.

Recognising the diverse nature of disability

2.3 Around one in five people in Northern Ireland have a disability (approximately 340,0001¹). The nature and extent of their disabilities vary widely, as do their requirements for overcoming any difficulties they may face. If employers are to avoid discriminating, they need to understand this, and to be aware of the effects their decisions and actions - and those of their agents and employees - may have on disabled people. The evidence shows that many of the steps that can be taken to avoid discrimination cost little or nothing and are easy to implement.

Avoiding making assumptions

- 2.4 It is advisable to avoid making assumptions about disabled people. Disabilities will often affect different people in different ways and their needs may be different as well. The following suggestions may help to avoid discrimination:
 - Do not assume that because a person's disability cannot be seen, that person is not disabled. Many disabilities are not obvious.
 - Do not assume that because you do not know of any disabled people working within an organisation there are none.
 - Do not assume that most disabled people use wheelchairs.
 - Do not assume that people with learning disabilities cannot be valuable employees, or that they can only do low status jobs.

¹ This figure is obtained from the Census 2001. It should be noted that the term 'disability' was defined in this Census as 'any long-term illness, health problem or disability which limits daily activities or work'.

- Do not assume that a person with mental ill health cannot do a demanding job.
- Do not assume that all blind people read Braille or have guide dogs.
- Do not assume that all deaf people use sign language.
- Do not assume that because a disabled person may have less employment experience (in paid employment) than a non-disabled person, s/he has less to offer.

Seeking expert advice

2.5 It may be possible to avoid discrimination by using personal or in-house knowledge and expertise particularly if information or views are obtained from the disabled person concerned. However, although the Act does not specifically require anyone to obtain expert advice about meeting the requirements of disabled people with regard to employment, in practice it may sometimes be necessary to do so in order to comply with the principal duties set out in the Act. Expert advice might be especially useful if a person is newly disabled or if the effects of a person's disability become more marked. Expert advice about meeting the needs of disabled people may be available from the Disablement Advisory Service through JobCentres and Jobs and Benefits Offices across Northern Ireland, or from local and national disability organisations. Appendix C gives information about getting advice or help.

Planning ahead

- 2.6 The duties which the Act places on employers are owed to the individual disabled people with whom they have dealings. There is no duty owed to disabled people in general. Nevertheless, it is likely to be cost effective for employers to plan ahead. Considering the needs of a range of disabled people when planning for change (such as when planning a building refurbishment, a new IT system, or the design of a website) is likely to make it easier to implement adjustments for individuals when the need arises.
- 2.7 It is good practice for employers to have access audits carried out to identify any improvements which can be made to a building to make it more accessible. Access audits should be carried out by suitably qualified people. Websites and intranet sites can also be reviewed to see how accessible they are to disabled people using access software.

The owner of a small shop is planning a re-fit of her premises. As part of the re-fit she asks the designers to comply with British Standard 8300 to ensure that the shop has a good standard of access for a variety of disabled people, whether customers or employees. BS 8300 is a code of practice on the design of buildings and their approaches to meet the needs of disabled people (see Appendix C for details). The designers make her aware that any work carried out to her premises which may be a "structural alteration" will also have to meet the requirements of the Building Regulations (see paragraph 12.6). An employer is re-designing its website, which it uses to promote the company as well as to advertise vacancies. The employer ensures that the new design for the website is easy to read for people with a variety of access software; has the website checked for accessibility; and invites disabled readers of the website to let the employer know if they find any part of it inaccessible.

Implementing anti-discriminatory policies and practices

- 2.8 Employers are more likely to comply with their duties under the Act, and to minimise the risk of legal action being taken against them, if they implement antidiscriminatory policies and practices. Additionally, in the event that legal action is taken, employers may be asked to demonstrate to an industrial tribunal that they have effective policies, practices and procedures in place to minimise the risk of discrimination.
- 2.9 Outlined below are certain good practice measures which it is recommended employers put in place in order to prevent disability discrimination in the workplace. The good practice recommendations are also designed to help employers build an equality culture and to create a good and harmonious workplace environment in which all employees are treated with dignity and respect regardless of their disability. They aim to assist employers develop an environment in which employees understand in clear terms what behaviour is and is not acceptable and that appropriate disciplinary action will be taken against those who act in a discriminatory manner. They also give guidance on how employers can

build an environment in which individuals can raise complaints of discrimination without fear of repercussions or victimisation from work colleagues or management, and in which individuals can be confident that their complaints of discrimination will be treated seriously and dealt with effectively.

2.10 Employers should remember that the successful implementation of the following good practice measures requires the commitment of management, particularly senior management. Management should by words and actions demonstrate their commitment to eradicating unlawful discrimination and fully integrating equality of opportunity into the workplace. Adequate resources, in terms of staff, time, funding, etc should also be allocated.

Implement an equal opportunities policy

2.11 Employers should develop and implement a clear, comprehensive, effective and accessible equal opportunities policy, which specifically covers disability.

> The equal opportunities policy should spell out in clear terms the organisation's commitment to the promotion of equality of opportunity. The policy should also make it clear to employees that disability discrimination is unlawful and will not be tolerated in the workplace.

A model equal opportunities policy is available from the Equality Commission (for contact details please refer to paragraph 1.21 above).

Implement a harassment policy / procedure

2.12 It is also recommended that employers have a clear, comprehensive, effective and accessible harassment policy and procedure which covers complaints of harassment for a reason related to a person's disability. This policy and procedure should outline how employees can raise complaints of harassment and how complaints of harassment will be dealt with.

A model harassment policy and procedure is available from the Equality Commission (for contact details please refer to paragraph 1.21 above).

Consult on policies

2.13 It is also good practice for employers to consult with the appropriate recognised trade union(s), employee representatives or the workforce when developing and implementing an equal opportunities policy and harassment policy and procedure.

Communicate policies

2.14 It is vital that employers effectively communicate their equal opportunities policy, harassment policy and procedure and all other policies or procedures they may have relating to equal opportunities to all employees, contractors and agency staff.

> This could be done, for example, through staff briefings, contracts of employment, staff handbooks, notice boards, circulars, written notifications to individual employees, equal opportunities training, induction training, management training, training manuals, etc.

2.15 Employers should also take all available opportunities, especially when recruiting new staff, to ensure that their policies, practices and procedures are widely known. No one should be in any doubt about their equal opportunities policies and practice.

They should make it clear to their agents, for example recruitment consultants, what is required of them with regard to their duties and responsibilities under the Act and the extent of their authority. In addition, as a visible external sign of their commitment to equality of opportunity, it is recommended that employers note this commitment on relevant publications (e.g. company reports) and particularly in job advertisements. This commitment should also be made known to all job applicants.

2.16 An employer's equal opportunities policies and procedures should be accessible to all employees, particularly disabled employees. They should be accessible in respect of their format, content and implementation. Further information as regards accessibility can be obtained from the Equality Commission (see paragraph 1.21 for contact details).

Allocate responsibility

2.17 It is also recommended that employers allocate overall responsibility for their equal opportunities policies, practices and procedures to a member of senior management.

In addition, responsibility for ensuring compliance with equal opportunities policies and procedures should be incorporated into the job descriptions of all employees.

2.18 In larger organisations, as part of the appraisal process, employers should set objectives for personnel staff and those with managerial responsibilities in relation to how they have contributed to the implementation of the organisation's equal opportunities policies and procedures.

> In addition, where applicable, equal opportunities should be incorporated into an employer's business plans and strategies.

Deal effectively with complaints

2.19 It is essential that employers deal effectively with all complaints of disability discrimination.

Disability discrimination complaints should be dealt with promptly, seriously, sympathetically, confidentially and effectively. By dealing with complaints in this way, employers are reinforcing their message to their employees that they consider complaints of disability discrimination a serious matter.

- 2.20 Also, where reasonably practicable, employers should have a designated adviser(s) to whom individuals can speak to in confidence in order to provide support, advice and assistance to disabled people who believe they have been discriminated against. Designated advisers should receive appropriate training in order to enable them to carry out their roles effectively.
- 2.21 It is also vital that employers take all reasonably practicable steps to ensure that individuals who do raise complaints of disability discrimination are not victimised because of their complaints.
- 2.22 Employers should ensure that their complaints and grievance procedures are accessible to all employees, particularly disabled employees. They should be accessible in respect of their format, content and implementation.

Check disciplinary rules

2.23 It is recommended that employers check their disciplinary rules and/or procedures to ensure that unlawful discrimination is regarded as misconduct and could lead to disciplinary proceedings.

In addition appropriate action, including disciplinary action, should be taken in relation to employees who unlawfully discriminate under the Act.

Provide training and guidance

2.24 Employers should provide training and guidance for all employees to ensure that they understand their duties and responsibilities under the Act and the organisation's equal opportunities policies and procedures. The training and guidance should also include disability awareness and etiquette training.

> It is particularly important that personnel staff, managers and supervisors, designated advisers (see 2.20), and other employees who have key roles in relation to the planning and implementation of equality of opportunity in the workplace, receive appropriate training, so that they can perform their roles sensitively and effectively. The training should also ensure that such staff are familiar with the provisions of the Code.

Employers should ensure that all training and guidance is regular, relevant and up-to-date.

A large company provides disability awareness and equality training and guidance to all employees. The training and guidance is organised with the assistance of and input from disabled people who provide training in this area.

A large employer, with the assistance of the Equality Commission, trains all its employees in disability equality, the organisation's disability policy and the Disability Discrimination Act. It also trains all occupational health advisers with whom it works to ensure that they have the necessary expertise about the Act and the organisation's disability policy.

A small employer insists that all of the occupational health advisers that it uses can demonstrate that they have knowledge of the Act.

Disability related leave policy

2.25 In addition to the general good practice measures outlined above, it is recommended that employers establish a policy in relation to disability-related leave which fully complies with the provisions of the Act. Employers should also regularly monitor the implementation and effectiveness of such a policy.

Auditing policies and procedures

- 2.26 Although there is no duty under Part 2 to anticipate the requirements of disabled people in general, it is recommended that employers monitor and review their employment policies, practices and procedures to ensure that they are not unlawfully discriminating under the Act. Employers should consider the requirements of disabled people as part of this process. It is advisable for employers to do this in addition to having a specific policy to prevent discrimination. Employers are likely to have policies and/or procedures about matters such as:
 - flexible working arrangements;
 - appraisal and performance-related pay systems;

- sickness absence;
- redundancy selection criteria;
- emergency evacuation procedures;
- procurement of equipment, IT systems, software and websites;
- information provision;
- employee training and development;
- employee assistance schemes offering financial or emotional support;
- grievance and disciplinary procedures.

An organisation has a policy to ensure that all employees are kept informed about the organisation's activities through an intranet site. The policy says that the intranet site should be accessible to all employees, including those who use access software (such as synthetic speech output) because of their disabilities.

An employer has a policy of having annual appraisal interviews for all employees. The policy says that during the interviews, disabled employees should be asked whether they need any (further) reasonable adjustments. This could equally apply to a large or small employer. An employer introduces a system for performance related pay. It takes advice on performance related pay systems from an employers' organisation, to ensure that the system it introduces is an effective tool for improving performance and is fair to all employees. It also ensures that every year the system is monitored in order to assess the profile of pay awards in relation to disabled and other employees. The employer uses this information in order to check that disabled people, in addition to other groups protected by the equality legislation, do not, on average, get lower awards.

A redundancy policy that has sickness absence as a selection criterion is amended to exclude disability-related absence. The sickness absence policy is also changed so that disability-related sickness is recorded separately.

A new procurement policy sets out a number of factors that should be taken into account when procuring equipment. These factors include cost and energy efficiency. It is good practice for such factors to include accessibility for disabled people as well.

Emergency evacuation policies and procedures are reviewed to ensure that disabled people who require particular arrangements for emergency evacuation have individual plans.

Monitoring and Review

- 2.27 It is recommended that employers should also monitor and review the implementation of their equal opportunities policies and procedures. Monitoring and reviewing is an important way of determining whether anti-discrimination measures taken by an organisation are effective, and of ensuring that disability equality is achieved within that organisation.
- 2.28 There are various ways in which employers can monitor in the area of disability. Designated public authorities should refer to the 'Guide to the Statutory Duties' produced by the Equality Commission for further guidance in relation to their responsibilities as regards the collection of data in this area. Quantitative and qualitative monitoring techniques, as well as additional monitoring methods, are outlined below.
- 2.29 It is important to stress that monitoring will be more effective if employees and job applicants feel comfortable about disclosing information about their disabilities. Employers, who implement the recommended monitoring measures outlined below, should therefore take steps to reassure employees and job applicants why the information is required and how it will be used.
- 2.30 Employees and job applicants should also be reassured that any information given will be treated, subject to statutory requirements, in the strictest confidence. Employers should ensure that appropriate safeguards are put in place in order to achieve such confidentiality. Employers should also ensure that all monitoring forms are accessible as regards their format, content and implementation. Remember that all monitoring methods used should comply with the human rights and data protection legislation.

Quantitative Monitoring

- 2.31 It is recommended that employers carry out quantitative monitoring in relation to their employees and job applicants. Quantitative monitoring is concerned with the systematic collection of statistical data. It involves collecting data in order to calculate or estimate numbers or percentages. Employers in Northern Ireland will already be familiar with quantitative monitoring. For example, employers registered with the Equality Commission carry out quantitative monitoring in relation to the community background of their employees and job applicants.
- 2.32 By issuing employees and job applicants with a monitoring questionnaire, employers can, for example, obtain information on the number of disabled people it employs or who are applying for jobs in the organisation, the type of disabilities covered and the type of reasonable adjustment(s) which the disabled employee or applicant requires. A sample monitoring disability questionnaire is available from the Equality Commission.
- 2.33 Knowing the proportion of disabled people at various levels of the organisation, and at various stages in relation to the recruitment process, can help an organisation determine where practices and policies need to be improved.

Through the monitoring of candidates at the recruitment stage, an employer becomes aware that although several disabled people applied for a post, none were short-listed for interview. It uses this information to review the essential requirements for the post and other aspects of the recruitment process to ensure full compliance with the Act. 2.34 It is important, particularly in large organisations, to monitor by broad type of disability to understand the barriers faced by people with different types of impairment.

A large employer notices through monitoring that the organisation has been successful at retaining most groups of disabled people, but not people with mental ill health. It acts on this information by contacting a specialist organisation (for example, a user-led advocacy group) for advice about good practice in retaining people with mental ill health.

2.35 Monitoring can also give an employer an opportunity to ask disabled employees and job applicants what, if any, reasonable adjustments they require. In addition to seeking this information, employers should also regularly review the effectiveness of reasonable adjustments made for disabled people in accordance with the Act, and act on the findings of those reviews.

Qualitative Monitoring

- 2.36 It is also recommended that employers adopt qualitative monitoring techniques in the area of disability, in addition to quantitative monitoring. Qualitative monitoring involves directly consulting with disabled people or their representatives in order to obtain in depth their views, attitudes and opinions and to identify concerns or issues.
- 2.37 Listening carefully to disabled people and finding out what they want will help employers to meet their obligations by identifying the best way of meeting disabled people's requirements. There is a better chance of reaching the best outcome if discussions are held with disabled people at an early stage.

- 2.38 Often, discussing with disabled people what is necessary to meet their requirements will reassure an employer that suitable adjustments can be carried out at little expense and with very little inconvenience.
- 2.39 Evidence shows that in meeting the requirements of disabled employees an organisation learns how to meet those of disabled customers, and vice versa. By consulting with disabled employees, an organisation can therefore improve the service it provides to its disabled customers and enhance its business.
- 2.40 Measures can include consulting with individual disabled employees or their representatives, issuing anonymous questionnaires to employees or consulting with external disability organisations. Measures could also include consulting with a focus group within the organisation. The focus group may deal solely with issues surrounding disability or could also look at other equality issues in the workplace.
- 2.41 The type of measures used will often depend on the size of the organisation in question. Many larger employers have established formal structures for seeking and representing the views of disabled people. Small employers can also consult with disabled employees, although the methods may be less formal.
- 2.42 Qualitative monitoring allows employees and/or others the opportunity to provide feedback to an employer on the impact of the employer's policies, practices and procedures and the effectiveness of any reasonable adjustments made.

Additional measures

2.43 There are additional measures which it is recommended employers take in order to monitor and review the implementation of their equal opportunities policies and procedures.

- 2.44 As regards grievances or complaints of disability discrimination, an employer should monitor the number of complaints made, how they were resolved, what the outcomes were and how long each took to be resolved. Employers should also consider whether there are a greater number of complaints or grievances in certain departments or branches. Appropriate remedial action should be taken, when required.
- 2.45 Exit interviews can also provide a useful means of monitoring how well an organisation is integrating equal opportunities into the workplace. It is recommended that employers, where reasonably practicable, carry out exit interviews, either in person or by questionnaire, with employees who resign from their employment in order to check whether or not they believe they had been discriminated against contrary to the Act or any of the other grounds protected under the equality legislation. Employers should implement all required changes as a result of information obtained through the use of exit interviews.

Ensuring good practice in recruitment

Attracting disabled applicants

2.46 An organisation which recognises that suitably qualified disabled people have not applied to work for it may want to make contact with local employment services, including the Disablement Advisory Service through their local JobCentre and Jobs and Benefits Office and user-led advocacy groups, to encourage disabled people to apply. It is normally lawful for an employer to advertise a vacancy as open only to disabled people (see paragraph 7.5).

By monitoring the recruitment process a small employer notices that very few disabled people apply to work for it. In the light of this information, it decides to notify local disability employment projects of its vacancies. A retailer has a number of vacancies to fill. It contacts the Disablement Advisory Service through its local JobCentre and Jobs and Benefits Office and arranges an open morning for local disabled people to find out more about working for this employer.

Through its monitoring process, a medium-sized employer becomes aware of the fact that disabled people are under-represented in its workforce. It is looking for people to fill three work experience placements and decides to offer these placements to disabled people only.

A museum wants to understand the needs of its disabled visitors better. It decides to change its person specifications for posts in the visitor services department to include a requirement to have knowledge of disability access issues. It notifies local disability employment projects of these posts.

2.47 It is good practice to consider carefully what information should be included in advertisements and where they should be placed.

An advertisement which specifies that flexible working is available may encourage more disabled applicants to apply. An advertisement that appears in the disability press and a local talking newspaper may encourage disabled applicants to apply.

Promoting a positive image

2.48 It is good practice for an employer to consider its image to ensure that it gives an impression of itself as an organisation that is aware of the needs of disabled people and is striving to create a more diverse workforce.

A large employer ensures that its recruitment brochure includes images of disabled employees, and contains information about its disability policy.

A small employer advertises in a local newspaper. The advertisement states that disabled people are encouraged to apply.

Resolving disputes

2.49 Having policies and practices to combat discrimination, together with regular consultation with employees, is likely to minimise disputes about disability discrimination. When such disputes do occur, it is in the interests of employers to attempt wherever possible to resolve them as they arise. Grievance procedures can provide an open and fair way for employees to make their concerns known, and can enable grievances to be resolved quickly before they become major problems. Use of the procedures may highlight areas in which the duty to make reasonable adjustments has not been observed, and can prevent misunderstandings leading to complaints to industrial tribunals. It is important to ensure that grievance procedures are accessible to disabled people.

2.50 In certain circumstances, employers and employees, are required by law to comply with internal dispute resolution procedures before making a complaint to a tribunal. Chapter 13 contains further information about grievance procedures and about resolving disputes under the Act. Whether or not an attempt at internal resolution of a dispute is made as a result of a legal requirement, it should be carried out in a non-discriminatory way to comply with the Act.

The Act's provisions on employment and occupation an overview

Introduction

3.1 This chapter gives an overview of the provisions of the Act relating to employment and occupation. It explains who has rights and duties under those provisions and outlines what is made unlawful by them. Later chapters explain the provisions in greater detail.

Who has rights under the Act?

Disabled people

- 3.2 The Act gives protection from discrimination to a 'disabled' person within the meaning of the Act. A disabled person is someone who has a physical or mental impairment which has an effect on his or her ability to carry out normal day-to-day activities. That effect must be:
 - substantial (that is, more than minor or trivial); and
 - adverse; and
 - long term (that is, it has lasted or is likely to last for at least a year or for the rest of the life of the person affected).
- 3.3 Physical or mental impairment includes sensory impairment. Hidden impairments are also covered (for example, mental ill health, learning disabilities, dyslexia, diabetes and epilepsy).

ss 1 & 2 Sch 1 & 2 3.4 In considering its duties under the Act, an employer should not use any definition of 'disabled person' which is narrower than that in the Act. An employer who is requested to make a disability-related adjustment may ask the person requesting it for evidence that the impairment is one which meets the definition of disability in the Act. It may be appropriate to do so where the disability is not obvious. However, employers should not ask for more information about the impairment than is necessary for this purpose. Nor should they ask for evidence of disability where it ought to be obvious that the Act will apply.

A woman with ME (chronic fatigue syndrome) asks for time off to attend regular hospital appointments. The employer could legitimately ask to see a letter from the doctor or an appointment card. However, the employer then asks her questions about the likely progress of the illness so that he can bear this in mind when thinking about restructuring the department. This is likely to be unlawful.

People who have had a disability in the past

3.5 People who have had a disability within the meaning of the Act in the past are protected from discrimination even if they no longer have the disability.

A job applicant discloses on her application form that while at university from 1992 to 1993 she had long-term clinical depression after her father died. It would be discrimination to refuse to interview or recruit her because she has had a disability in the past. The fact that the disability preceded the Disability Discrimination Act 1995 is irrelevant.

More information about the meaning of disability

3.6 For a fuller understanding of the concept of disability under the Act, reference should be made to Appendix B. A Government publication, 'Guidance on matters to be taken into account in determining questions relating to the definition of disability' provides additional help in understanding the concept of disability and in identifying who is a disabled person. Where relevant, this Guidance must be taken into account in any legal proceedings.

People who have been victimised

3.7 The Act also gives rights to people who have been victimised, whether or not they have a disability or have had one in the past. (see paragraphs 4.33 to 4.36).

Who has obligations under the Act?

Employers

- 3.8 Later chapters explain in detail the duties which the Act imposes upon employers. The Act defines 'employment' as employment under a contract of service or of apprenticeship or a contract personally to do any work. Anyone who works under a contract falling within this definition is an employee, whether or not, for example, s/he works full time.
- 3.9 Members of the armed forces are excluded from protection under the Act's provisions on employment and occupation. Otherwise, those provisions now apply to all employers in respect of people they employ wholly or partly at an establishment in Northern Ireland. Protection under the Act extends to employment wholly outside Northern Ireland, provided that the employment has a sufficiently close connection with Northern Ireland - and

s 64(7) s 4(6)

s 68(2)-(2D)

the Act sets out the circumstances in which this will be the case.

Certain employment on board ships, hovercraft and aircraft is also covered.

3.10 A person who is recruiting an employee has duties under the Act even if s/he is not yet an employer (because the new recruit will be his or her first employee).

People or bodies concerned with certain occupations

- 3.11 The Act's definition of employment is wide enough to include people who are employed under a contract personally to do work. The provisions of Part 2 of the Act also extend to the following occupations which do not fall within the definition of employment:
 - contract workers;
 - office holders;
 - police officers;
 - partners in firms;
 - barristers;
 - people undertaking practical work experience for a limited period for the purposes of vocational training.

Many of the principles which apply to employers under Part 2 are equally applicable in respect of these occupations. Further details about the application of the Act's provisions on employment and occupation in this regard are set out in Chapter 9.

Others to whom Part 2 applies

- 3.12 In addition, the Act's provisions on employment and occupation may also impose obligations upon the following people and organisations:
 - trustees and managers of occupational pension schemes (see Chapter 10);
 - insurers who provide group insurance services for an employer's employees (see Chapter 10);
 - landlords of premises occupied by an employer or other person to whom Part 2 of the Act applies (see Chapter 12);
 - employees and agents of a person to whom Part 2 of the Act applies;
 - Ministers of the Crown, government departments and agencies.

Providers of employment services

3.13 Although not covered by Part 2 of the Act, the Act also contains provisions to prevent discrimination by people or organisations who provide employment services - such as employment agencies and careers guidance services (see Chapter 11).

Trade organisations and qualifications bodies

3.14 Finally, Part 2 makes special provision in respect of discrimination against disabled people by trade organisations and qualifications bodies. The nature and effect of the provisions in question is explained in a separate code of practice issued by the Equality Commission (see Appendix C for details).

ss 13-14B

Effect of the Act

- 3.15 The Act makes it unlawful for an employer to discriminate against a disabled person in relation to the recruitment or retention of staff.
- 3.16 However, the Act does not prohibit an employer from appointing the best person for the job. Nor does it prevent employers from treating disabled people more favourably than those who are not disabled.

Forms of discrimination

- 3.17 The four forms of discrimination which are unlawful under Part 2 are:
 - **direct discrimination** (the meaning of which is explained at paragraphs 4.5 to 4.23);
 - failure to comply with a duty to make reasonable adjustments (explained in Chapter 5);
 - 'disability-related discrimination' (see paragraphs 4.27 to 4.32); and
 - victimisation of a person (whether or not s/he is disabled) what the Act says about victimisation is explained at paragraphs 4.33 to 4.36.

Aspects of employment in respect of which discrimination is unlawful

s 4(1) 3.18 In relation to recruitment, the Act says that it is unlawful for an employer to discriminate against a disabled person:

	•	in the arrangements made for determining who should be offered employment;	
	•	in the terms on which the disabled person is offered employment; or	
	•	by refusing to offer, or deliberately not offering, the disabled person employment.	
	What this means in practice is explained in Chapter 7.		
3.19	In relation to the retention of staff, the Act says that it is unlawful for an employer to discriminate against a disabled person whom it employs:		s 4(2)
	•	in the terms of employment which it affords him or her;	
	•	in the opportunities which it affords him or her for promotion, a transfer, training or receiving any other benefit;	
	•	by refusing to afford him or her, or deliberately not affording him or her, any such opportunity; or	
	•	by dismissing him or her, or subjecting him or her to any other detriment.	
	Wha	at this means in practice is explained in Chapter 8.	
3.20	The Act also makes it unlawful for an employer to discriminate against a disabled person after that person's employment has come to an end (see paragraph 8.28).		16A(3)

What else is unlawful under the Act's provisions on employment and occupation?

Harassment

s 3B 3.21 In addition to what it says about discrimination, Part 2 makes it unlawful, in relation to the recruitment or retention of staff, for an employer to subject a disabled person to harassment for a reason which relates to his or her disability. What the Act says about harassment is explained in more detail at paragraphs 4.38 and 4.39.

Instructions and pressure to discriminate

s 16C
 3.22 It is also unlawful for a person who has authority or influence over another to instruct him or her, or put pressure on him or her, to act unlawfully under the provisions of Part 2 of the Act (or, insofar as they relate to employment services, Part 3). This covers pressure to discriminate, whether applied directly to the person concerned, or indirectly but in a way in which s/he is likely to hear of it. However, the Act does not give individual disabled people the right to take legal action in respect of unlawful instructions or pressure to discriminate. Such action may only be taken by the Equality Commission (see paragraphs 13.27 to 13.30).

Who is liable for unlawful acts?

Responsibility for the acts of others

s 58 3.23 Employers who act through agents (such as occupational health advisers or recruitment agencies) are liable for the actions of their agents done with the employer's express or implied authority. The Act also says that employers are responsible for the actions of their employees in the course of their employment. However, in legal proceedings against an employer

based on the actions of an employee, it is a defence that the employer took 'such steps as were reasonably practicable' to prevent such actions. It is not a defence for the employer simply to show that the action took place without its knowledge or approval. Chapter 2 gives guidance on the steps which it might be appropriate to take for this purpose.

A shopkeeper goes abroad for three months and leaves another person in charge of the shop. While he is away this person picks on a shop assistant with a learning disability, by constantly criticising the disabled person's work unfairly. The shop assistant resigns as a result of this bullying. The shopkeeper is responsible for the actions of the person left in charge.

Aiding an unlawful act

3.24 A person who knowingly helps another to do something made unlawful by the Act will be treated as having done the same kind of unlawful act. This means that, where an employer is liable for an unlawful act of its employee or agent, that employee or agent will be liable for aiding the unlawful act of the employer.

A recruitment consultant engaged by an engineering company refuses to consider a disabled applicant for a vacancy, because the company has told the consultant that it does not want the post filled by someone who has a disability. Under the Act the consultant could be liable for aiding the company to discriminate, in addition to the company's own liability for its unlawful act. 3.25 Where an employee discriminates against or harasses a disabled employee, it is the employer who will be liable for that unlawful act - unless it can show that it took such steps as were reasonably practicable to prevent the unlawful act in question. The employee who committed the discrimination or harassment will be liable for aiding the unlawful act - and this will be the case even if the employer is able to show that it took such steps as were reasonably practicable to prevent the act.

An employer has policies relating to harassment and disability. It ensures that all employees are aware of the policies and of the fact that harassment of disabled employees is subject to disciplinary action. It also ensures that managers receive training in applying the policies. A woman with a learning disability is humiliated by a colleague and disciplinary action is taken against the colleague. In these circumstances the colleague would be liable for aiding the unlawful act of the employer (the harassment) even though the employer would itself avoid liability because it had taken reasonably practicable steps to prevent the unlawful act.

Enforcing rights under Part 2

s 17A 3.26 Enforcement of rights under Part 2 takes place in the industrial tribunals. Enforcement of rights under the Act in relation to the provision of employment services also takes place in the industrial tribunals. More information about enforcement is given in Chapter 13.

4 What is discrimination and harassment?

Introduction

- 4.1 As noted at paragraph 3.17, the forms of discrimination which the Act makes unlawful in relation to employment are:
 - direct discrimination;
 - failure to comply with a duty to make reasonable adjustments;
 - disability-related discrimination; and
 - victimisation.
- 4.2 This chapter describes these four forms of discrimination in more detail, and explains the differences between them. It explores, in particular, the distinction between direct discrimination and disability-related discrimination (see paragraphs 4.27 to 4.31). These two forms of discrimination both depend on the way in which the employer treats the disabled person concerned - both require the disabled person to have been treated less favourably than other people are (or would be) treated. However, whether such treatment amounts to one of these forms of discrimination or the other (and, indeed, whether the treatment is unlawful in the first place) depends on the circumstances in which it arose.
- 4.3 The chapter examines the four forms of discrimination in the order in which they are listed in paragraph 4.1. This is because less favourable treatment which does not

amount to direct discrimination can sometimes be justified. (In contrast, neither direct discrimination, failure to comply with a duty to make a reasonable adjustment nor victimisation can ever be justified.) In deciding whether the treatment is justified, and therefore whether there has been disability-related discrimination, the Act requires the question of reasonable adjustments to be taken into account (see paragraphs 6.4 and 6.5 where this is explained in more detail). Consequently, although the chapter describes direct discrimination first, it touches on the subject of reasonable adjustments before moving on to disability-related discrimination.

4.4 This chapter also explains what the Act means by 'harassment'. The concepts of discrimination and harassment are relevant not only in relation to employment but also to the application of Part 2 of the Act in other situations - for example, in relation to the occupations mentioned in Chapter 9. The provisions about discrimination and harassment in Part 2 are also relevant to what the Act says about employment services in Part 3. This is described in Chapter 11.

What does the Act mean by 'direct discrimination'?

What does the Act say?

- **s 3A(5)** 4.5 The Act says that an employer's treatment of a disabled person amounts to direct discrimination if:
 - it is on the ground of the disabled person's disability;
 - the treatment is less favourable than the way in which a person not having that particular disability is (or would be) treated; and

- the relevant circumstances, including the abilities, of the person with whom the comparison is made are the same as, or not materially different from, those of the disabled person.
- 4.6 It follows that direct discrimination depends on an employer's treatment of a disabled person being on the ground of his or her disability. It also depends on a comparison of that treatment with the way in which the employer treats (or would treat) an appropriate comparator. If, on the ground of his or her disability, the disabled person is treated less favourably than the comparator is (or would be) treated, the treatment amounts to direct discrimination.

When is direct discrimination likely to occur?

- 4.7 Treatment of a disabled person is 'on the ground of' his or her disability if it is caused by the fact that the person is disabled or has the disability in question. In general, this means that treatment is on the ground of disability if a disabled person would not have received it but for his or her disability. Disability does not have to be the only (or even the main) cause of the treatment complained of. However, the effective cause of the less favourable treatment must be related to a person's disability and needs to be determined objectively from all the circumstances.
- 4.8 Consequently, if the less favourable treatment occurs because of the employer's generalised or stereotypical assumptions about the disability or its effects, it is likely to be direct discrimination. This is because an employer would not normally make such assumptions about a nondisabled person, but would instead consider the person's individual abilities.

Δ

A blind person is not short-listed for a job involving computers because the employer wrongly assumes that blind people cannot use them. The employer makes no attempt to look at the individual circumstances. The employer has treated the disabled person less favourably than other people by not short-listing that person for the job. The treatment was on the ground of the person's disability (because assumptions would not have been made about a non-disabled person).

4.9 In addition, less favourable treatment which is disabilityspecific, or which arises out of prejudice about disability (or about a particular type of disability), is also likely to amount to direct discrimination.

> An employer seeking a sales representative turns down a disabled applicant with a severe facial disfigurement solely on the ground that other employees would be uncomfortable working alongside the disabled person. This would amount to direct discrimination and would be unlawful.

A disabled woman who uses a wheelchair applies for a job. She can do the job just as well as any other applicant, but the employer wrongly assumes that the wheelchair will cause an obstruction in the office. The employer therefore gives the job to a person who is no more suitable for the job but who is not a wheelchair-user. This would amount to direct discrimination and would be unlawful.

- 4.10 In some cases, an apparently neutral reason for less favourable treatment of a disabled person may, on investigation, turn out to be a pretext for direct discrimination.
- 4.11 Direct discrimination will often occur where the employer is aware that the disabled person has a disability, and this is the reason for the employer's treatment of him or her. Direct discrimination need not be conscious - people may hold prejudices that they do not admit, even to themselves. Thus, people may behave in a discriminatory way whilst believing that they would never do so. Moreover, direct discrimination may sometimes occur even though the employer is unaware of a person's disability.

An employer advertises a promotion internally to its workforce. The personnel specification states that people with a history of mental illness would not be suitable for the post. An employee who would otherwise be eligible for the promotion has a history of schizophrenia, but the employer is unaware of this. The employee would, nevertheless, have a good claim for unlawful direct discrimination in relation to the promotion opportunities afforded to him by his employer. The act of direct discrimination in this case is the blanket ban on anyone who has had a mental illness, effectively rejecting whole categories of people with no consideration of their individual abilities.

4.12 In situations such as those described in the above examples, it will often be readily apparent that the disabled person concerned has been treated less favourably on the ground of his or her disability. In other cases, however, this may be less obvious. Whether or

not the basis for the treatment in question appears to be clear, a useful way of telling whether or not it is discriminatory, (and of establishing what kind of discrimination it is), is to focus on the person with whom the disabled person should be compared. That person may be real or hypothetical (see paragraph 4.18).

Identifying comparators in respect of direct discrimination

4.13 In determining whether a disabled person has been treated less favourably in the context of direct discrimination, his or her treatment must be compared with that of an appropriate comparator. This must be someone who does not have the same disability. It could be a non-disabled person or a person with other disabilities.

A person who becomes disabled takes six months' sick leave because of his disability, and is dismissed by his employer. A non-disabled fellow employee also takes six months' sick leave (because he has broken his leg) but is not dismissed. The difference in treatment is attributable to the employer's unwillingness to employ disabled staff and the treatment is therefore on the ground of his disability. The non-disabled employee is an appropriate comparator in the context of direct discrimination because his relevant circumstances are the same as those of the disabled person. It is the fact of having taken six months' sick leave which is relevant in these circumstances. As the disabled person has been treated less favourably than the comparator, this is direct discrimination.

4.14 It follows that, in the great majority of cases, some difference will exist between the circumstances (including the abilities) of the comparator and those of the disabled person - there is no need to find a comparator whose circumstances are the same as those of the disabled person in every respect. What matters is that the comparator's **relevant** circumstances (including his or her abilities) must be the same as, or not materially different from, those of the disabled person.

In the previous example, the position would be different if the employer's policy was to dismiss any member of staff who had been off sick for six months, and that policy was applied equally to disabled and non-disabled staff. In this case there would be no direct discrimination because the disabled person would not have been treated less favourably than the comparator - both would have been dismissed. Nevertheless, there may be a claim for failure to make reasonable adjustments to the policy, for example by allowing disability leave (see paragraph 4.25). In addition, the employer's policy may give rise to a claim for disability-related discrimination (see paragraph 4.27).

4.15 Once an appropriate comparator is identified, it is clear that the situation described in the example at paragraph 4.8 amounts to direct discrimination:

In the example about the blind person who is not short-listed for a job involving computers, there is direct discrimination because the disabled person was treated less favourably on the ground of the person's disability than an appropriate comparator (that is, a person who is not blind but who has the same abilities to do the job as the blind applicant): such a person would not have been rejected out of hand without consideration of his or her individual abilities.

4.16 The examples of direct discrimination in paragraph 4.9 also become clearer when the appropriate comparator is identified in each case:

In the example about the disabled person with a severe facial disfigurement who applies to be a sales representative, there is direct discrimination because the disabled person was treated less favourably on the ground of the person's disability than an appropriate comparator (that is, a person who does not have such a disfigurement but who does have the same abilities to do the job): such a person would not have been rejected in the same way.

In the example about the disabled woman who is not offered a job because she uses a wheelchair, there is direct discrimination because the woman was treated less favourably on the ground of her disability than an appropriate comparator (that is, a person who does not use a wheelchair but who does have the same abilities to do the job): such a person would not have been rejected in the same way.

- 4.17 The comparator used in relation to direct discrimination under the Act is the same as it is for other types of direct discrimination - such as direct sex discrimination. It is, however, made explicit in the Act that the comparator must have the same relevant abilities as the disabled person.
- 4.18 It may not be possible to identify an actual comparator whose relevant circumstances are the same as (or not materially different from) those of the disabled person in question. In such cases a hypothetical comparator may

be used. Evidence which helps to establish how a hypothetical comparator would have been treated is likely to include details of how other people (not satisfying the statutory comparison test as explained in paragraphs 4.13 and 4.14 above) were treated in circumstances which were broadly similar.

A disabled person works in a restaurant and makes a mistake on the till and this results in a small financial loss to the employer. The disabled person is dismissed because of this. The situation has not arisen before, and so there is no actual comparator. Nevertheless, six months earlier a non-disabled fellow employee was disciplined for taking home items of food without permission and received a written warning. The treatment of that person might be used as evidence that a hypothetical non-disabled member of staff who makes an error on the till would not have been dismissed for that reason.

4.19 It should be noted that the type of comparator described in the preceding paragraphs is only relevant to disability discrimination when assessing whether there has been **direct** discrimination. A different comparison is used when assessing whether there has been a failure to comply with a duty to make reasonable adjustments (see paragraphs 5.3 and 5.4) or when considering disabilityrelated discrimination (see paragraph 4.30).

Focusing on relevant circumstances

4.20 As stated in paragraph 4.14, direct discrimination only occurs where the **relevant** circumstances of the comparator, including his or her abilities, are the same as, or not materially different from, those of the disabled person. It is therefore important to focus on those

Δ

circumstances which are, in fact, relevant to the matter to which the less favourable treatment relates. Although, in some cases, the effects of the disability may be relevant, the fact of the disability itself is not a relevant circumstance for these purposes. This is because the comparison must be with a person **not** having that particular disability.

A disabled person with arthritis who can type at 30 words per minute (wpm) applies for an administrative job which includes typing. The employer rejects the application on the ground that the disabled person's speed is too slow. The correct comparator in a claim for direct discrimination would be a person not having arthritis who also has a typing speed of 30 wpm (with the same accuracy rate).

A disabled person with a severe visual impairment applies for a job as a bus driver. The employer rejects the application for the reason that the disabled person fails to meet the minimum level of visual acuity which is essential to the safe performance of the job. The correct comparator is a person not having that particular disability (for example, a person who merely has poorer than average eyesight) also failing to meet that minimum standard. A disabled person with schizophrenia applies for a job with his district council, and declares his history of mental illness. The district council refuses him employment, relying on a negative medical report from the council's occupational health adviser which is based on assumptions about the effects of schizophrenia, without adequate consideration of the individual's abilities and the impact of the impairment in his particular case. This is likely to amount to direct discrimination and to be unlawful. The comparator here is a person who does not have schizophrenia, but who has the same abilities to do the particular job in question (including relevant gualifications and experience) as the disabled applicant: such a person would not have been rejected without adequate consideration of his individual abilities.

4.21 If (as in the above examples) a disabled person alleges that s/he has been refused the offer of a job on the ground of his or her disability, it is only appropriate to compare those of the disabled person's circumstances which are relevant to his or her ability to do the job. It is not appropriate to compare other circumstances which are not relevant to this issue. The need to focus on relevant circumstances applies not only to recruitment cases of this kind, but also to any other situation where direct discrimination may have occurred. A disabled man with arthritis applies for an administrative job which includes typing, but is rejected in favour of a non-disabled candidate. The man has a slow typing speed and has difficulty walking because of his arthritis. The job is entirely desk-based, and does not require the person doing it to be able to walk further than a few metres within the office. The comparator in a claim of direct discrimination would be a non-disabled applicant with the same slow typing speed (and with the same abilities to do the job - e.g. the same typing accuracy rate, and the same knowledge of word-processing packages) - but it would not be necessary for the comparator to have mobility problems (because the ability to walk further than a few metres is not relevant to the candidates' ability to do the job).

Relevance of reasonable adjustments to comparison

4.22 In making the comparison in respect of a claim of direct discrimination, the disabled person's abilities must be considered as they in fact are. In some cases, there will be particular reasonable adjustments which an employer was required by the Act to make, but in fact failed to make. It may be that those adjustments would have had an effect on the disabled person's abilities to do the job. In making the comparison, the disabled person's abilities should be considered as they in fact were, and not as they would or might have been had those adjustments been made. On the other hand, if adjustments have in fact been made which have had the effect of enhancing the disabled person's abilities, then it is those enhanced abilities which should be considered. The disabled person's abilities are being considered as they in fact are (and not as they might have been if the adjustments had not been made).

A disabled person who owns an adapted keyboard applies for an administrative job which includes typing. The disabled person is not allowed to use the adapted keyboard (even though it would have been reasonable for the employer to allow this) and types a test document at 30 wpm. The disabled person's speed with the adapted keyboard would have been 50 wpm. A non-disabled candidate, who has a typing speed of 45 wpm with the same accuracy rate, is given the job. This is not direct discrimination, as the comparator is a non-disabled person typing at 30 wpm. (The disabled person would be likely to have good claims in respect of two other forms of discrimination - failure to make reasonable adjustments and disability-related discrimination - see paragraph 4.37.)

A disabled person with arthritis who applies for a similar job is allowed to use an adapted keyboard and types a test document at 50 wpm. A non-disabled candidate types at 30 wpm with the same accuracy rate. However, the disabled candidate is rejected because of prejudice and the other candidate is offered the job instead. This is direct discrimination, as the comparator would be a person not having arthritis who could type at 50 wpm.

Can direct discrimination be justified?

4.23 Treatment of a disabled person which amounts to direct discrimination under Part 2 of the Act is unlawful. It can never be justified.

s 3A(4)

61

Δ

Failure to make reasonable adjustments - relationship to discrimination

- 4.24 For the reason given in paragraph 4.3, it may be necessary to consider whether an employer has failed to comply with a duty to make a reasonable adjustment in order to determine whether disability-related discrimination has occurred.
- **s 3A(2)** 4.25 Irrespective of its relevance to disability-related discrimination, a failure to comply with a duty to make a reasonable adjustment in respect of a disabled person amounts to discrimination in its own right. Such a failure is therefore unlawful. Chapter 5 explains the circumstances in which an employer has such a duty, and gives guidance as to what employers need to do when the duty arises.
 - 4.26 As with direct discrimination, the Act does not permit an employer to justify a failure to comply with a duty to make a reasonable adjustment (see paragraphs 5.43 and 5.44).

What is disability-related discrimination?

What does the Act say?

- **s 3A(1)** 4.27 The Act says that an employer's treatment of a disabled person amounts to discrimination if:
 - it is for a reason related to his or her disability;
 - the treatment is less favourable than the way in which the employer treats (or would treat) others to whom that reason does not (or would not) apply; and

4

- the employer cannot show that the treatment is justified.
- 4.28 Although the Act itself does not use the term 'disabilityrelated discrimination', this expression is used in the Code when referring to treatment of a disabled person which:
 - is unlawful because each of the conditions listed in paragraph 4.27 is satisfied; but
 - does **not** amount to direct discrimination under the Act.
- 4.29 In general, direct discrimination occurs when the reason for the less favourable treatment in question is the disability, while disability-related discrimination occurs when the reason **relates to** the disability but **is not** the disability itself. The expression 'disability-related discrimination' therefore distinguishes less favourable treatment which amounts to direct discrimination from a wider class of less favourable treatment which, although not amounting to direct discrimination, is nevertheless unlawful.

When does disability-related discrimination occur?

4.30 In determining whether disability-related discrimination has occurred, the employer's treatment of the disabled person must be compared with that of a person **to whom the disability-related reason does not apply**. This contrasts with direct discrimination, which requires a comparison to be made with a person **without** the disability in question but whose relevant circumstances are the same. The comparator may be non-disabled or disabled - but the key point is that the disability-related reason for the less favourable treatment must not apply to him or her. 4

A disabled man is dismissed for taking six months' sick leave which is disability-related. The employer's policy, which has been applied equally to all staff (whether disabled or not) is to dismiss all employees who have taken this amount of sick leave. The disability-related reason for the less favourable treatment of the disabled person is the fact of having taken six months' sick leave, and the correct comparator is a person to whom that reason does not apply - that is, someone who has not taken six months' sick leave. Consequently, unless the employer can show that the treatment is justified (see paragraph) 6.6), it will amount to disability-related discrimination because the comparator would not have been dismissed. However, the reason for the treatment is not the disability itself (it is only a matter related thereto, namely the amount of sick leave taken). So there is no direct discrimination.

A disabled applicant is refused an administrative job due to an inability to type. The applicant cannot type because of arthritis. A non-disabled person who was unable to type would also have been turned down. The disability-related reason for the less favourable treatment is the applicant's inability to type, and the correct comparator is a person to whom that reason does not apply - that is, someone who can type. Such a person would not have been refused the job. Nevertheless, the disabled applicant has been treated less favourably for a disability-related reason and this will be unlawful unless it can be justified (see paragraph 6.6). There is no direct discrimination, however, because the comparator for direct discrimination is a person who does not have arthritis, but who is also unable to type.

4.31 The relationship between a disabled person's disability and the employer's treatment of him or her must be judged objectively. The reason for any less favourable treatment may well relate to the disability even if the employer does not have knowledge of the disability as such, or of whether its salient features are such that it meets the definition of disability in the Act. Less favourable treatment which is not itself direct discrimination will still be unlawful (subject to justification) if, in fact, the reason for it relates to the person's disability.

> A woman takes three periods of sickness absence in a two month period because of her disability, which is multiple sclerosis (MS). Her employer is unaware that she has MS and dismisses her, in the same way that it would dismiss any employee for a similar attendance record. Nevertheless, this is less favourable treatment for a disability-related reason (namely, the woman's record of sickness absence) and would be unlawful unless it can be justified.

4.32 The circumstances in which justification may be possible are explained in Chapter 6. However, it is worth noting that the possibility of justifying potential discrimination only arises at all when the form of discrimination being considered is disability-related discrimination, rather than direct discrimination or a failure to make reasonable adjustments.

What does the Act say about victimisation?

s 55(1) & (2)

- 4.33 Victimisation is a particular form of discrimination which is made unlawful by the Act. It is unlawful for one person to treat another ('the victim') less favourably than s/he treats or would treat other people in the same circumstances because the victim has:
 - brought, or given evidence or information in connection with, proceedings under the Act (whether or not proceedings are later withdrawn);
 - done anything else under or by reference to the Act; or
 - alleged someone has contravened the Act (whether or not the allegation is later withdrawn);

or because the person believes or suspects that the victim has done or intends to do any of these things.

A disabled employee complains of discrimination, having been refused promotion at work. A colleague gives evidence at the tribunal hearing on his behalf. The employer makes the disabled person's colleague redundant because of this. This amounts to victimisation. It would also be unlawful to subject a colleague to any detriment where he attends the tribunal not to give evidence but purely to offer support to the disabled person bringing the complaint because this would be something which is done by reference to the Act.

s 55(4) 4.34 It is not victimisation to treat a person less favourably because that person has made an allegation which was false and not made in good faith.

- 4.35 However, the fact that a person has given evidence on behalf of an applicant in a claim which was unsuccessful does not, of itself, prove that his or her evidence was false or that it was not given in good faith.
- 4.36 Unlike the other forms of discrimination which are made unlawful by the Act, victimisation may be claimed by people who are not disabled as well as by those who are.

How do the different forms of discrimination compare in practice?

4.37 The way in which the different forms of discrimination which are unlawful under the Act's provisions on employment and occupation may operate in practice can be demonstrated by the following series of examples.

A person with arthritis applies for a secretarial job in a local business. There is a question on the application form about disability. The applicant has arthritis and states this on the form. The arthritis does not affect the applicant's typing ability. This fact is also stated by the applicant on the form. The employer rejects the application because it nevertheless wrongly assumes that the disabled person will not be able to carry out the job due to the arthritis. This is **direct discrimination**. It cannot be justified.

In the situation described above, the arthritis instead does affect the applicant's typing ability and this is stated in the application form. The applicant is called for an interview and is told that a typing test forms part of the selection process. The applicant requests the use of an adapted keyboard in order to take the test, but this is not provided on the day of the interview, and s 55(5)

4

the applicant fails the test as a result. As a consequence of failing the test, the applicant is turned down for the job. This is not direct discrimination, as the reason for the employer's rejection of the applicant was not the disability, but the fact that the applicant failed the typing test.

However, in such circumstances the employer has a duty to make reasonable adjustments to its selection arrangements. Depending on the circumstances, it may be a reasonable adjustment for the employer to provide the adapted keyboard or allow the applicant to use the applicant's own keyboard in order that the applicant is not placed at a substantial disadvantage by the test. If this is the case, then the employer will be unlawfully discriminating against the applicant by **failing to make the adjustment.**

Although there is no direct discrimination, the employer has still treated the applicant less favourably for a reason relating to the applicant's disability (namely the fact that the applicant failed the typing test). This will be **disability-related discrimination** unless the employer can show that it is justified (see paragraph 6.6) - and the employer will be unable to show this if it would have been reasonable for it to provide the applicant with an adapted keyboard or allow the applicant to use the applicant's own keyboard in order to take the typing test.

As a result of this treatment, the applicant makes a claim against the employer under Part 2 of the Act. Some time later, however, the same employer advertises a further secretarial vacancy. The applicant applies again, but the employer rejects the application because the applicant has previously made a claim under the Act. This is victimisation.

What does the Act say about harassment?

- 4.38 The Act says that harassment occurs where, for a reason which relates to a person's disability, another person engages in unwanted conduct which has the purpose or effect of:
 - violating the disabled person's dignity; or
 - creating an intimidating, hostile, degrading, humiliating or offensive environment for him or her.
- 4.39 If the conduct in question was engaged in with the intention that it should have either of these effects, then it amounts to harassment irrespective of its actual effect on the disabled person. In the absence of such intention, however, the conduct will only amount to harassment if it should reasonably be considered as having either of these effects. Regard must be had to all the circumstances in order to determine whether this is the case. Those circumstances include, in particular, the perception of the disabled person.

A man with a learning disability is often called 'stupid' and 'slow' by a colleague at work. This is harassment, whether or not the disabled man was present when these comments were made, because they were said with the intention of humiliating him.

A man with a stammer feels he is being harassed because his manager makes constant jokes about people with speech impairments. He asks his manager to stop doing this, but the manager says he is being 'oversensitive' as he habitually makes jokes in the office about many different sorts of people. This is likely to amount to harassment because making remarks of this kind should reasonably be considered as having either of the effects mentioned above. s 3B(1)

Δ

s 3B(2)

An employee with symptomatic HIV uses a colleague's mug. The colleague then makes a point of being seen washing the mug with bleach, which is not something she would do if anyone else used her mug. She also makes offensive comments about having her mug used by someone with symptomatic HIV. This is likely to amount to harassment.

An employee circulates by email a joke about people with autism. A colleague with autism receives the email and finds the joke offensive. This is likely to amount to harassment.

A woman with depression considers that she is being harassed by her manager who constantly asks her if she is feeling all right, despite the fact that she has asked him not to do so in front of the rest of the team. This could amount to harassment.

What does the Act say about statutory obligations?

s 59

9 4.40 Nothing is made unlawful by the Act if it is required by an express statutory obligation. However, it is only in cases where a statutory obligation is specific in its requirements, leaving an employer with no choice other than to act in a particular way that the provisions of the Act may be overridden. The provision in relation to statutory obligations has a narrow application, and it is likely to permit disability discrimination only in rare circumstances.

What evidence is needed to prove that discrimination or harassment has occurred?

- 4.41 As stated in paragraph 3.26, enforcement of rights under the Act's provisions on employment and occupation takes place in the industrial tribunals. A person who brings a claim for unlawful discrimination or harassment must show that discrimination has occurred. S/he must prove this on the balance of probabilities in order to succeed with a claim in an industrial tribunal.
- 4.42 However, the Act says that, when such a claim is heard by an industrial tribunal, the tribunal must uphold the claim if:
 - the claimant proves facts from which the tribunal could conclude in the absence of an adequate explanation that the person against whom the claim is made (the respondent) has acted unlawfully; and
 - the respondent fails to prove that s/he did not act in that way.

A disabled employee scores very poorly in a redundancy selection process in comparison with other employees in the same position as himself, getting low marks for skill and performance in his job. The employee has always had good appraisals in comparison with other employees and no action has ever been taken against him in respect of competence. Unless the employer demonstrates a nondiscriminatory reason for the low scores, unlawful discrimination will be inferred in these circumstances. s 17A(1B)

- 4.43 Consequently, where a disabled person is able to prove on the balance of probabilities facts from which an inference of unlawful discrimination or harassment could be drawn, the burden of proof shifts to the respondent for example, the disabled person's employer. This means that the employer must show that it is more likely than not that its conduct was not unlawful. This principle applies to allegations in respect of all forms of discrimination, including victimisation and harassment. Its practical effect in relation to the three principal forms of disability discrimination can be summarised as follows:
 - To prove an allegation of **direct discrimination**, an employee must prove facts from which it could be inferred in the absence of an adequate explanation that s/he has been treated less favourably on the ground of his or her disability than an appropriate comparator has been, or would be, treated. If the employee does this, the claim will succeed unless the employer can show that disability was not any part of the reason for the treatment in question.
 - To prove an allegation that there has been a **failure to comply with a duty to make reasonable adjustments**, an employee must prove facts from which it could be inferred in the absence of an adequate explanation that such a duty has arisen, and that it has been breached. If the employee does this, the claim will succeed unless the employer can show that it did not fail to comply with its duty in this regard.
 - To prove an allegation of **disability-related discrimination**, an employee must prove facts from which it could be inferred in the absence of an adequate explanation that, for a reason relating to his or her disability, s/he has been treated less favourably than a person to whom that reason does not apply has been, or would be, treated. If the

4

employee does this, the burden of proof shifts, and it is for the employer to show that the employee has not received less favourable treatment for a disability-related reason. Even if the employer cannot show this, however, the employee's claim will not succeed if the employer shows that the treatment was justified.

- 4.44 The Act provides a means by which a disabled person can seek evidence about whether s/he has been discriminated against, or subjected to harassment, under the Act's provisions on employment and occupation. S/he may do this by using a questionnaire to obtain further information from a person s/he thinks has acted unlawfully in relation to him or her (see paragraph 13.13). If there has been a failure to provide a satisfactory response to questions asked by the disabled person in this way, inferences may be drawn from that failure.
- 4.45 In addition, the fact that there has been a failure to comply with a relevant provision of the Code must be taken into account by a court or industrial tribunal, where it considers it relevant, in determining whether there has been discrimination or harassment (see paragraph 1.6).

s 56

Δ

Introduction

- 5.1 In Chapter 4 it was noted that one of the ways in which discrimination occurs under Part 2 of the Act is when an employer fails to comply with a duty imposed on it to make 'reasonable adjustments' in relation to the disabled person. This chapter examines the circumstances in which a duty to make reasonable adjustments arises and outlines what an employer needs to do in order to discharge such a duty.
- 5.2 The concept of a duty to make reasonable adjustments is also relevant to the application of Part 2 of the Act in other situations - for example, in relation to the occupations mentioned in Chapter 9, and to the provision of occupational pensions, as explained in Chapter 10.

When does an employer's duty to make reasonable adjustments arise?

5.3 The duty to make reasonable adjustments arises where a provision, criterion or practice applied by or on behalf of the employer, or any physical feature of premises occupied by the employer, places a disabled person at a substantial disadvantage compared with people who are not disabled. An employer has to take such steps as it is reasonable for it to have to take in all the circumstances to prevent that disadvantage - in other words the s 4A(1)

employer has to make a 'reasonable adjustment'. Where the duty arises, an employer cannot justify a failure to make a reasonable adjustment.

A person with dyslexia applies for a job which involves writing letters. The employer gives all applicants a test of their letter-writing ability. The disabled applicant can generally write letters very well but finds it difficult to do so in stressful situations and within short deadlines. The disabled applicant is given longer to take the test. This adjustment is likely to be a reasonable one for the employer to make.

5.4 It does not matter if a disabled person cannot point to an actual non-disabled person compared with whom s/he is at a substantial disadvantage. The fact that a non-disabled person, or even another disabled person, would not be substantially disadvantaged by the provision, criterion or practice or by the physical feature in question is irrelevant. The duty is owed specifically to the individual disabled person.

Which disabled people does the duty protect?

5.5 The duty to make reasonable adjustments applies in recruitment and during all stages of employment, including dismissal. It may also apply after employment has ended. The duty relates to all disabled employees of an employer and to any disabled applicant for employment. In the case of a provision, criterion or practice for determining to whom employment should be offered, the duty also applies in respect of any disabled person who has notified the employer that s/he may be an applicant for that employment.

- 5.6 The extent of the duty to make reasonable adjustments depends on the employment circumstances of the disabled person in question. For example, more extensive duties are owed to employees than to people who are merely thinking about applying for a job. More wide ranging duties are also owed to current employees than to former employees. The extent to which employers have knowledge of relevant circumstances is also a factor (see paragraphs 5.12 to 5.16).
- 5.7 In order to avoid discrimination, it would be prudent for employers not to attempt to make a fine judgement as to whether a particular individual falls within the statutory definition of disability, but to focus instead on meeting the requirements of each employee and job applicant.

What are 'provisions, criteria and practices'?

5.8 'Provisions, criteria and practices' include arrangements, for example, for determining to whom employment should be offered, and terms, conditions or arrangements on which employment, promotion, a transfer, training or any other benefit is offered or afforded. The duty to make reasonable adjustments applies, for example, to selection and interview procedures and the premises used for such procedures, as well as to job offers, contractual arrangements and working conditions.

A call centre normally employs supervisors on a fulltime basis. A person with sickle cell anaemia applies for a job as a supervisor. The person asks to do the job on a part-time basis because of pain and fatigue relating to this condition. The call centre agrees. The hours of work which are offered amount to an adjustment to a working practice. This is likely to be a reasonable adjustment to the call centre's working practice. s 18D(2)

An employer has a policy that designated car parking spaces are only offered to senior managers. An employee who is not a manager, but who has a mobility impairment and needs to park very close to the office, is given a designated car parking space. This is likely to be a reasonable adjustment to the employer's car parking policy.

What is a 'physical feature'?

5.9 The Act says that the following are to be treated as a physical feature:

- any feature arising from the design or construction of a building on the premises occupied by the employer;
- any feature on the premises of any approach to, exit from, or access to such a building;
- any fixtures, fittings, furnishings, furniture, equipment or material in or on the premises; and
- any other physical element or quality of any land comprised in the premises occupied by the employer.

All these features are covered, whether temporary or permanent. Considerations which need to be taken into account when making adjustments to premises are explained in Chapter 12.

The design of a particular workplace makes it difficult for someone with a hearing impairment to hear, because the main office is open plan and has hard flooring. That is a substantial disadvantage caused by the physical features of the workplace.

s 18D(2)

Clear glass doors at the end of a corridor in a particular workplace present a hazard for a visually impaired employee. This is a substantial disadvantage caused by the physical features of the workplace.

5.10 Physical features will include steps, stairways, kerbs, exterior surfaces and paving, parking areas, building entrances and exits (including emergency escape routes), internal and external doors, gates, toilet and washing facilities, lighting and ventilation, lifts and escalators, floor coverings, signs, furniture, and temporary or movable items. This is not an exhaustive list.

What disadvantages give rise to the duty?

5.11 The Act says that only substantial disadvantages give rise to the duty. Substantial disadvantages are those which are not minor or trivial. Whether or not such a disadvantage exists in a particular case is a question of fact. What matters is not that a provision, criterion or practice or a physical feature is capable of causing a substantial disadvantage to the disabled person in question, but that it actually has this effect on him or her, or (where applicable) that it would have this effect if the disabled person was doing the job at the time.

What if the employer does not know that the person is disabled?

5.12 Although (as explained in paragraphs 4.11 and 4.31) less favourable treatment can occur even if the employer does not know that an employee is disabled, the employer only has a duty to make an adjustment if it s 4A(3)(b)

knows, or could reasonably be expected to know, that the employee has a disability and is likely to be placed at a substantial disadvantage. The employer must, however, do all it can reasonably be expected to do to find out whether this is the case.

An employee with depression sometimes gets upset at work, but the reason for this behaviour is not known to her employer. The employer makes no effort to find out if the employee is disabled and whether a reasonable adjustment could be made to the person's working arrangements. The employee is disciplined without being given any opportunity to explain that the problem arises from a disability. The employer may be in breach of the duty to make reasonable adjustments because it failed to do all it could reasonably be expected to do to establish if the employee was disabled and substantially disadvantaged.

An employer has an annual appraisal system which specifically provides an opportunity for employees to notify the employer in confidence if they are disabled and are put at a substantial disadvantage by the working arrangements or premises. This gives the employer the opportunity to find out if an employee requires reasonable adjustments, although it does not mean that the employer should not consider reasonable adjustments for an employee at other times of the year.

s 4A(3) 5.13 The principle stated in paragraph 5.12 applies equally to a disabled person who is an actual or potential applicant for employment.

An applicant is not short-listed for interview for the position of administrative assistant, a post which mainly involves typing. The applicant cannot type and states this on the application form. The applicant cannot type because of severe arthritis, but this is not stated anywhere on the form. The employer would not be expected to make an adjustment to the typing requirement in these circumstances as it had no knowledge of the disability and could not reasonably be expected to know of it.

- 5.14 In addition, an employer only has a duty to make an adjustment if it knows, or could reasonably be expected to know, that the actual or potential applicant is disabled.
- 5.15 If an employer's agent or employee (such as an occupational health adviser, a personnel officer or line manager or recruitment agent) knows, in that capacity, of an employee's disability, the employer will not usually be able to claim that it does not know of the disability, and that it therefore has no obligation to make a reasonable adjustment. The same applies in respect of actual or potential applicants for employment. Employers therefore need to ensure that where information about disabled people may come through different channels, there is a means suitably confidential for bringing the information together, to make it easier for the employer to fulfil its duties under the Act.

An occupational health adviser is engaged by a large employer to provide it with information about its employees' health. The occupational health adviser becomes aware of an employee's disability, which the employee's line manager does not know about. The employer's working arrangements put the employee at a substantial disadvantage because of the effects of her disability and she claims that a reasonable adjustment should have been made. It will not be a defence for the employer to claim that it did not know of her disability. This is because the information gained by the occupational health adviser on the employer's behalf is imputed to the employer. The occupational health adviser's knowledge means that the employer's duty under the Act applies. If the employee did not give consent for the occupational health adviser to pass on personal information to the line manager, it might be necessary for the line manager to implement the reasonable adjustment without knowing precisely why s/he has to do so.

5.16 Information will not be imputed to the employer if it is gained by a person providing services to employees independently of the employer. This is the case even if the employer has arranged for those services to be provided.

An employer contracts with an agency to provide an independent counselling service to employees. The contract says that the counsellors are not acting on the employer's behalf while in the counselling role. Any information about a person's disability obtained by a counsellor during such counselling would not be imputed to the employer and so would not trigger the employer's duty to make reasonable adjustments.

Does the duty to make reasonable adjustments apply in other situations related to employment and occupation?

5.17 Paragraphs 5.3 to 5.16 explain when it may be necessary to make an adjustment in relation to employment. Part 2 of the Act imposes similar requirements in relation to the occupations it covers, subject to certain differences as explained in Chapter 9. Reasonable adjustments may also be required in relation to occupational pension schemes and group insurance services, as explained in Chapter 10.

What adjustments might an employer have to make?

5.18 The Act gives a number of examples of adjustments, or 'steps', which employers may have to take, if it is reasonable for them to have to do so (see paragraphs 5.24 to 5.42). Any necessary adjustments should be implemented in a timely fashion, and it may also be necessary for an employer to make more than one adjustment. It is advisable to agree any proposed adjustments with the disabled person in question before they are made. The Act does not give an exhaustive list of the steps which may have to be taken to discharge the duty. Steps other than those listed here, or a combination of steps, will sometimes have to be taken. However, the steps in the Act are:

s 18B(2)

making adjustments to premises;

An employer makes structural or other physical changes such as widening a doorway, providing a ramp or moving furniture for a wheelchair user; relocating light switches, door handles or shelves for someone who has difficulty in reaching; or providing appropriate colour contrast in decor to help the safe mobility of a visually impaired person.

 allocating some of the disabled person's duties to another person;

An employer reallocates minor or subsidiary duties to another employee as a disabled person has difficulty doing them because of the disability. For example, the job involves occasionally going onto the open roof of a building but the employer transfers this work away from an employee whose disability involves severe vertigo.

transferring the disabled person to fill an existing vacancy;

An employer should consider whether a suitable alternative post is available for an employee who becomes disabled (or whose disability worsens), and where no reasonable adjustment would enable the employee to continue doing the current job. Such a post might also involve retraining or other reasonable adjustments such as equipment for the new post. altering the disabled person's hours of working or training;

This could include allowing a disabled person to work flexible hours to enable him to have additional breaks to overcome fatigue arising from his disability. It could also include permitting part time working, or different working hours to avoid the need to travel in the rush hour if this is a problem related to an impairment. A phased return to work with a gradual build-up of hours might also be appropriate in some circumstances.

 assigning the disabled person to a different place of work or training;

During the course of her employment, an employee becomes disabled and now requires the use of a wheelchair. Assuming that there are no reasonable steps which could be taken to make her work station accessible, it might be reasonable to relocate that employee from an inaccessible third floor office to an accessible one on the ground floor. It could mean moving the employee to other premises of the same employer if the original place of work is inaccessible.

 allowing the disabled person to be absent during working or training hours for rehabilitation, assessment or treatment;

An employer allows a person who has become disabled more time off during work than would be allowed to non-disabled employees to enable him to have rehabilitation training. A similar adjustment would be appropriate if a disability worsens or if a disabled person needs occasional treatment anyway. giving, or arranging for, training or mentoring (whether for the disabled person or any other person);

This could be training in particular pieces of equipment which the disabled person uses, or an alteration to the standard employee training to reflect the employee's particular disability. For example, all employees are trained in the use of a particular machine but an employer provides slightly different or longer training for an employee with restricted hand or arm movements, or training in additional software for a visually impaired person so that he can use a computer with speech output.

An employer provides training for employees on conducting meetings in a way that enables a deaf staff member to participate effectively.

A disabled man returns to work after a six-month period of absence due to a stroke. His employer pays for him to see a work mentor, and allows time off to see the mentor, to help with his loss of confidence following the onset of his disability.

acquiring or modifying equipment;

An employer might have to provide special equipment (such as an adapted keyboard for someone with arthritis or a large screen for a visually impaired person), an adapted telephone for someone with a hearing impairment, or other modified equipment for disabled employees (such as longer handles on a machine). There is no requirement to provide or modify equipment for personal purposes unconnected with an employee's work, such as providing a wheelchair if a person needs one in any event but does not have one. The disadvantage in such a case does not flow from the employer's arrangements or premises.

• modifying instructions or reference manuals;

The format of instructions and manuals might need to be modified for some disabled people (eg, produced in Braille or on audio tape) and instructions for people with learning disabilities might need to be conveyed orally with individual demonstration.

modifying procedures for testing or assessment;

This could involve ensuring that particular tests do not adversely affect people with particular types of disability. For example, a person with restricted manual dexterity would be disadvantaged by a written test, so the employer gives that person an oral test instead.

• providing a reader or interpreter;

A colleague reads mail to a person with a visual impairment at particular times during the working day. Alternatively, the employer might hire a reader or sign language interpreter. • providing supervision or other support.

An employer provides a support worker, or arranges help from a colleague, in appropriate circumstances, for someone whose disability leads to uncertainty or lack of confidence.

5.19 It may sometimes be necessary for an employer to take a combination of steps.

A woman who is deafblind is given a new job with her employer in an unfamiliar part of the building. The employer (i) arranges facilities for her guide dog in the new area, (ii) arranges for her new instructions to be in Braille, (iii) trains colleagues to communicate with her, and (iv) provides disability equality training to all staff.

- 5.20 As mentioned above, it might be reasonable for employers to have to take other steps, which are not given as examples in the Act. These steps could include:
 - conducting a proper assessment of what reasonable adjustments may be required;
 - permitting flexible working;
 - allowing a disabled employee to take a period of disability leave.

An employee who has cancer needs to undergo treatment and rehabilitation. His employer allows a period of disability leave and permits him to return to his job at the end of this period. participating in supported employment schemes, provided by the Government;

A person takes a job as an office assistant after several years not working due to depression. The person has been participating in a supported employment scheme. As a reasonable adjustment, the person is allowed by the employer to make private phone calls during the working day to a support worker at the supported employment scheme for additional support and confidence building.

 employing a support worker to assist a disabled employee;

An adviser with a visual impairment is sometimes required to make home visits. The employer employs a support worker to assist the disabled employee on these visits.

modifying disciplinary or grievance procedures;

An employee with a learning disability is allowed to take a friend (who does not work with the employee) to act as an advocate at a meeting with the employer about a grievance. The employer also ensures that the meeting is conducted in a way that does not disadvantage or patronise the disabled employee.

adjusting redundancy selection criteria;

A woman with an autoimmune disease has taken several short periods of absence during the year because of the condition. When her employer is taking absences into account as a criterion for selecting people for redundancy, periods of disabilityrelated absence are discounted.

modifying performance-related pay arrangements.

A disabled woman who is paid purely on her output needs frequent short additional breaks during her working day - something her employer agrees to as a reasonable adjustment. It is likely to be a reasonable adjustment for her employer to pay her at an agreed rate (e.g. her average hourly rate) for these breaks.

- 5.21 Advice and assistance (which may include financial assistance) in relation to making adjustments may be available from the Access to Work scheme (see paragraphs 8.19 and 8.20).
- 5.22 In some cases a reasonable adjustment will not work without the co-operation of other employees. Employees may therefore have an important role in helping to ensure that a reasonable adjustment is carried out in practice. Subject to considerations about confidentiality (explained at paragraphs 8.21 to 8.23), employers must ensure that this happens. It is unlikely to be a valid defence to a claim under the Act that staff were obstructive or unhelpful when the employer tried to make reasonable adjustments. An employer would at least need to be able to show that it took such behaviour seriously and dealt with it appropriately. Employers will

be more likely to be able to do this if they establish and implement the type of policies and practices described in Chapter 2.

An employer ensures that an employee with autism has a structured working day as a reasonable adjustment. As part of the reasonable adjustment it is the responsibility of the employer to ensure that other employees co-operate with this arrangement.

5.23 Further examples of the way in which reasonable adjustments work in practice are given in Chapters 7 and 8, which deal with recruitment and retention.

When is it 'reasonable' for an employer to have to make adjustments?

5.24 Whether it is reasonable for an employer to make any particular adjustment will depend on a number of things, such as its cost and effectiveness. However, if an adjustment is one which it is reasonable to make, then the employer must do so. Where a disabled person is placed at a substantial disadvantage by a provision, criterion or practice of the employer, or by a physical feature of the premises it occupies, the employer must consider whether any reasonable adjustments can be made to overcome that disadvantage. There is no onus on the disabled person to suggest what adjustments should be made (although it is good practice for employers to ask) but, where the disabled person does so, the employer must consider whether such adjustments would help overcome the disadvantage, and whether they are reasonable.

A disabled employee has been absent from work as a result of depression. Neither the employee nor his doctor is able to suggest any adjustments that could be made. Nevertheless the employer should still consider whether any adjustments, such as working from home for a time or working less hours, would be reasonable.

- 5.25 Effective and practicable adjustments for disabled people often involve little or no cost or disruption and are therefore very likely to be reasonable for an employer to have to make. Even if an adjustment has a significant cost associated with it, it may still be cost-effective in overall terms and so may be a reasonable adjustment to make. Many adjustments do not involve making physical changes to premises. However, where such changes do need to be made, employers may need to take account of the considerations explained in Chapter 12 which deals with issues about making alterations to premises.
- 5.26 If making a particular adjustment would increase the risks to the health and safety of any person (including the disabled person in question) then this is a relevant factor in deciding whether it is reasonable to make that adjustment. Suitable and sufficient risk assessments, such as those carried out for the purposes of the Management of Health and Safety at Work Regulations (NI) 2000, should be used to help determine whether such risks are likely to arise.
 - **s 18B(1)** 5.27 The Act lists a number of factors which may, in particular, have a bearing on whether it will be reasonable for the employer to have to make a particular adjustment. These factors make a useful checklist,

92

particularly when considering more substantial adjustments. The effectiveness and practicability of a particular adjustment might be considered first. If it is practicable and effective, the financial aspects might be looked at as a whole - the cost of the adjustment and resources available to fund it. Other factors might also have a bearing. The factors in the Act are listed below:

- the effectiveness of the step in preventing the disadvantage;
- the practicability of the step;
- the financial and other costs of the adjustment and the extent of any disruption caused;
- the extent of the employer's financial or other resources;
- the availability to the employer of financial or other assistance to help make an adjustment;
- the nature of the employer's activities, and the size of its undertaking;
- in relation to private households, the extent to which taking the step would disrupt the household or disturb any person residing there.

The effectiveness of the step in preventing the disadvantage

5.28 It is unlikely to be reasonable for an employer to have to make an adjustment involving little benefit to the disabled person.

A disabled employee cannot physically access the stationery cupboard at work. It is unlikely to be reasonable for the employer to have to make the cupboard accessible, unless distribution of stationery was a significant part of the employee's job. 5.29 However, an adjustment which, taken alone, is of marginal benefit, may be one of several adjustments which, when looked at together, would be effective. In that case, it is likely to be reasonable to have to make it.

The practicability of the step

5.30 It is more likely to be reasonable for an employer to have to take a step which is easy to take than one which is difficult. In some circumstances it may be reasonable to have to take a step, even though it is difficult.

> It might be impracticable for an employer who needs to appoint an employee urgently to have to wait for an adjustment to be made to an entrance. How long it might be reasonable for the employer to have to wait would depend on the circumstances. However, it might be possible to make a temporary adjustment in the meantime, such as using another, less convenient entrance.

The financial and other costs of the adjustment and the extent of any disruption caused

5.31 If an adjustment costs little or nothing to make and is not disruptive, it would be reasonable unless some other factor (such as practicability or effectiveness) made it unreasonable. The costs to be taken into account include those for staff and other resources. The significance of the cost of a step may depend in part on what the employer might otherwise spend in the circumstances. In assessing the likely costs of making an adjustment, the availability of external funding (such as that provided by Access to Work) should be taken into account.

It would be reasonable for an employer to have to spend at least as much on an adjustment to enable the retention of a disabled person - including any retraining - as might be spent on recruiting and training a replacement.

- 5.32 The significance of the cost of a step may also depend in part on the value of the employee's experience and expertise to the employer. However, these factors and the factors listed at paragraph 5.33 below, are not likely to be relevant where the adjustment costs little or nothing and is otherwise reasonable to make.
- 5.33 Examples of the factors that might be considered as relating to the value of an employee would include:
 - the amount of resources (such as training) invested in the individual by the employer;
 - the employee's length of service;
 - the employee's level of skill and knowledge;
 - the level of the employee's pay.

Employers should ensure when applying any of the above factors or the factors mentioned in paragraph 5.32 that they do not indirectly discriminate on any of the other grounds protected by the equality legislation.

5.34 It is more likely to be reasonable for an employer to have to make an adjustment with significant costs for an employee who is likely to be in the job for some time than for a temporary employee. 5.35 An employer is more likely to have to make an adjustment which might cause only minor inconvenience to other employees or the employer than one which might unavoidably prevent other employees from doing their job, or cause other significant disruption.

The extent of the employer's financial or other resources

5.36 It is more likely to be reasonable for an employer with substantial financial resources to have to make an adjustment with a significant cost, than for an employer with fewer resources. The resources in practice available to the employer as a whole should be taken into account as well as other calls on those resources. For larger employers, it is good practice to have a specific budget for reasonable adjustments - but limitations on the size of any such budget does not mean that an employer does not have duties to its disabled employees. The reasonableness of an adjustment will depend not only on the resources in practice available for the adjustment but also on all other relevant factors (such as effectiveness and practicability).

If a shop is part of a retail chain, the total resources of that business would be taken into account when assessing whether an adjustment is reasonable. Competing demands on those resources will also be taken into account.

5.37 It is more likely to be reasonable for an employer with a substantial number of staff to have to make certain adjustments, than for a smaller employer.

It would generally be reasonable for an employer with a large number of staff to make significant efforts to reallocate duties or identify a suitable alternative post or provide supervision from existing staff. It may also be reasonable for a small company to make these adjustments but not if it involved disproportionate effort.

The availability to the employer of financial or other assistance to help make an adjustment

5.38 The availability of outside help (such as advice and assistance from Access to Work) may well be a relevant factor.

A small employer, in recruiting a disabled person, finds that the only feasible adjustment is too costly for it alone. However, if assistance is available e.g. from the Access to Work scheme or a voluntary body, it may well be reasonable for the employer to make the adjustment.

5.39 A disabled person is not required to contribute to the cost of a reasonable adjustment. However, if a disabled person has a particular piece of special or adapted equipment which s/he is prepared to use for work, this might make it reasonable for the employer to have to take some other step (as well as allowing the use of the equipment).

An employer requires its employees to use company cars for all business travel. One employee's disability means she would have to use an adapted car or an alternative form of transport. If she has an adapted car of her own which she is willing to use on business, it might well be reasonable for the employer to allow this and pay her an allowance to cover the cost of doing so, even if it might not have been reasonable for it to provide an adapted company car (because of the additional expense), or to pay an allowance to cover alternative travel arrangements in the absence of an adapted car. This would be a reasonable step to take because it would be cost-effective for the employer, easy to implement and would remove the disadvantage to the disabled employee immediately.

A disabled person employed as a computer engineer uses a piece of communications equipment obtained through the Access to Work scheme. The disabled person's employer pays the cost of repair when it breaks down.

The nature of the employer's activities, and the size of its undertaking

5.40 As explained in paragraphs 3.9 and 3.10, Part 2 now applies to all employers (except for the armed forces), irrespective of their size. However, the size of an employer's undertaking and the nature of its activities may be relevant in determining the reasonableness of a particular step. A small manufacturing company making chairs employs a craftsperson who becomes disabled. The disabled employee can no longer make chairs, even after adjustments have been considered. The only jobs available in the company are production-based as the company owner personally carries out all other functions, such as marketing and running the office. Given the nature of the business, it is not likely to be reasonable for the employer to provide an alternative job for the employee.

In contrast, a business of the same size which designs, manufactures and retails games is likely to have a wide range of jobs. In these circumstances, if an employee was no longer able to work in a production role it might be reasonable for the employer to provide an alternative job.

A sales assistant in a small shop who has mental ill health requests time off each week to attend a psychotherapy appointment. It would be more likely to be reasonable for the employer to agree to this if there were other sales assistants who could cover for the absence.

In relation to private households, the extent to which taking the step would disrupt the household or disturb any person residing there

5.41 The duty to make reasonable adjustments may apply in respect of a disabled person who works in private households. However, even if the financial cost would be minimal, it may not be reasonable to take a particular

step if doing so would cause disruption to the household or disturbance to people who live there.

A man employs a deaf cleaner in his house. They communicate with each other by writing notes. It is likely to be a reasonable adjustment for him to communicate with the cleaner in this way.

A person with a severe dust allergy applies for the position of nanny. When interviewed, it becomes apparent that the nanny can only work in an environment which is dust free. The prospective employers take the view that this would be too disruptive to their home life. It is unlikely to be reasonable for them to have to make the adjustments needed to employ this person.

Other factors

- 5.42 Although the Act does not mention any further factors, others may be relevant depending on the circumstances. For example:
 - effect on other employees;

A disabled person wants to work in a cold office as heat aggravates his skin condition. This is not reasonable in the open plan office where he works because it would be uncomfortable for other employees to have to work in these conditions. Moving him to a small office on his own may be a reasonable adjustment in these circumstances. A disabled employee working as a baker in a small bakery wants to avoid night shifts because sleep pattern disturbances trigger migraines. Other employees would also like to avoid night shifts, for reasons unrelated to disability. This objection by other employees is unlikely to be a relevant factor for the employer in considering whether it is reasonable to allow the disabled employee's request.

adjustments made for other disabled employees;

If an employer has a number of staff with mobility problems this may mean that it would be reasonable to make significant structural changes to the workplace.

 the extent to which the disabled person is willing to co-operate;

An employee with a mobility impairment works in a team located on an upper floor to which there is no access by lift. Getting there is very tiring for the employee, and the employer could easily make a more accessible location available for him. Following a workplace assessment the employer decides to move the employee to the alternative location, but the employee refuses to co-operate. If there is no other adjustment that the employer can reasonably make, it does not have to do any more.

Can failure to make a reasonable adjustment ever be justified?

- **s 3A(2)** 5.43 The Act does not permit an employer to justify a failure to comply with a duty to make a reasonable adjustment.
 - 5.44 Clearly, however, an employer will only breach such a duty if the adjustment in question is one which it is reasonable for it to have to make. So, where the duty applies, it is the question of 'reasonableness' which alone determines whether the adjustment has to be made.



Introduction

- 6.1 Most conduct which is potentially unlawful under Part 2 of the Act cannot be justified. Conduct which amounts to:
 - direct discrimination;
 - failure to comply with a duty to make a reasonable adjustment;
 - victimisation;
 - harassment;
 - instructions or pressure to discriminate; or
 - aiding an unlawful act;

cannot be justified. It is unlawful irrespective of the reason or motive for it.

When does the Act permit justification?

6.2 Paragraph 4.27 explains that one of the forms of discrimination which is unlawful under Part 2 is disability-related discrimination. However, an employer's conduct towards a disabled person does not amount to disability-related discrimination if it can be justified. This chapter explains the limited circumstances in which this may happen.

s 3A(3) & (4)

6

6.3

Where less favourable treatment of a disabled person is capable of being justified (that is, where it is **not** direct discrimination), the Act says that it will, in fact, be justified if, but only if, the reason for the treatment is both **material** to the circumstances of the particular case **and substantial**. This is an objective test. 'Material' means that there must be a reasonably strong connection between the reason given for the treatment and the circumstances of the particular case. 'Substantial' means, in the context of justification, that the reason must carry real weight and be of substance.

A person who has severe back pain and is unable to bend is rejected for a job as a carpet fitter. The employer rejects the application because the applicant cannot carry out the essential requirement of the job, which is to fit carpets. This would be lawful as the reason the applicant is rejected is a substantial one and is clearly material to the circumstances.

6.4 In certain circumstances, the existence of a material and substantial reason for disability-related less favourable treatment is not enough to justify that treatment. This is the case where an employer is also under a duty to make reasonable adjustments in relation to the disabled person but fails to comply with that duty.

s 3A(6) 6.5 In those circumstances, it is necessary to consider not only whether there is a material and substantial reason for the less favourable treatment, but also whether the treatment would still have been justified even if the employer had complied with its duty to make reasonable adjustments. In effect, it is necessary to ask the question 'would a reasonable adjustment have made any difference?' If a reasonable adjustment would have

made a difference to the reason that is being used to justify the treatment, then the less favourable treatment cannot be justified.

An applicant for a typing job appears not to be the best person for the job, but only because of a slow typing speed. The applicant's hands have arthritis and this affects the applicant's typing speed. If a reasonable adjustment - perhaps an adapted keyboard - would overcome this, the applicant's typing speed would not in itself be a substantial reason for not employing the applicant. Therefore the employer would be unlawfully discriminating if, on account of the slow typing speed, it did not employ the applicant or provide that adjustment.

- 6.6 In relation to disability-related discrimination, the fact that an employer has failed to comply with a duty to make a reasonable adjustment means that the sequence of events for justifying disability-related less favourable treatment is as follows:
 - The disabled person proves facts from which it could be inferred in the absence of an adequate explanation that:
 - a. for a reason related to his or her disability, s/he has been treated less favourably than a person to whom that reason does not apply, has been, or would be, treated; and
 - b. a duty to make a reasonable adjustment has arisen in respect of him or her and the employer has failed to comply with it.

- The employer will be found to have discriminated unless it proves that:
 - a. the reason for the treatment is both material to the circumstances of the particular case and substantial; and
 - b. the reason would still have applied if the reasonable adjustment had been made.

Can health and safety concerns justify less favourable treatment?

6.7 Stereotypical assumptions about the health and safety implications of disability should be avoided, both in general terms and in relation to particular types of disability. Indeed, less favourable treatment which is based on such assumptions may itself amount to direct discrimination - which is incapable of justification (see paragraph 4.5). The fact that a person has a disability does not necessarily mean that s/he represents an additional risk to health and safety.

An employer has a policy of not employing anyone with diabetes because it believes that people with this condition are a health and safety risk. A person with diabetes applies to work for this employer and is turned down on the basis of her disability, without regard to her personal circumstances. A stereotypical assumption has been made which is likely to amount to direct discrimination and is therefore unlawful.

- 6.8 Under health and safety law it is the duty of every employer to ensure, so far as is reasonably practicable, the health, safety and welfare at work of all employees. Part of this duty is a requirement for all employers to assess the risks to the health and safety of all employees in the workplace and then to put in place measures that reduce the risks to as low a level as can reasonably be achieved. Genuine concerns about the health and safety of anybody (including a disabled employee) may be relevant when seeking to establish that disability-related less favourable treatment of a disabled person is justified. However, it is important to remember that health and safety law does not require employers to remove all conceivable risk, but to ensure that risk is properly appreciated, understood and managed. Further information can be obtained from the Health and Safety Executive for Northern Ireland (see Appendix C for details).
- 6.9 It is the employer who must decide what action to take in response to concerns about health and safety. However, when an employer has reason to think that the effects of a person's disability may give rise to an issue about health and safety, it is prudent for it to have a new risk assessment carried out by a suitably qualified person. This is because:
 - If an employer treats a disabled person less favourably merely on the basis of generalised assumptions about the health and safety implications of having a disability, such treatment may itself amount to direct discrimination - which is incapable of justification.
 - Even where there is no direct discrimination, an employer which treats a disabled person less favourably without having a suitable and sufficient risk assessment carried out is unlikely to be able to show that its concerns about health and safety justify the less favourable treatment.

A pilot develops a heart condition and as a result is asked by his employer to undertake a risk assessment to be carried out by an appropriate consultant. This is likely to be justifiable.

6.10 Nevertheless, an employer should not subject a disabled person to a risk assessment if this is not merited by the particular circumstances of the case.

A person with a learning disability has been working as a shop assistant for many years, without any problems. A new manager is appointed who insists that a risk assessment is carried out for the disabled shop assistant but not for all the other shop assistants. This is unlikely to be warranted - and indeed it is likely to amount to direct discrimination.

- 6.11 A risk assessment must be suitable and sufficient. It should identify the risks associated with a work activity, taking account of any reasonable adjustments put in place for the disabled person, and should be specific for the individual carrying out a particular task. It is therefore unlikely that an employer which has a **general** policy of treating people with certain disabilities (such as epilepsy, diabetes or mental ill health) less favourably than other people will be able to justify doing so - even if that policy is in accordance with the advice of an occupational health adviser.
- 6.12 A 'blanket' policy of this nature will usually be unlawful. This is because it is likely to amount to direct discrimination (which cannot ever be justified) or to disability-related less favourable treatment which is not

justifiable in the circumstances - i.e. disability-related discrimination (see paragraphs 7.8 and 7.9).

6.13 Reasonable adjustments made by an employer may remove or reduce health and safety risks related to a person's disability. A suitable and sufficient assessment of such risks therefore needs to take account of the impact which making any reasonable adjustments would have. If a risk assessment is not conducted on this basis, then an employer is unlikely to be able to show that its concerns about health and safety justify less favourable treatment of the disabled person.

Can medical information justify less favourable treatment?

Consideration of medical information (such as a doctor's 6.14 report or the answers to a medical questionnaire) is likely to form part of an assessment of health and safety risks. In most cases, however, having a disability does not adversely affect a person's general health. In other cases, its effect on a person's health may fluctuate. Although medical information about a disability may justify an adverse employment decision (such as a decision to dismiss or not to promote), it will not do so if there is no effect on the person's ability to do the job (or if any effect is less than substantial), no matter how great the effects of the disability are in other ways. Indeed, less favourable treatment of a disabled person in a case where his or her disability has no effect on his or her ability to do the job may well amount to direct discrimination - which is incapable of being justified.

An employer requires all candidates for a job as a solicitor in a law firm to complete a medical questionnaire. Medical information about one candidate shows that she has a degenerative condition which is likely to affect her ability to walk. This is not relevant to her ability to do the job as the job in question is of a sedentary nature. It would be unlawful for the employer to reject her on the ground of her disability as her disability is irrelevant to her ability to do the job. This would amount to direct discrimination.

The same employer is looking for a solicitor to work on a specific project for two years. A medical questionnaire shows that a candidate has a degenerative condition which could mean that he would not be able to work for that long. As a result of this, further medical evidence is requested from his doctor and this confirms that he would not be able to work for two years. It is likely to be lawful to reject this candidate if the two-year requirement is justified in terms of the work and if there are no reasonable adjustments that could be made.

6.15 In addition, where medical information is available, employers must weigh it up in the context of the actual job, and the capabilities of the individual. An employer should also consider whether reasonable adjustments could be made in order to overcome any problems which may have been identified as a result of the medical information. It should not be taken for granted that the person who provides the medical information will be aware that employers have a duty to make reasonable adjustments, what these adjustments might be, or of the relevant working arrangements. It is good practice, therefore, to ensure that medical advisers are made aware of these matters. Information provided by a medical adviser should only be relied on if the adviser has the appropriate knowledge and expertise.

An administrative assistant is diagnosed with Repetitive Strain Injury. An occupational health adviser recommends that the administrative assistant cannot carry on in this job due to the disability. The employer has another member of staff who uses voice recognition software, and considers this technology may be of relevance. The employer asks the adviser to review the recommendation taking this into account. The adviser revises the recommendation and concludes that, with appropriate software, the administrative assistant can continue in the role.

6.16 In any event, although medical evidence may generally be considered as an 'expert contribution', it should not ordinarily be the sole factor influencing an employer's decision on employment related matters. The views of the disabled person (about his or her own capabilities and possible adjustments) should also be sought. In addition, and subject to the considerations about confidentiality explained in paragraphs 8.21 to 8.23, other contributions could come from the disabled person's line manager (about the nature of the job and possible adjustments). It may also be possible to seek help from disability organisations or from the Disablement Advisory Service through JobCentres and Jobs and Benefits Offices, who have staff trained to advise about disability issues in the workplace. Ultimately, it is for the employer - and not the medical adviser - to take decisions as to whether, for example, to reject a job applicant or to maintain a disabled person's employment.

An employer receives advice from an occupational health adviser stating simply that an employee is 'unfit for work'. In spite of this the employer must consider whether there are reasonable adjustments which should be made.

7 Discrimination in the recruitment of employees

Introduction

- 7.1 It has already been explained (at paragraph 3.18) that it is unlawful for an employer to discriminate against a disabled person:
 - in the arrangements made for determining who should be offered employment;
 - in the terms on which the disabled person is offered employment; or
 - by refusing to offer, or deliberately not offering, the disabled person employment.
- 7.2 This chapter examines these principles in more detail. In order to do so, it is necessary to look at the various stages of the recruitment process, from specifying the job and advertising the vacancy, to the process of assessing candidates, interview and selection.
- 7.3 Although the following paragraphs refer only to the recruitment of employees, it should be remembered that the same principles apply to the recruitment of people to occupations covered by the Act's provisions on employment and occupation. Any variations in the way the Act applies to such occupations are explained in Chapter 9.

General considerations

Recruiting the best person for the job

- 7.4 Before considering the recruitment process itself it should be noted that, although an employer must not discriminate against a disabled candidate, there is no requirement (aside from the duty to make reasonable adjustments) to treat a disabled person more favourably than it treats or would treat others. An employer will have to assess an applicant's merits as they would be if any reasonable adjustments required under the Act had been made. If, after allowing for those adjustments, a disabled person would not be the best person for the job, the employer does not have to recruit that person.
- 7.5 On the other hand, the Act does not prevent posts being advertised as open only to disabled applicants, or to an applicant being preferred for the job because of his or her disability.

A note about 'arrangements' for determining who should be offered employment

7.6 Although the statutory provisions specifically deal with recruitment in relation to employment as defined by the Act (see paragraph 3.8), the meaning of 'arrangements' - that is, arrangements for determining who should be offered employment - is wide. Such arrangements are not confined to those which an employer makes in deciding who should be offered a specific job, but also include arrangements for deciding who should be offered employment more generally. Thus, for example, participation in a pre-employment training programme could be 'an arrangement' if its completion is a necessary step along the road to gaining an offer of employment.

Specifying the job

How does the Act affect the way in which a job description or person specification should be prepared?

7.7 The inclusion of unnecessary or marginal requirements in a job description or person specification can lead to discrimination. All job criteria should be relevant and objectively justifiable.

> An employer stipulates that employees must be 'active and energetic', when in fact the job in question is largely sedentary in nature. This requirement could unjustifiably exclude some people whose disabilities result in them getting tired more easily than others.

An employer specifies that a driving licence is required for a job which involves limited travel. An applicant for the job has no driving licence because of the particular effects in his case of cerebral palsy. He is otherwise the best candidate for that job. He could easily and cheaply do the travelling involved other than by driving and it is likely to be a reasonable adjustment for his employer to let him do so. It would be discriminatory to insist on the specification and reject his application solely because he has no driving licence.

An employer stipulates that employees must be 'good team players', when in fact the job in question does not involve working in a team. This requirement could unjustifiably exclude some people who have difficulty communicating, such as some people with autism. 7.8 Blanket exclusions (i.e. exclusions which do not take account of individual circumstances) can also lead to discrimination. Indeed, such exclusions are likely to amount to direct discrimination, and so be incapable of justification (see paragraph 4.5).

An employer excludes people with epilepsy from all driving jobs. One of the jobs, in practice, only requires a standard licence and standard insurance cover. If, as a result, someone with epilepsy, who has such a licence and can obtain such insurance cover, is turned down for the job, the employer will have discriminated unlawfully in excluding that person from consideration.

An employer stipulates that candidates for a job must not have a history of mental illness, believing that such candidates will have poor attendance. The employer rejects an applicant solely because he has had a mental illness, without checking the individual's actual attendance record. This will amount to discrimination and will be unlawful.

An employer stipulates that anyone with an infectious disease cannot work in the food preparation area. It refuses to employ applicants with AIDS in this area, believing them to be a health and safety risk. Whether or not the employer has a written policy to this effect, this action will amount to discrimination, as the employer has not considered the actual circumstances of the case. 7.9 In addition, stating that a certain personal, medical or health-related characteristic is essential or desirable can lead to discrimination if the characteristic is not necessary for the performance of the job. An employer would therefore need to ensure that any such requirements were genuinely essential to the job, and that it would not be reasonable to waive them in any individual case.

A television company requires all television engineers to have a high standard of hearing and vision. A woman with a hearing impairment is turned down for a job in the graphic design department because she does not pass a hearing test. If this standard of hearing is not necessary in order to do the particular job she applied for, the employer will have unlawfully discriminated against her by failing to make a reasonable adjustment to its policy of requiring job applicants to pass the test.

7.10 Likewise, although an employer is entitled to specify that applicants for a job must have certain qualifications, it will have to justify rejecting a disabled person for lacking a qualification if the reason why the disabled person lacks it is related to his or her disability. Justification will involve showing that the qualification is relevant and significant in terms of the particular job and the particular applicant, and that there is no reasonable adjustment which would change this. In some circumstances it might be feasible to reassign the duties to which the qualification relates, or to waive the requirement for the qualification if this particular applicant has alternative evidence of the necessary level of competence. An employer seeking someone to work in an administrative post specifies that candidates must have the relevant NVQ Level 4 qualification. If Level 4 fairly reflects the complex and varied nature and substantial personal responsibility of the work, and these aspects of the job cannot reasonably be altered, the employer will be able to justify rejecting disabled applicants who have only been able to reach Level 3 because of their disability and who cannot show the relevant level of competence by other means.

An employer specifies that two GCSEs are required for a certain post. This is to show that a candidate has the general level of ability required. No particular subjects are specified. An applicant whose dyslexia prevented her from passing written examinations cannot meet this requirement. The employer would be unable to justify rejecting her on this account alone if she could show in some other way that she had the expertise called for in the post.

Advertising the vacancy

Can a job advertisement encourage applications from disabled people?

7.11 The Act does not prevent a job advertisement saying that the employer would welcome applications from disabled people. This would be a positive and public statement of the employer's policy. More information about good practice in relation to attracting disabled job applicants is given at paragraphs 2.46 and 2.47.

What about discriminatory job advertisements?

- 7.12 The Act says that, when advertising a job vacancy, it is unlawful for the person offering the job to publish an advertisement (or cause it to be published) which indicates, or might reasonably be understood to indicate:
 - that the success of a person's application for the job may depend to any extent on his or her not having any disability, or any particular disability; or
 - that the person determining the application is reluctant to make reasonable adjustments.

An employer advertises a work placement for an office worker, stating 'We are sorry but because our offices are on the first floor, they are not accessible to disabled people'. This is likely to be unlawful. It would be better for the advertisement to state 'although our offices are on the first floor, we welcome applications from disabled people and are willing to make reasonable adjustments.'

An advertisement for a manager in office supplies stipulates that a driving licence is required. The post itself does not involve significant amounts of driving, and reasonable adjustments to this element of the job would be possible. However, the advertisement implies that the employer is unwilling to make such reasonable adjustments, for example, by allowing travel by taxi or allocating the driving to someone else. This is likely to be unlawful. s 16B(1)

S	16B(2)	-
	(4)	

7.13 The Act applies to every form of advertisement or notice, whether to the public or not, for any employment, promotion or transfer of employment. However, an advertisement may still be lawful even if it does indicate that having a particular disability will adversely affect an applicant's prospects of success. This will be the case where, for example, because of the nature of the job in question, the employer is entitled to take the effects of the disability into account when assessing the suitability of applicants.

It would be lawful for a company specialising in inner city bicycle courier services to advertise for couriers who 'must be able to ride a bicycle.'

s 17B(1) 7.14 The Act does not give individual job applicants the right to take legal action in respect of discriminatory advertisements. Such action may only be taken by the Equality Commission (see paragraphs 13.27 to 13.29).

An employer states in an advertisement for an office worker, 'Sorry, but gaining access to our building is difficult for disabled people.' The Equality Commission could take proceedings on the grounds that this constitutes a discriminatory advertisement.

7.15 In addition, it should be noted that the content of the advertisement could be taken into account by an industrial tribunal in determining a claim brought by a disabled person under the Act, who had applied for the job and was subsequently rejected.

A disabled person who walks with the aid of crutches applies for the job mentioned in the previous example and is turned down. He could ask the industrial tribunal to take the content of the advertisement into account when determining whether he was discriminated against.

Application forms and information

Does an employer have to provide information about jobs in accessible formats?

7.16 Where an employer provides information about a job, it is likely to be a reasonable adjustment for it to provide on request information in a format that is accessible to a disabled applicant - particularly if the employer's information systems, and the time available before the new employee is needed, mean it can easily be done. Accessible formats include email, Braille, Easy Read, large print, audio tape and computer disc. A disabled person's requirements will depend not just upon his or her impairment, but on other factors too. For example, many blind people do not read Braille but prefer to receive information by email or on audio tape.

A person whom the employer knows to be disabled asks to be given information about a job in a format that is accessible to her. It is likely to be a reasonable adjustment for the employer to provide the information in an accessible format. An employer advertises vacancies on its website. The website is not accessible to disabled people who use particular types of software on their computers. A man with a visual impairment, who uses 'screen reader' software on his computer and has notified the employer that he would like to work for it, cannot read the job vacancies on the employer's website. It is likely to be unlawful for the employer to refuse to make its website accessible to the disabled man, unless it is prepared to provide him with the same information in an accessible format.

Does an employer have to accept applications in accessible formats?

7.17 Where an employer invites applications by completing and returning an application form, it is likely to be a reasonable adjustment for it to accept applications which contain the necessary information in accessible formats. However, a disabled person might not have a right to submit an application in his or her preferred format (such as Braille) if s/he would not be substantially disadvantaged by submitting it in some other format (such as email) which the employer would find easier to access. Where applications are invited by completing and returning a form on-line, that form should be accessible to disabled people (or an accessible alternative should be provided).

Due to his disability, a candidate asks to submit an application in a particular format, different from the one specified for candidates in general (e.g. on audio tape). It is likely to be a reasonable adjustment for the employer to allow this.

122

7.18 Whether or not an application is submitted in an accessible format, employers and their staff or agents must not discriminate against disabled people in the way that they deal with their applications.

Selection, assessment and interview arrangements

When must an employer make adjustments to its selection, assessment and interview arrangements?

7.19 An employer is not required to make changes in anticipation of applications from disabled people in general, although it would be good practice to do so. It is only if the employer knows or could reasonably be expected to know that a particular disabled person is, or may be, applying and is likely to be substantially disadvantaged by the employer's premises or arrangements, that the employer may have to make changes.

When should an employer short-list a disabled person for interview?

- 7.20 Many employers operate, as a matter of good practice, a guaranteed interview scheme, under which a disabled candidate who wishes to use the scheme will be short-listed for interview automatically if s/he demonstrates that s/he meets the minimum criteria for getting the job.
- 7.21 Regardless of whether an employer operates a guaranteed interview scheme, it will need to consider whether it should make reasonable adjustments when short-listing for interview. This will be the case if an employer knows or ought to know that an applicant has a

disability and is likely to be at a substantial disadvantage because of its recruitment arrangements or the premises in which any interviews are held. In these circumstances, the employer should consider whether there is any reasonable adjustment which would remove the disadvantage. Any such adjustment should be taken into account when short-listing applicants. If the employer cannot make this judgment without more information, it would be discriminatory for it not to put the disabled person on the short-list for interview - if that is how it would normally seek additional information about candidates.

What adjustments might an employer have to make when arranging or conducting interviews?

7.22 Employers should plan ahead for interviews. Depending upon the circumstances, changes may need to be made to arrangements for interviews or to the way in which interviews are carried out.

> A hearing impaired candidate informs a potential employer that he can lip read but will need to be able to see the interviewer's face clearly. The interviewer ensures that her face is well lit, that she faces the applicant when speaking, that she speaks clearly and is prepared to repeat questions if the candidate does not understand her. These are likely to be reasonable adjustments for the employer to make.

An employer arranges for a sign language interpreter to attend an interview with a deaf candidate who uses sign language to communicate. The interviewer also allows extra time for the interview. (Communication support for interviews such as sign language interpreters may also be available through the Access to Work scheme - see Appendix C for contact details). These are likely to be reasonable adjustments for the employer to make.

An employer pays expenses to candidates who come for interview. A disabled candidate brings his support worker with him to the interview. The employer pays the expenses of the support worker as well. This is likely to be a reasonable adjustment to the usual policy of paying only the candidate's own expenses.

An employer allows a candidate who has a learning disability to bring a supportive person to an interview to assist when answering questions that are not part of the assessment itself. This is likely to be a reasonable adjustment to the selection process.

- 7.23 It is a good idea to give applicants the opportunity on an application form to indicate any relevant effects of a disability and to suggest adjustments to help overcome any disadvantage the disability may cause at interview. This could help the employer avoid discrimination in the interview and in considering the application, by clarifying whether any reasonable adjustments may be required. However, an employer must not assume that no adjustments need to be made simply because the applicant has not requested any (see paragraph 5.24).
- 7.24 The practical effects of an employer's duties may be different if a person whom the employer previously did

not know, and could not reasonably be expected to have known, to be disabled arrives for interview and is substantially disadvantaged because of the arrangements. The employer may still be under a duty to make a reasonable adjustment from the time that it first learns of the disability and the disadvantage. However, the employer might not be required to do as much as might have been the case if it had known (or if it ought to have known) in advance about the disability and its effects.

A job applicant does not tell an employer in advance that she uses a wheelchair and the employer does not know of her disability. On arriving for the interview she discovers that the room is not accessible. Although the employer could not have been expected to make the necessary changes in advance, it would be a reasonable adjustment to hold the interview in an alternative, accessible room if one was available without too much disruption or cost. Alternatively, it might be a reasonable adjustment to reschedule the interview if this was practicable.

What about aptitude or other tests in the recruitment process?

7.25 The Act does not prevent employers carrying out aptitude or other tests, including psychological tests. However, routine testing of all candidates may still discriminate against particular individuals or substantially disadvantage them. In those cases, the employer would need to revise the tests - or the way the results are assessed - to take account of a disabled candidate. This does not apply, however, where the nature and form of the test is necessary to assess a matter relevant to the job. The following are examples of adjustments which may be reasonable:

- allowing a disabled person extra time to complete the test if his or her disability is such that s/he would otherwise be substantially disadvantaged;
- permitting a disabled person the assistance of a reader or scribe during the test;
- accepting a lower 'pass rate' for a person whose disability inhibits performance in such a test.

The extent to which such adjustments might be required would depend on how closely the test is related to the job in question and what adjustments the employer might have to make if the applicant were given the job.

An employer sets a word processing test for candidates for a position as administrative officer. A person with repetitive strain injury (RSI) takes the test using voice-activated software, as this is how she would carry out the job if she were appointed. Permitting her to take the test in this way is likely to be a reasonable adjustment for the employer to make.

An employer sets candidates a short oral test. An applicant is disabled by a severe stammer, but only under stress. It is likely to be a reasonable adjustment to allow her more time to complete the test. Alternatively, it may be a reasonable adjustment to give the test in written form instead - though not if excellent oral communication skills are necessary for the job and assessing those skills was the purpose of the test. 7.26 However, employers would be well advised to seek professional advice in the light of individual circumstances before making adjustments to psychological or aptitude tests.

When can an employer ask questions about a disability?

7.27 The Act does not prohibit an employer from seeking information about a disability. However, the data protection legislation only allows sensitive personal data, such as information about a person's mental or physical health, to be obtained when certain conditions apply. These conditions are outlined in the legislation and explained in more detail in the 'Employment Practices Data Protection Code Part 1: Recruitment and Selection' (see Appendix C for details). Such information can, for example, be sought for equal opportunities monitoring purposes (see paragraphs 2.27 - 2.45).

Employers can also obtain such information in order to comply with any legal obligations in connection with employment. Employers can, for example, seek information in relation to a person's disability in order to ascertain whether any reasonable adjustments are required or about the effects of his or her disability where it is relevant to the person's ability to do the job. It is important to stress that disability-related questions must not be used to discriminate against a disabled person. An employer should only ask such questions if they are, or may be, relevant to the person's ability to do the job - after a reasonable adjustment(s), if necessary. An applicant with a visual impairment is asked at interview whether or not she was born with that condition. This is irrelevant to her ability to do the job and may upset the applicant, potentially preventing her from performing as well as she would otherwise have done. This is likely to be unlawful.

An applicant who is a wheelchair user is asked whether any changes may be needed to the workplace to accommodate him. This would not be unlawful.

- 7.28 Asking a basic question as to whether or not a person is disabled is unlikely to yield any useful information for the employer and may simply lead to confusion. The fact that such a question was asked might subsequently be used as evidence of discrimination. In addition, short-listing on the basis of an applicant's responses to a medical questionnaire may be discriminatory if the employer has not ascertained the likely effects of a disability on the applicant's ability to do the job, or whether reasonable adjustments would overcome any disadvantage it causes. Even where there are medical requirements which must be met, it is good practice for employers not to require job applicants to answer a medical questionnaire until after a conditional job offer has been made.
- 7.29 On the other hand, when inviting a job applicant to attend an interview, it is good practice for an employer to ask whether any adjustments might be needed to enable him or her to participate fully in the process, and what those adjustments might be.

An application form includes the statement 'Please let us know if you require any reasonable adjustments, due to disability, to enable you to attend an interview, or which you wish us to take into account when considering your application. Reasonable adjustments are things like sign language interpreters, altering the time of the interview, or making the interview room accessible for you. If you would like to discuss your disability requirements further, please contact the Human Resources manager'. This will not be discriminatory and is likely to help an employer comply with its duties under the Act.

7.30 In addition, once a decision has been made to appoint a disabled person, it is good practice for an employer to discuss reasonable adjustments with him or her before s/he starts work. It is recommended that an employer monitors disabled applicants, as part of an overall monitoring policy, although this information should be kept separately from an application form. More information about good practice in relation to monitoring is given in paragraphs 2.27 - 2.45.

Can a disabled person be required to have a medical examination?

7.31 Although the Act does not prevent an employer from asking a disabled person to have a medical examination, an employer will probably be discriminating unlawfully if, without justification, it insists on a medical check for a disabled person but not for others. The fact that a person has a disability is, in itself, unlikely to justify singling out that person to have a health check - although this might be justified in relation to some jobs. Paragraphs 6.14 to 6.16 explain the circumstances in which medical information may legitimately influence an employer's decision as to whether to offer a job to a disabled person. An employer requires all candidates for employment to have a medical examination. The employer would be entitled to apply that requirement to a disabled person who applies for employment.

An employer issues a health questionnaire to all job applicants and requires any successful job applicant who states they are disabled to undergo a medical examination. This is likely to be unlawful.

An employer issues a health questionnaire to all successful job applicants but does not require them to undergo a medical examination unless they have a condition which may be relevant to the job, or the working environment. A successful job applicant indicates that he has a disabling heart condition. It is likely that the employer would be justified in asking him to have a medical examination provided it is restricted to assessing the implications for the particular job in question.

Offers of employment

7.32 Terms and conditions of service should not discriminate against a disabled person. In general, an employer should not offer a job to a disabled person on terms which are less favourable than those which would be offered to other people.

A person with a history of depression is offered employment with a six month probationary period, even though other employees are only required to serve a three month probationary period. This will amount to direct discrimination and will be unlawful.

8 Discrimination against employees

Introduction

- 8.1 It has already been explained (at paragraph 3.19) that it is unlawful for an employer to discriminate against a disabled person whom it employs:
 - in the terms of employment which it affords him or her;
 - in the opportunities which it affords him or her for promotion, transfer, training or receiving any other benefit;
 - by refusing to afford him or her, or deliberately not affording him or her, any such opportunity; or
 - by dismissing him or her, or subjecting him or her to any other detriment.
- 8.2 This chapter examines these principles in more detail. In order to do so, it is necessary to look at various aspects of the employment relationship and its associated conditions and benefits, from terms and conditions of service to arrangements for induction and training, and from opportunities for promotion or transfer to arrangements for managing disability or ill health, and terminating employment.
- 8.3 Although the following paragraphs refer only to the treatment of employees, it should be remembered that the same principles apply to people in occupations covered by the Act's provisions on employment and occupation. Any variations in the way the Act applies to such occupations are explained in Chapter 9. In addition,

particular issues about discrimination in providing occupational pensions and group insurance services are considered in Chapter 10.

Terms and conditions of service

8.4 As stated at paragraph 7.32, terms and conditions of service should not discriminate against a disabled person. The employer should consider whether any reasonable adjustments need to be made to the terms and conditions which would otherwise apply.

An employer's terms and conditions state the hours an employee has to be in work. It might be a reasonable adjustment to change these hours for someone whose disability means that she has difficulty using public transport during rush hours.

- 8.5 Where the terms and conditions of employment include an element of performance-related pay, the employer must ensure that the way such pay arrangements operate does not discriminate against a disabled employee. If, on the ground of his or her disability, an employee is denied the opportunity to receive performance-related pay, this is likely to be direct discrimination. Even if less favourable treatment of an employee in relation to performance-related pay is not directly discriminatory, it will amount to disability-related discrimination unless the employer can show that it is justified.
- 8.6 If an employee has a disability which adversely affects his or her rate of output, the effect may be that s/he receives less under a performance-related pay scheme

than other employees. The employer must consider whether there are reasonable adjustments which would overcome this substantial disadvantage.

A disabled man with arthritis works in telephone sales and is paid commission on the value of his sales. He is advised to switch to new computer equipment because of a worsening of his impairment. This equipment slows his work down for a period of time while he gets used to it and consequently the value of his sales falls. It is likely to be a reasonable adjustment for his employer to continue to pay him his previous level of commission for the period in which he adjusts to the new equipment.

A disabled home-worker, who is paid a fixed rate for each item he produces, has a reduced output rate because he does not have the right equipment to do the job to the best of his ability. It is likely to be a reasonable adjustment for the employer to provide that equipment, possibly with funding or advice from the Access to Work scheme (see Appendix C for contact details), to improve the disabled worker's output and consequently his pay. A woman who has recently become disabled because of diabetes works for an employer that operates a performance related bonus scheme. When she has her annual appraisal, the woman is unable to demonstrate that she has met all her objectives for the year, unlike in previous years when she had in fact exceeded her objectives. The reason why the woman has not met her objectives this year is that she has been adjusting to her disability (attending hospital appointments, paying careful attention to her diet, taking regular breaks, etc). The disabled woman's employer is likely to be discriminating against her if, because she has not met her objectives for the year, it refuses to pay her a bonus.

Induction, training and development

s 4(2) & 8.7 Employers must not discriminate in their induction procedures. The employer may have to make adjustments to ensure a disabled person is introduced into a new working environment in a clearly structured and supported way, with an individually tailored induction programme if necessary.

A small manufacturing company usually hands out written copies of all its policies by way of induction to new employees and gives them half a day to read the documentation and to raise any questions with their line manager. A new employee has dyslexia and the employer arranges for her supervisor to spend a morning with her talking through the relevant policies. This is likely to be a reasonable adjustment.

136

An employer runs a one day induction course for new recruits. A recruit with a learning disability is put at a substantial disadvantage by the way the course is normally run. The employer is likely to have to make an alternative arrangement such as permitting the person to attend a longer course, or allowing someone to sit in on the course to provide support, assistance or encouragement to the disabled person.

8.8 In addition, employers must not discriminate in selection for training and must make reasonable adjustments in order to avoid disabled people being placed at a substantial disadvantage.

Instead of taking an informed decision, an employer wrongly assumes that a disabled person will be unwilling or unable to undertake demanding training or attend a residential training course. This is likely to amount to direct discrimination.

An employer may need to alter the time or the location of the training for someone with a mobility problem, make training manuals, slides or other visual media accessible to a visually impaired employee (perhaps by providing Braille versions or having them read out), or ensure that an induction loop is available for someone with a hearing impairment. s 4(2) & s 4A(1) An employee with a hearing impairment is selected for a post as a TV engineer. The employee attends the induction course which consists of a video and discussion. The video is not subtitled and thus the employee cannot participate fully in the induction. This is likely to be unlawful.

An employer refuses to allow a disabled employee to be coached for a theory examination relating to practical work which the employee is unable to do because of his disability. This is likely to be justified because the employee would never be suited for the area of work for which the coaching was designed, and a reasonable adjustment could not alter that position.

However, if the disabled employee in the above example required coaching to enable him to understand the requirements of the practical work because he would be managing staff carrying out the work, a decision not to provide coaching would be unlikely to be justified.

Benefits provided by employers

s 4(2) & s 4A(1) 8.9 Employers often provide a range of benefits to their staff. 'Benefits' include 'facilities' and 'services'. An employer must not discriminate in the way that it makes benefits available to disabled employees where those benefits are available to other employees. The employer must make any necessary reasonable adjustment to the way the benefits are provided. As explained in Chapter 10, an employer's duty to make reasonable adjustments now extends to the way it provides any benefits relating to occupational pension schemes or group insurance services.

8.10 Benefits might include canteens, meal vouchers, social clubs and other recreational activities, dedicated car parking spaces, discounts on products, bonuses, share options, hairdressing, clothes allowances, financial services, healthcare, medical assistance/insurance, transport to work, company car, education assistance, workplace nurseries, and rights to special leave. This is not an exhaustive list.

If physical features of a company's social club would inhibit a disabled person's access, it might be a reasonable adjustment for the employer to make suitable modifications.

An employer provides dedicated car parking spaces close to the workplace. A disabled employee finds it very difficult to get from the public car parks further away. It is likely to be a reasonable adjustment for the employer to allocate one of the dedicated spaces to that employee.

- 8.11 Some benefits may continue after employment has ended. An employer's duties under the Act extend to its former employees in respect of such benefits.
- 8.12 The provisions on employment and occupation in Part 2 of the Act do not normally apply to benefits which an employer provides to the public, or to a section of the (4)

s 16A

public which includes the disabled employee. This is because the provision of those benefits relates to the employer's activities as a service provider rather than as an employer. Such activities are usually subject to the duties in Part 3 of the Act instead. However, the Act's provisions on employment and occupation will apply if the benefit to employees is materially different (for example, at a discount) or is governed by the contract of employment, or relates to training.

A disabled employee of a supermarket chain is discriminated against for a reason related to his disability when buying goods as a customer of the supermarket. Even though he is an employee, he would have no claim under the Act's provisions on employment and occupation, because he is merely buying goods as a customer. However, if the discrimination related to the use of his employee's discount card, that would relate to his employment benefits and the Act's provisions on employment and occupation would apply.

Promotion and transfer

s 4(2) & s 4A(1)

8.13 Employers must ensure that arrangements for promoting staff, or for transferring staff between jobs, do not discriminate against disabled people. It is likely to be direct discrimination if a disabled employee is treated less favourably in this regard on the ground of his or her disability (see paragraph 4.5). If the treatment is not directly discriminatory, but is for a reason related to the disability, it will amount to disability-related discrimination unless the employer can show that it is justified. Employers must not discriminate in the practical

arrangements necessary to enable the promotion or transfer to take place or, of course, in the new job itself. Reasonable adjustments may need to be made to the various stages in the promotion or transfer process.

An employer does not consider an employee, who has lost the use of an arm, for promotion to assistant manager because the employer wrongly and unreasonably believes that the disability might prevent the employee performing competently in a managerial post. This is likely to be discrimination.

An employer interviewing a number of people for promotion is aware that one of the candidates has a hearing impairment, but does not find out whether he needs any special arrangements. The candidate requires a sign language interpreter. It is likely to be a reasonable adjustment to arrange for an interpreter.

An employee who has a disability resulting from a back injury is seeking a transfer to a different department. A minor aspect of the job the employee seeks involves assisting with unloading the weekly delivery van. The employee is unable to do this because of the disability. In assessing the employee's suitability for transfer, the employer should consider whether reallocating this duty to someone else would be a reasonable step to take. A disabled employee who has depression applies for a promotion to be a senior manager in her organisation. The manager responsible for filling this post says that the woman would be unsuitable because she would not be able to cope with the stress of the job. The manager has no evidence that this would be the case, and is merely making an assumption about her disability. This is likely to be unlawful.

8.14 As with other aspects of employment, employers will be better placed to ensure that promotion and transfer arrangements do not discriminate against disabled people if they have established and implemented policies and practices to counter discrimination generally (see paragraphs 2.8 - 2.25). These will help employers to check, for example, that qualifications required for promotion or transfer are justified for the job to be done, and to monitor other arrangements - such as systems for determining criteria for a particular job - so that they do not exclude disabled people who may have been unable to meet those criteria because of their disability but who would be capable of performing well in the job.

Managing disability or ill health

Retention of disabled employees

s 4(2) 8.15 An employer must not discriminate against an employee who becomes disabled, or who has a disability which worsens. Employers will often find that it is of benefit to their organisation to retain a disabled employee as this will prevent their knowledge and skills from being lost to the enterprise. In addition, the cost of retaining such an employee will frequently be less than the cost of recruiting and training a new member of staff.

- 8.16 If as a result of the disability an employer's arrangements or a physical feature of the employer's premises place the employee at a substantial disadvantage in doing his or her existing job, the employer must consider any reasonable adjustment that would resolve the difficulty. The nature of the adjustments which an employer may have to consider will depend on the circumstances of the case, but the following considerations will always be relevant:
 - The first consideration in making reasonable adjustments should be to enable the disabled employee to continue in his or her present job if at all possible.
 - The employer should consult the disabled person at appropriate stages about what his or her requirements are and, where the employee has a progressive condition, what effect the disability might have on future employment, so that reasonable adjustments may be planned.
 - In appropriate cases, the employer should also consider seeking expert advice (for example, from an occupational health adviser) on the extent of a disabled person's capabilities and on what might be done to change premises or working arrangements (for example, from an access auditor). Where an employee has been off work, particularly for a long period of time, a phased return might be appropriate.
 - If there are no reasonable adjustments which would enable the disabled employee to continue in his or her present job, the employer must consider whether there are suitable alternative positions to which s/he could be redeployed.

An employee becomes disabled after a back injury. After talking to her and taking specialist advice, the employer decides that there are no reasonable adjustments that could be made to her present role. The employer then considers whether there is another role that would be suitable and offers an alternative post to the employee, at roughly the same level of seniority. However, if after considering these steps, it is apparent that there is no alternative position on a similar salary and with similar conditions, a position on a lower salary or with worse conditions could be offered as a reasonable adjustment.

8.17 It may be possible to modify a job to accommodate an employee's changing needs. This might be by rearranging working methods or giving another employee certain minor tasks that the disabled person can no longer do, or by providing practical aids or adaptations to premises or equipment, or allowing the disabled person to work at different times or places from those with equivalent jobs. It may be that a reduction in working hours is appropriate.

A newly disabled employee may need time to readjust. In those circumstances, an employer might allow a trial period to assess whether the employee is able to cope with the current job, with or without modifications; the employee may initially be permitted to work from home, or with a gradual build-up to full time hours. It may be a reasonable adjustment for an employer to move a newly disabled person to a different post within the organisation if a suitable vacancy exists or is expected shortly.

Additional job coaching may be necessary to enable a disabled person to take on a new job.

8.18 The issue of job retention might also arise when an employee has a stable impairment but the nature of his or her job changes. In these circumstances an employer may also have a duty to make reasonable adjustments.

A woman with a learning disability is allocated to a new department because of a business reorganisation. She is given extra training to enable her to carry out her new role. This is likely to be a reasonable adjustment for the employer to make.

An employer installs a new software system for all computer users in the office. A disabled man who uses voice-activated software finds that this software is not compatible with the new office computer software. The employer provides him with new voice-activated software which is compatible with the office system. It also offers him appropriate training in the use of the two new software systems. These are likely to be reasonable adjustments for the employer to make.

Access to Work Scheme

- 8.19 In determining what adjustments to make to facilitate the employment of a disabled person, employers should have regard to the range of advice and assistance which is available from the Disablement Advisory Service in JobCentres and Jobs and Benefits Offices through the Access to Work scheme. The purpose of the scheme is to provide practical support to disabled people in, or entering, paid employment to help overcome work related obstacles resulting from disability. Access to Work may provide a grant towards these additional employment costs. However, there is no automatic right to support from the Access to Work scheme this is subject to an assessment when an application for support is made by the disabled person.
- 8.20 The Access to Work scheme may assist an employer to decide what steps to take. If financial assistance is available from the scheme, it may also make it reasonable for an employer to take certain steps which would otherwise be unreasonably expensive. However, Access to Work does not diminish any of an employer's duties under the Act. In particular:
 - the responsibility for making a reasonable adjustment is the responsibility of the employer even where Access to Work is involved in the provision of advice or funding in relation to the adjustment;
 - it is likely to be a reasonable step for the employer to assist a disabled person in making an application for assistance from Access to Work and to provide ongoing administrative support (by completing claim forms, for example); and
 - it may be unreasonable for an employer to decide not to make an adjustment if the decision is based on the cost of the adjustment but before it is known

if financial assistance for it is available from Access to Work or another source.

Further information about what the Access to Work scheme can offer, the scheme's eligibility requirements, and how to apply, is set out in Appendix C.

Confidential information

- 8.21 The extent to which an employer is entitled to let other staff know about an employee's disability will depend partly on the terms of employment. An employer could be discriminating against the employee by revealing such information if the employer would not reveal similar information about another person for an equally legitimate management purpose; or if the employer revealed such information without consulting the individual, instead of adopting the usual practice of talking to an employee before revealing personal information about him or her. Employers also need to be aware that they have obligations under the data protection legislation in respect of personal data.
- 8.22 However, as noted at paragraph 5.22, sometimes a reasonable adjustment will not work without the cooperation of other employees. In order to secure such co-operation, it may be necessary for the employer to tell one or more of a disabled person's colleagues (in confidence) about a disability which is not obvious. This may be limited to the disabled person's supervisor, or it may be appropriate to involve other colleagues, depending on the nature of the disability and the reason they need to know about it. In any event, an employee without his or her consent. A disabled person's refusal to give such consent may impact upon the effectiveness of the adjustments which the employer is able to make or its ability to make adjustments at all.

In order for a person with epilepsy to work safely in a particular factory, it may be necessary to advise fellow workers about how they can assist the disabled worker to manage the condition.

An office worker with cancer says that he does not want colleagues to know of his condition. As an adjustment he needs extra time away from work to receive treatment and to rest. Neither his colleagues nor his line manager need to be told the precise reasons for the extra leave but the latter will need to know that the adjustment is required in order to implement it effectively.

8.23 The Act does not prevent a disabled person keeping a disability confidential from an employer. However keeping a disability confidential is likely to mean that, unless the employer could reasonably be expected to know about it anyway, the employer will not be under a duty to make a reasonable adjustment. If a disabled person expects an employer to make a reasonable adjustment, s/he will need to provide the employer - or someone acting on its behalf - with sufficient information to carry out that adjustment.

An employee has symptomatic HIV. He prefers not to tell his employer of the condition. However, as the condition progresses, he finds it increasingly difficult to work the required number of hours in a week. Until he tells his employer of his condition - or the employer becomes aware of it (or could reasonably be expected to be aware of it) - the employer does not have to make a reasonable adjustment by changing his working hours to overcome the difficulty. However, once the employer is informed he may then have to make a reasonable adjustment.

Termination of employment

8.24 Where a disabled person is dismissed or is selected for redundancy or for compulsory early retirement (including compulsory ill-health retirement), the employer must ensure that the disabled person is not being discriminated against. It is likely to be direct discrimination if the dismissal or selection is made on the ground of his or her disability (see paragraph 4.5). If the dismissal or selection is not directly discriminatory, but is made for a reason related to the disability, it will amount to disability-related discrimination unless the employer can show that it is justified. The reason would also have to be one which could not be removed by any reasonable adjustment.

It would be justifiable to terminate the employment of an employee whose disability makes it impossible for him to perform the main functions of his job, if an adjustment such as a move to a vacant post elsewhere in the business is not reasonable for the employer to have to make.

8.25 When setting criteria for redundancy selection, employers should consider whether any proposed criterion would adversely impact upon a disabled employee. If so, it may be necessary for the employer to make reasonable adjustments. For example, it is likely to be a reasonable adjustment to discount disability-related sickness absence when assessing attendance as part of a redundancy selection scheme. Some employers use 'flexibility' as a selection criterion for redundancy (for example, willingness to re-locate or to work unpopular hours, or ability to carry out a wide variety of tasks). An employer should carefully consider how to apply this criterion to a disabled employee as it might be discriminatory.

8.26 Where the dismissal of a disabled person is being considered for a reason relating to that person's conduct, the employer should consider whether any reasonable adjustments need to be made to the disciplinary or dismissal process. In addition, if the conduct in question is related to the employee's disability that may be relevant in determining the sanction which it is appropriate to impose.

A man with learning disabilities asks if he can bring a friend, rather than a work colleague, to a disciplinary hearing. It is likely to be a reasonable adjustment for his employer to allow this.

A disabled employee shouts at his line manager in front of work colleagues and uses inappropriate language. The employer would usually consider dismissal as a sanction for such behaviour, but takes into account the fact that the employee was in great pain on the day in question because of his disability and instead issues a warning. This is likely to be a reasonable adjustment to make.

- **s 4(5)** 8.27 A disabled person will be taken to have been dismissed for the purposes of the Act if:
 - s/he is expressly dismissed; or
 - the period for which s/he is employed expires without his or her employment being immediately renewed on the same terms; or

s/he gives notice, or does some other act to bring his or her employment to an end in circumstances in which s/he is entitled to terminate it without notice by reason of the employer's conduct (this is known as 'constructive dismissal'). After the termination of employment 8.28 Where a disabled person's employment has come to an s 16A(3) end, the Act says that it will still be unlawful for his or her former employer: to discriminate against him or her by subjecting him or her to a detriment; or to subject him or her to harassment; if the discrimination or harassment arises out of the employment which has come to an end and is closely connected to it. A disabled person has to go back to his former workplace for a meeting to finalise handover of his customer files to his replacement. On arrival, he is verbally abused by one of his former colleagues. The abuse relates to his disability. This is unlawful. s 55 8.29 It is also unlawful to victimise a person (whether or not s/he is disabled) after his or her employment has come to an end (see paragraphs 4.33 to 4.36).

A disabled person gives the name of his former employer as a referee for a new job. The employer gives him a poor reference, referring to his disability as being a hindrance. This poor reference is an untrue reflection of the standard of work carried out by the disabled person. The poor reference was given because he brought a claim of disability discrimination against his former employer. Consequently it is unlawful.

s 16A(4)-(6)

- 8.30 An employer's duty to make reasonable adjustments may also apply in respect of a former employee who is a disabled person. This will be the case where:
 - the disabled person is placed at a substantial disadvantage in comparison with other former employees:
 - a. by a provision, practice or criterion applied by the employer to the disabled person in relation to any matter arising out of his or her former employment; or
 - b. by a physical feature of premises occupied by the employer; and
 - the employer either knows, or could reasonably be expected to know, that the former employee in question has a disability and is likely to be affected in this way.

A former employee with life-time membership of the works social club is no longer able to access the club because of a mobility impairment. Once the employer becomes aware of the problem, it should make the required reasonable adjustments.

- 8.31 The former employees with whom the position of the disabled person should be compared must be people who are not disabled, but who are former employees of the same employer. If it is not possible to identify an actual comparator for this purpose, then a hypothetical comparator may be used (see paragraph 4.18).
- 8.32 The principles relating to post-termination discrimination also apply to other relationships covered by Part 2 of the Act, and to the relationship between a provider of employment services and its former clients. These relationships are considered in the subsequent chapters.

Discrimination in occupation

Introduction

- 9.1 The preceding chapters focus on discrimination against employees. However, as noted at paragraph 3.11, Part 2 of the Act also applies to certain occupations. This chapter explains the relevant provisions.
- 9.2 The following paragraphs explain the provisions of Part 2 which focus specifically on the occupations in question. In other respects, however, the employment provisions of the Act apply in the usual way. So, where appropriate, regard should be had to the matters concerning recruitment and retention which are explained in Chapters 7 and 8; occupational pension schemes and group insurance services (Chapter 10); adjustments to premises (Chapter 12); and to the other relevant provisions (Chapter 13).

Discrimination against contract workers

What does the Act say about contract workers?

- 9.3 The Act says that it is unlawful for a 'principal' to discriminate against a disabled contract worker. As explained in paragraph 9.5 below, a 'principal' is a person who hires a contract worker under a contract made with the contract worker's employer. A principal cannot discriminate against a contract worker:
 - in the terms on which s/he is allowed to do 'contract work';

s 4B(1)

		 by not allowing him or her to do, or continue to do, contract work;
		 in the way it affords him or her access to, or by failing to afford him or her access to, benefits in relation to contract work; or
		 by subjecting him or her to any other detriment.
s 4B(2) s 55	9.4	The Act also says that it is unlawful for a principal to subject a disabled contract worker to harassment or to victimise any contract worker, whether or not s/he is disabled.
s 4B(9)	9.5	For the purposes of the Act, 'contract work' is work which an individual carries out for a person (a 'principal') who hires him or her under a contract made with his or her employer. Usually, that contract is made directly between the principal and the contract worker's employer, (which is usually an employment agency), but this is not always the case. Provided that there is an unbroken chain of contracts between the individual and the end- user of his or her services, that end-user is a principal for the purposes of the Act, and so the individual is a contract worker.
		A disabled person works for a computer software company which sometimes uses an employment agency to deploy staff to work on projects for other

companies. The employment agency arranges for the disabled person to work on a project for a large supermarket chain. In this case the 'principal' is the supermarket chain and the disabled person is the 'contract worker.'

9

9.6 The effect of the Act is that, where a person is a principal for the purposes of the Act, that person is treated as if s/he were, or would be, the actual employer of the disabled contract worker. Therefore, the same principles relating to discrimination apply to a principal as to an employer.

The employer of a labourer, who some years ago was disabled by clinical depression but has since recovered, proposes to supply the labourer to a contractor to work on a building site. Although the past disability is covered by the Act, the contractor's site manager refuses to accept the labourer because of this person's medical history. The contractor is likely to be acting unlawfully.

How does the duty to make reasonable adjustments apply in respect of disabled contract workers?

- 9.7 The duty to make reasonable adjustments applies to a principal in the same way as it applies to an employer.
- 9.8 However, in deciding whether any, and if so, what, adjustments would be reasonable for a principal to make, the period for which the disabled contract worker will work for the principal is important. It might well be unreasonable for a principal to have to make certain adjustments if the worker will be with the principal for only a short time.

An employment business enters into a contract with a firm of accountants to provide an assistant for two weeks to cover an unexpected absence. The employment business proposes the name of a person s 4B(6)

who is disabled. The disabled person finds it difficult, because of the disability, to travel during the rush hour and requests that the working hours be changed accordingly. It might not be reasonable for the firm to have to agree, given the short time in which to negotiate and implement the new hours.

Does this mean that a principal has the same duties to make adjustments as the employer of a disabled contract worker?

9.9 In the case of a disabled contract worker, both his or her employer and the principal to whom s/he is supplied may separately be under a duty to make reasonable adjustments.

> A travel agency hires a clerical worker from an employment business to fulfil a three month contract to file travel invoices during the busy summer holiday period. The contract worker is a wheelchair user, and is quite capable of doing the job if a few minor, temporary changes are made to the arrangement of furniture in the office. It is likely to be reasonable for the travel agency to make these adjustments.

s 4A(1)
9.10 A disabled contract worker's employer will have to make reasonable adjustments if the contract worker is substantially disadvantaged by the employer's own premises - or by a provision, criterion or practice applied by it. The Act says that the employer will also have to make reasonable adjustments where the contract worker is likely to be substantially disadvantaged by the premises - or by a provision, criterion or practice - of all or most of the principals to whom s/he might be supplied. This duty only arises if the contract worker is likely to be affected in this way each time s/he is

supplied for work. The employer would have to make reasonable adjustments to overcome the disadvantage wherever it might arise, rather than taking separate steps in relation to each principal.

An employment agency employs a secretary who is blind and who is supplied to other organisations for secretarial work. As a result of this disability, the secretary has difficulty in using the computer equipment at the offices of all or most of the principals to whom the secretary might be supplied. As a result, the secretary is placed at a substantial disadvantage. The employment agency provides the disabled secretary with a specially adapted portable computer and Braille keyboard.

9.11 A principal has the same duties to make reasonable adjustments as a disabled contract worker's employer, but does not have to make any adjustment which the employer should make. So, in effect, the principal is responsible for any additional reasonable adjustments which are necessary solely because of its own arrangements or premises.

> In the preceding example, a bank which hired the blind secretary may have to make changes which are necessary to ensure that the computer provided by the employment agency is compatible with the system which the bank is already using.

9.12 It would be reasonable for a principal and the employer of a contract worker to co-operate with any steps taken by the other to assist the disabled contract worker. It is s 4A(1) applied by s 4B(6) & (7) good practice for the principal and the employer to discuss what adjustments should be made, and who should make them.

The bank and the employment agency in the preceding examples would need to co-operate with each other so that, for example, the employment agency allows the bank to make any necessary adaptations to the equipment which the employment agency has provided to ensure its compatibility with the bank's existing systems.

What about Employment Support?

9.13 These arrangements also apply to the Employment Support Scheme operated through Disablement Employment Advisers in JobCentres and Jobs and Benefits Offices for particular groups of disabled people. The 'contractor' under the scheme is the equivalent of the contract worker's own employer, and the 'host employer' is the equivalent of the principal. An organisation can even be both the contractor and the host employer at the same time.

Discrimination against office holders

Who are office holders?

9.14 Examples of office holders include some company directors, judges, and chairpersons or members of non-departmental public bodies (for example, the Equality Commission). What the holders of such offices or posts have in common is that they are not regarded as 'employees' by the law, even though they may be similar

s 4C

to employees in providing services personally under the direction of another in return for remuneration. The Act gives specific protection to such office holders against discrimination if they are not otherwise protected under Part 2. This protection extends to applicants for such appointments and applies also to office holders appointed by, or on the recommendation of, the Government (including the devolved administration), or subject to its approval. However, it does not extend to the holders of political office.

What does the Act say about office holders?

- 9.15 The Act says that, in relation to an appointment to such an office or post, it is unlawful to discriminate against a disabled person:
 - in the arrangements which are made to determine who should be offered the appointment;
 - in the terms on which the appointment is offered; or
 - by refusing to offer him or her the appointment.
- 9.16 In addition, where the appointment is made on the Government's recommendation (or is subject to its approval) the Act says that it is unlawful to discriminate against a disabled person:
 - in the arrangements which are made to determine who should be recommended or approved; or
 - in making or refusing to make a recommendation, or in giving or refusing to give approval.

s 4D(2)

s 4D(1)

	s 4D(3)	9.17	A deaf woman who communicates using sign language applies for appointment as a member of a public body. Without interviewing her, the public body making the appointments writes to her saying that she would not be suitable as good communication skills are a requirement. This is likely to be unlawful.
			The Act also says that it is unlawful, in relation to a disabled person who has been appointed to such an
			office or post, to discriminate against him or her:
			 in the terms of the appointment;
			 in the opportunities which are afforded (or refused) for promotion, transfer, training or receiving any other benefit;
			 by terminating the appointment; or
			 by subjecting him or her to any other detriment in relation to the appointment.
	s 4D(4) s 55	9.18	In addition, the Act says that it is unlawful to subject a disabled person to harassment if s/he is an office holder, or if s/he is seeking or being considered for an appointment (or a related Government recommendation or approval). It is also unlawful to victimise such a person, whether s/he is disabled or not.
	s 4D(5)	9.19	The Act mirrors the position in respect of benefits to employees (see paragraphs 8.9 to 8.12).
	s 16B	9.20	The Act gives the Equality Commission power to take action in respect of a discriminatory advertisement for appointment to an office or post (see paragraphs 13.27

to 13.29). This mirrors the Equality Commission's power in respect of advertisements for employment (see paragraph 7.14).

How does the duty to make reasonable adjustments apply in respect of office holders?

9.21 Part 2 of the Act also requires reasonable adjustments to be made for disabled people holding relevant offices or posts, or seeking such appointments. The duty relates to any provision, criterion or practice applied by or on behalf of the relevant person (see paragraph 9.22), and to any physical feature of premises under the control of such a person where the functions of the office or post are performed.

A selection process is carried out to appoint the chair of a public health body. The best candidate is found to be a disabled person with a progressive condition who is not able to work full-time because of her disability. Whoever makes or recommends the appointment should consider whether it would be a reasonable adjustment to appoint the disabled person on a jobshare or part-time basis.

Who needs to avoid discriminating against office holders?

9.22 The effect of the Act is to give a disabled person holding a relevant office or post, or seeking or being considered for appointment, similar rights to those of an employee or job applicant. In the employment context, it is clearly the employer who has the obligations under the Act. However, in relation to office holders, the person on s 4E(1)

s 4F(2)

whom the duties are imposed ('the relevant person') depends on the circumstances. It may be:

- the person with power to make the appointment;
- the person or body with power to recommend or approve the appointment;
- the person with power to determine the terms or working conditions of the appointment (including any benefit or physical feature); or
- the person with power to terminate the appointment.

Discrimination against police officers

- 9.23 Not all police officers are regarded as 'employees' in law. However, the Act says that all police officers count as employees for the purposes of Part 2. The Act also applies to police trainees. As a result, disabled police officers and police trainees (and disabled people applying to join the police service) have the same rights as other employees and job applicants under Part 2 of the Act. These include rights in respect of harassment and victimisation (the latter extending to non-disabled people as well as to those who are disabled).
 - 9.24 These rights which also apply in respect of full and part-time Reserve Officers - are enforceable against the Chief Constable of the Police Service of Northern Ireland. In addition, the Chief Constable is liable under the Act for the discriminatory acts of one police officer (or trainee) committed against another. The Act also applies to other police services, such as the Harbour Police, and the rights are enforceable against the relevant chief officer.

Discrimination against partners in firms

What does the Act say about partners in firms?

- 9.25 The Act says that it is unlawful for a firm, in relation to a position as a partner in the firm, to discriminate against a disabled person:
 - in the arrangements it makes to determine who should be offered that position;
 - in the terms on which it offers him or her that position;
 - by refusing or deliberately omitting to offer him or her that position; or
 - where the disabled person is already a partner in the firm:
 - a. in the way it affords him or her access to any benefits, or by refusing or deliberately omitting to afford him or her access to them; or
 - b. by expelling him or her from the partnership, or subjecting him or her to any other detriment.
- 9.26 The Act also says that it is unlawful for a firm to subject a disabled person who is an existing or prospective partner to harassment, or to victimise any existing or prospective partner, whether or not s/he is disabled.
- 9.27 The effect of the Act is to give a partner or applicant for partnership similar rights against the firm to those of an employee or job applicant against an employer. The

9

s 6A(2)

s 55

s 6C

same applies where people are proposing to form themselves into a partnership and a disabled person is a prospective partner. Limited liability partnerships are also covered. A group of self-employed accountants decide to go into partnership. One of them is disabled because of cancer and discloses this to the other accountants. The others decide to go ahead and form a partnership without the disabled person because they are worried that this person may be absent from work for medical treatment. This is likely to be unlawful. s 6A(3) 9.28 As far as benefits to partners are concerned, the position is the same as that which applies to benefits to employees and office holders (see paragraphs 8.9 to 8.12). s 16B 9.29 The Act gives the Equality Commission power to take action in respect of a discriminatory advertisement for any partnership in a firm (see paragraphs 13.27 to 13.29). This mirrors the Equality Commission's power in respect of advertisements for employment (see paragraphs 7.14). How does the duty to make reasonable adjustments apply in respect of partners in firms? s 6B(1) 9.30 The duty to make reasonable adjustments applies to a firm in the same way as it applies to an employer. It relates to any provision, criterion or practice applied by or on behalf of the firm and to any physical feature of premises occupied by the firm.

9

9.31 Although, in general, the cost of making a reasonable adjustment cannot be passed on to the disabled person concerned, where a firm is required to make adjustments in respect of a disabled partner or prospective partner, the cost of doing so is an expense of the firm. Provided that the disabled person is, or becomes, a partner in the firm, s/he may be required to make a reasonable contribution towards this expense. In assessing the reasonableness of any such contribution, particular regard should be had to the proportion in which the disabled partner is entitled to share in the firm's profits.

A disabled person who uses a wheelchair as a result of a mobility impairment joins a firm of architects as a partner, receiving 20% of the firm's profits. The disabled partner is asked to pay 20% towards the cost of a lift which must be installed in order to enable the disabled partner to work on the premises. This is likely to be reasonable.

Discrimination against barristers

What does the Act say about barristers?

- 9.32 The Act says that it is unlawful for a barrister in relation to any offer of a pupillage to discriminate against a disabled person:
 - in the arrangements which are made to determine to whom it should be offered;
 - in respect of any terms on which it is offered; or
 - by refusing or deliberately omitting to offer it to him or her.

9

s 7A(1)

S	5 7A(2)	9.33	The Act also says that it is unlawful for a barrister, in relation to a disabled pupil, to discriminate against him or her:
			 in respect of any terms applicable to him or her as a pupil;
			 in the opportunities for training, or gaining experience, which are afforded or denied to him or her;
			 in the benefits which are afforded or denied to him or her;
			 by terminating his or her pupillage or by subjecting him or her to any pressure to leave; or
			 by subjecting him or her to any other detriment.
:	s 7A(3) s 55	9.34	In addition, the Act says that it is unlawful for a barrister to subject a disabled person who is a pupil, or who has applied to be a pupil, to harassment, or to victimise such a person, whether s/he is disabled or not. In effect, therefore, barristers and pupils have rights under Part 2 of the Act which are similar to the rights of employees.
	s 7A(4)	9.35	The Act also says that it is unlawful to discriminate against a disabled person in relation to the giving, withholding or acceptance of instructions to a barrister. It would therefore be unlawful for a solicitor to refuse to instruct a barrister merely because the barrister has a disability.
			A solicitor who does not know that a particular barrister is disabled instructs this barrister in a case. The solicitor, on realising that the barrister is disabled, passes the brief to another barrister. This is likely to be unlawful.

9

9.36 The Act gives the Equality Commission power to take action in respect of a discriminatory advertisement for any pupillage (see paragraphs 13.27 to 13.29). This mirrors the Equality Commission's power in respect of advertisements for employment (see paragraphs 7.14).

How does the duty to make reasonable adjustments apply in respect of barristers?

9.37 The duty to make reasonable adjustments applies to barristers in the same way as it applies to an employer. It relates to any provision, criterion or practice applied by or on behalf of a barrister and to any physical feature of premises occupied by or under the control of a barrister which places the disabled person concerned at a substantial disadvantage in comparison with persons who are not disabled. Nevertheless, although it is unlawful to discriminate against a disabled barrister or pupil in relation to the giving, withholding or acceptance of instructions, it should be noted that a solicitor is under no duty to make a reasonable adjustment in relation to a disabled barrister whom s/he instructs.

However, where a disabled pupil is engaged by a barrister who practises from home that may affect the extent of the adjustments which it is reasonable for the barrister to make (see paragraph 5.41).

Discrimination in relation to practical work experience

What does the Act say about practical work experience?

9.38 The Act says that it is unlawful for a 'placement provider' s 14C(1) to discriminate against a disabled person who is seeking or undertaking a work placement:

s 16B

s 7B(1)

- in the arrangements s/he makes for determining who should be offered a work placement;
- in the terms on which s/he affords him or her access to any work placement or any facilities concerned with such placement;

A bank offers three month placements in its accounts department to students on a business studies course. A disabled student is offered a placement on the basis that it lasts for only two weeks as it is believed without asking the student herself - that she will not be able to cope with a three month placement. This is likely to be unlawful.

 by refusing or deliberately omitting to afford the disabled person such access;

A student with ME is refused a work placement at a primary school, as it is believed that the disabled student will not be 'up to the job' because of this disability. This is likely to be unlawful.

by terminating the disabled person's placement;

A person with mental ill health on a four week placement has to take a morning off to attend a hospital appointment related to her disability. The placement is terminated as a result of this. This is likely to be unlawful.

 by subjecting the disabled person to any other detriment during the placement. A design company offers work placements in its creative department. People on placement do a range of things, including research and design. A person with a hearing impairment is given only photocopying to do on his placement because the supervisor does not think that he can do anything else. This would be subjecting the disabled person to a detriment and is likely to be unlawful.

9.39 The Act also states that it is unlawful for a placement provider to subject a disabled person to harassment if s/he is providing a work placement to that person, or if that person has applied to him or her for a work placement, or to victimise such a person, whether s/he is disabled or not.

An employee at a company gives evidence in a tribunal case on behalf of a disabled person who was refused a work placement because of his disability and who has brought a claim of discrimination. The employee is dismissed because of this. This is likely to be unlawful.

- 9.40 For these purposes, a work placement is practical work experience undertaken for a limited period for the purposes of a person's vocational training. A placement provider is any person who provides a work placement to a person whom s/he does not employ.
- 9.41 The above applies only to the extent that discrimination in relation to practical work experience is not the subject of other provisions of the Act, such as those relating to discrimination in the provision of goods, facilities and services. The Equality Commission has issued a

s 14C(2) s 55

s 14C(4)

		separate code of practice giving guidance on the operation of Part 3 of the Act (see Appendix C for details).
		A college arranges work placements in industry for its students. The placement provider in this case is the company offering the work experience and is covered by the provisions of Part 2 of the Act. The college itself is not covered by these provisions.
		A college offers work placements for trainee teachers. In this case the college is a placement provider and is covered by the provisions of Part 2 of the Act.
s 16B	9.42	The Act gives the Equality Commission power to take action in respect of a discriminatory advertisement for any work placement (see paragraphs 13.27 to 13.29). This mirrors the Equality Commission's powers in respect of advertisements for employment (see paragraphs 7.14).
		How does the duty to make reasonable adjustments apply in respect of work placements?
s 14D	9.43	The duty to make reasonable adjustments applies to a placement provider in the same way as it applies to an employer. The duty applies in respect of any provision, criterion or practice applied by or on behalf of the placement provider and to any physical feature of premises occupied by the provider.

A disabled person who has a heart condition obtains a six week placement at a computer company. Such placements are normally offered only on a full-time basis. However, because this would be too tiring for the disabled person, the placement provider allows the disabled person to work mornings only.

A disabled student needs to be accompanied by her support worker whilst on work placement. The placement provider facilitates this by providing an additional work station for the support worker.

While on a work placement, a student with learning disabilities is given personal instruction in health and safety procedures in the workplace, rather than written information.

9.44 The considerations outlined in Chapter 5 apply in determining what adjustments it is reasonable for a placement provider to make. However, the length of the placement will also be a relevant factor. Although many adjustments cost little or nothing to make, it is unlikely to be reasonable for a placement provider to spend significant sums on individually-tailored adjustments in respect of short placements. Nevertheless, some disabled students undertaking work placements may be able to fund adjustments out of their Disabled Students Allowance. Alternatively, some disabled people may already have equipment which they are prepared to use in the workplace. In that case, the placement provider may have to make reasonable adjustments in order to facilitate the use of that equipment (for example, by ensuring that it is transported and stored safely, and is adequately insured whilst in the workplace).

A person with a hearing impairment on a business studies course has a six week placement with a bank. As this person requires an induction loop to participate in meetings, the bank obtains a temporary induction loop as a reasonable adjustment.

A disabled person who uses voice activated software is assisted by the placement provider's IT department to install her own software onto a workplace computer. She is also provided with a headset and microphone to use with it.

A placement provider arranges to transport a disabled person's ergonomic chair to the workplace so that she can use it during a three week work placement.

9.45 An organisation which sends a disabled person on a work placement may also have a duty under the Act to make reasonable adjustments in respect of that person.

Where a course provider supplies a laptop computer for a visually impaired person to complete work, it would also be reasonable for it to supply the computer for that person to use during a related work placement. 9.46 It would be reasonable to expect the sending organisation and the placement provider to co-operate to ensure that appropriate adjustments are identified and made. It is good practice for a placement provider to ask a disabled person about reasonable adjustments before the placement begins, and to allow him or her to visit the workplace in advance to see how his or her requirements can be addressed. Once a particular adjustment has been identified, it would be reasonable for the sending organisation and the placement provider to discuss its implementation in the light of their respective obligations under the Act.

Discrimination in providing occupational pensions and group insurance services

Introduction

10.1 Chapter 8 explains how the Act protects disabled people from discrimination in employment. In particular, paragraph 8.9 makes the point that employers must not discriminate in the way that they make benefits available to disabled employees where those benefits are available to other employees. Some of the most significant benefits which employers commonly make available to their employees relate to occupational pensions and group insurance schemes. This chapter explains what the Act says about discrimination in providing these particular types of benefit. This includes discrimination by the employer and by others concerned with providing the benefit in question.

Occupational pension schemes

What does the Act say about employers?

10.2 As far as employers are concerned, Part 2 of the Act does not distinguish between discrimination in providing pensions and discrimination in relation to other benefits. This is subject to one qualification (explained at paragraphs 10.13 and 10.14) concerning the remedies available for discrimination under Part 2.

s 4(2) s 4A(1)

- 10.3 Thus, an employer must not discriminate against a disabled person in the opportunities it affords him or her for receiving pension benefits, or by refusing him or her, or deliberately not affording him or her, any such opportunity. The duty to make reasonable adjustments applies to the manner in which employers make pensions available to a disabled employee.
- 10.4 To the extent that an employer has control over matters relating to pension benefits afforded to employees, the usual principles about discrimination will apply under Part 2. As explained in Chapter 4, less favourable treatment of a disabled person in relation to such benefits may amount to direct discrimination or to disability-related discrimination. Discrimination may also occur because of a failure to comply with a duty to make a reasonable adjustment to any provision, criterion or practice in relation to pension benefits in respect of a disabled person.

A person with MS completes the first six months of her employment. After this period the employer usually writes to employees inviting them to join the pension scheme. The employer is worried, however, that the disabled employee may draw early on the pension scheme and so does not invite her to join. This is likely to be unlawful.

10.5 In relation to a well established pension scheme, the application of provisions, criteria and practices relating to pension benefits is likely to be controlled by the trustees or managers of the pension scheme. However, when new schemes are set up, employers are likely to have substantial input in relation to such matters (for example, setting scheme rules), and in doing so they must not discriminate against disabled people.

What does the Act say about pension scheme trustees and managers?

- 10.6 Part 2 of the Act contains provisions which relate specifically to discrimination against disabled people by the trustees or managers of occupational pension schemes. The main effect of these provisions is to make it unlawful for pension scheme trustees or managers to contravene a 'non-discrimination rule' which is deemed to be included in every occupational pension scheme.
- 10.7 This non-discrimination rule effectively makes the trustees and managers of the scheme subject to the general provisions of Part 2 by doing two things:
 - First, the non-discrimination rule prohibits pension scheme trustees or managers from discriminating against a disabled person who is a member or prospective member of the scheme in carrying out any of their functions in relation to the scheme. This includes functions relating to the admission and treatment of members of the scheme.
 - Second, the rule prohibits the trustees or managers from subjecting such a person to harassment in relation to the scheme.

The Act sets out the circumstances in which a person counts as a prospective member of a pension scheme.

A disabled person applies to join an occupational pension scheme. The trustees say that he can join but, as he has diabetes, he will not be able to take early retirement on health grounds. This decision is reached with no consideration of the individual circumstances and is likely to be unlawful. 10

s 4G(3)

s 4G(1)

s 4K(2)

```
s 4G(2) & (5)
```

10.8 The Act says that the other provisions of a pension scheme have effect subject to the non-discrimination rule explained above. This means that where there is a conflict between the non-discrimination rule and a rule of the pension scheme, the non-discrimination rule prevails. Thus, pension scheme trustees and managers do not have to act in accordance with a scheme rule which would produce a discriminatory result. The Act also ensures that pension scheme trustees and managers have power to make alterations to the scheme in order to make the scheme conform with the non-discrimination rule.

The rules of an employer's final salary scheme provide that the maximum pension receivable by a member is equivalent to two thirds of salary in the last year of work. An employee becomes disabled and as a result has to reduce her working hours for the remainder of her working life, which will amount to two years. She has worked for twenty years full time prior to this. The scheme's rules put the disabled person at a substantial disadvantage because, regardless of her previous twenty years' service, her pension will only be calculated on her part-time salary as a result of her disability. The trustees decide in her case to average out her salary over a period of years prior to her retirement date, which will enable her full-time earnings to be taken into account. This is likely to be a reasonable adjustment to make.

s 4H 10.9 The duty to make reasonable adjustments applies to pension scheme trustees and managers in the same way as it applies to an employer. The duty is owed to disabled people who are members or prospective members of the scheme, and relates to any provision, criterion or practice (including a scheme rule) applied by

or on behalf of the trustees or managers and any physical feature of premises which they occupy. The Act refers to making alterations to the rules of the pension scheme as an example of a reasonable adjustment.

10.10 The statutory non-discrimination rule does not apply in relation to rights accrued and benefits payable in respect of periods of service prior to 1 October 2004. However, it does apply to communications about such accrued rights or benefits with members or prospective members of the scheme. So far as communications generally are concerned, both the non-discrimination rule and the duty to make reasonable adjustments apply in relation to disabled people who are entitled to, and currently receiving, dependants' or survivors' benefits under a pension scheme as they do in relation to disabled pensioner members of the scheme. The same applies in respect of pension credit members of the scheme. Communications include the provision of information and the operation of a dispute resolution procedure, and reasonable adjustments could involve providing information in accessible formats (such as large print, Braille, audio-tape or disc), or providing a sign language interpreter for a meeting.

A blind person whose partner dies and who is in receipt of a survivor's pension asks for and receives information about the pension scheme on tape. The information relates to the period before 1 October 2004 but nevertheless this is likely to be a reasonable adjustment to make.

s 4G(4) & s 4K

How can disabled people complain about discrimination in relation to occupational pension schemes?

s 17A(1) 10.11 The Act says that if a disabled person believes that s/he has been unlawfully discriminated against in relation to an occupational pension scheme, s/he may make an application to an industrial tribunal. This is the case whether the allegation of discrimination is made against the employer or against the trustees or managers of the pension scheme. Chapter 13 gives more information about making a claim under Part 2 of the Act. However, the Act contains additional provisions which apply specifically to complaints relating to occupational pension schemes. These provisions are explained in the following paragraphs.

s 4l 10.12 When a disabled member or prospective member of an occupational pension scheme complains to an industrial tribunal that s/he has been treated unlawfully under Part 2 by the trustees or managers of the scheme, the employer is treated as a party to the proceedings. The employer is therefore entitled to appear at the tribunal hearing and to make representations to the tribunal. This is because the employer may be required (as a matter of obligation or practice) to fund any award made against the trustees or managers of the scheme.

s 4J(1)
 10.13 In addition, special rules may apply in respect of the remedies available when a complaint brought by a disabled person against the trustees or managers of an occupational pension scheme (or a complaint against the employer itself) is successful. Provided that the complainant is not a pensioner member of the scheme, these rules will apply to such a complaint if it relates to:

- the terms on which people become members of an occupational pension scheme; or
- the terms on which members of the scheme are treated.

183

- 10.14 In such circumstances, the Act modifies the usual rules governing the remedies which may be granted by an industrial tribunal when it upholds a complaint (see paragraphs 13.14 to 13.16). In particular:
 - the tribunal may make a declaration that the complainant has a right to be admitted to the scheme or to membership of the scheme without discrimination in respect of a specified period; but
 - the only compensation which the tribunal may award is compensation for injury to feelings and compensation of the kind described in paragraph 13.16.
- 10.15 A complaint about discrimination by pension scheme trustees or managers may also be made through the pensions dispute resolution mechanism which every occupational pension scheme is required to have for resolving disputes between individual pension scheme members and the trustees or managers. Information about the scheme should give details about this. The Pensions Ombudsman also has power to investigate complaints in certain circumstances. Details about the Pensions Ombudsman, and about the Pensions Advisory Service (which can provide an advice and conciliation service for members of the public who have problems with their occupational pensions), are given in Appendix C.

Group insurance services

What does the Act say about employers?

10.16 As far as employers are concerned, Part 2 of the Act does not distinguish between discrimination in providing benefits relating to insurance and discrimination in relation to other benefits. Thus, an employer must not s 4(2) s 4A(1) discriminate against a disabled person in the opportunities it affords him or her for receiving insurance-related benefits, or by refusing him or her, or deliberately not affording him or her, any such opportunity. The duty to make reasonable adjustments applies to the manner in which employers make insurance services available to a disabled employee. However, it should be noted that the employer's role is often limited to explaining the availability of group insurance services and to proposing employees to the insurer for cover under a group policy.

What does the Act say about group insurers?

s 18 10.17 The Act states that, where an insurer agrees with an employer to provide insurance services to its employees, or to give its employees an opportunity to receive such services, it is unlawful for the insurer to discriminate against a disabled person:

- who is an employee of that employer and who, by virtue of his or her employment, would ordinarily be eligible to receive such services; or
- who has applied for, or is contemplating applying for, such employment.

s 18(3) 10.18 The Act also states that the insurance services to which these principles apply are those for the provision of benefits in respect of:

- termination of service;
- retirement, old age or death; or
- accident, injury, sickness or invalidity.

When does a group insurer discriminate against a disabled person?

- 10.19 Although discrimination by **employers** in providing insurance-related benefits is defined in the usual way for the purposes of Part 2 of the Act (as explained in Chapter 4), the circumstances in which **insurers** will be in breach of Part 2 are defined differently. For the purposes of Part 2, an insurer is only taken to discriminate against a disabled person if it acts in a way which would amount to discrimination under Part 3 of the Act (which deals with discrimination by the providers of goods, facilities or services). For this purpose, the insurer's actions have to be assessed as if the insurer were providing the service in question to members of the public, and as if the disabled person were receiving the service (or trying to secure its provision) as a member of the public.
- 10.20 Part 3 makes it unlawful to discriminate against a disabled person for a disability-related reason in respect of the provision of goods, facilities or services to members of the public unless the conduct can be justified: for example, by reference to actuarial evidence. However, within the confines of the Code it is not possible to explain the operation of Part 3 in detail. The Equality Commission has issued a separate code of practice giving guidance in this regard (see Appendix C for details).

What does the Act say about bringing claims for discrimination in relation to group insurance services?

10.21 The Act says that if a disabled person believes that s/he has been unlawfully discriminated against in relation to group insurance services, s/he may make an application

s 18(2)

to an industrial tribunal. This is the case whether the allegation of discrimination is made against the employer or against the group insurer. Chapter 13 gives more information about making a claim under Part 2 of the Act.

11 Discrimination in providing employment services

Introduction

- 11.1 Although the operation of Part 3 of the Act is mentioned in Chapter 10 (in the context of group insurance services), the explanation of the law in other chapters of the Code is confined to the Act's provisions on employment and occupation - which are in Part 2. However, as the purpose of the Code is to give practical guidance on how to prevent discrimination against disabled people in employment or when seeking employment, it is also necessary to look at what the Act says about the provision of employment services. Such services are covered by Part 3, which relates to discrimination in the provision of goods, facilities or services.
- 11.2 It is not the purpose of this chapter to give a detailed explanation of how the provisions of Part 3 work, but simply to summarise their application to employment services. For a full explanation of the operation of Part 3, reference should be made to the separate Code of Practice: Rights of Access - Goods, Facilities, Services and Premises issued by the Equality Commission in this regard (see Appendix C for details).

What are employment services?

s 21A(1) 11.3 For the purposes of Part 3, 'employment services' are:

- vocational guidance or training services; or
- services designed to assist people to find or keep jobs, or to establish themselves in an occupation in a self-employed capacity.
- 11.4 Employment services therefore include services provided by an employment agency or employment business. This includes employment services provided by JobCentres and Jobs and Benefits Offices and other schemes that assist people to find work. Many people who receive employment services (or who seek the provision of such services) are engaged in, or seeking, contract work. This is particularly true of people who look for work by using the services of employment agencies. The Act gives rights to disabled contract workers not only in relation to the provision of employment services, but also in relation to the contract work itself. What the Act says about contract workers is explained in paragraphs 9.3 to 9.13.

What is unlawful under the Act?

Discrimination in providing employment services

- 11.5 Where a person or body is concerned with the provision of employment services to the public, or a section of the public, the Act says that it is unlawful for it to discriminate against a disabled person:
 - by refusing to provide (or deliberately not providing) to the disabled person any such service which it provides (or is prepared to provide) to members of the public;

11

s 19 applied by s 21A(4) An employment agency refuses to allow a disabled person with a mobility impairment to register with the agency as it says that it does not have any posts which would be 'suitable' and on the ground floor. This is likely to be unlawful.

 in the standard of service which it provides to the disabled person or the manner in which it provides the service;

An employment agency which has a person with a hearing impairment on its register, does not contact him with any work, despite the fact that there is plenty of suitable work for him to do. This is likely to be unlawful.

A woman who has cerebral palsy visits a careers guidance service. The adviser is dismissive of various professions that the woman expresses an interest in, saying that they would be too difficult for her to get into because of her disability. This is likely to be unlawful.

• in the terms on which it provides the service to the disabled person.

A disabled person with a speech impairment requests job advice from his local careers guidance service; he is asked to return next week when there is more time available to meet his needs, although other people, who do not have speech impairments, are being seen there and then. This is likely to be unlawful.

- 11.6 It is irrelevant whether or not a charge is made for the provision of the service. More detailed guidance on the kinds of treatment which are unlawful in relation to the provision of services is set out in Chapter 3 of the Code of Practice: Rights of Access Goods, Facilities, Services and Premises.
- 11.7 It is also unlawful for a provider of employment services to discriminate against a disabled person:
 - in failing to comply with a duty to make reasonable adjustments under Part 3 of the Act to a practice, policy or procedure in circumstances in which the effect of that failure is to place the disabled person at a substantial disadvantage in comparison with people who are not disabled in relation to the provision of the service;
 - in failing to comply with a duty to make reasonable adjustments in relation to a physical feature in circumstances in which the effect of that failure is to make it impossible or unreasonably difficult for the disabled person to make use of any such service; or
 - in failing to comply with a duty under Part 3 of the Act to provide an auxiliary aid or service in the same circumstances.

The duties to make reasonable adjustments under Part 3 of the Act are explained in paragraphs 11.15 to 11.19.

Harassment

s 21A(2) 11.8 It is unlawful for a provider of employment services to subject a disabled person to harassment if that person is someone to whom such services are being provided, or who has requested such services from the provider.

A disabled man who has autism visits a careers guidance service. The adviser makes offensive comments about the man's communication skills. This is likely to amount to harassment whether or not the adviser knows that the man has autism.

Victimisation

11.9 It is also unlawful for a provider of employment services to victimise a person (whether or not s/he is disabled).

Other unlawful acts

11.10 The provisions which apply under Part 2 of the Act in respect of relationships which have come to an end apply equally in respect of employment services, as do the provisions in respect of discriminatory advertisements. In addition, the sanctions which apply under Part 2 in respect of unlawful instructions or pressure to discriminate apply in respect of similar conduct relating to employment services.

What amounts to discrimination?

- 11.11 The Act says that, in the circumstances described in paragraphs 11.5 and 11.7, discrimination can occur in different ways.
- 11.12 One way in which discrimination occurs is when treatment of a disabled person by a provider of employment services:

s 20 applied by s 21A(5)

s 55

ss 16A-16C

- is for a reason related to his or her disability;
- the treatment is less favourable than the way in which the service provider treats (or would treat) others to whom that reason does not (or would not) apply; and
- the service provider cannot show that the treatment is justified.
- 11.13 Discrimination also occurs when a provider of such services fails to comply with a duty to make reasonable adjustments to a practice, policy or procedure.
- 11.14 In addition, discrimination occurs when a provider of employment services:
 - fails to comply with a duty to make reasonable adjustments in relation to a physical feature or a duty to provide an auxiliary aid or service; and
 - cannot show that the failure is justified.

What is the duty to make reasonable adjustments?

- **s 21** 11.15 The duty to make reasonable adjustments under Part 3 of the Act requires service providers to take positive steps to make their services accessible to disabled people. What is required is set out in Part 3 of the Act and comprises a series of duties falling into three main areas:
 - changing practices, policies and procedures (such as a policy of not admitting assistance dogs);

- providing auxiliary aids and services (such as information in accessible formats);
- overcoming a physical feature (such as stairs) by:

a. removing the feature; or

- b. altering it; or
- c. avoiding it; or
- d. providing services by alternative methods.
- 11.16 Where employment services are offered to the public, the provider of those services may have to make a reasonable adjustment if:
 - it has a practice, policy or procedure (including a provision or criterion) which places disabled people at a substantial disadvantage in comparison with people who are not disabled in relation to the provision of the service; or
 - a physical feature makes it impossible or unreasonably difficult for disabled people to make use of such a service; or
 - an auxiliary aid or service (such as information on audio tape or a sign language interpreter) would enable (or make it easier for) disabled people to make use of any such service.
- 11.17 To comply with the duty in these circumstances, the steps which the service provider has to take are those which it is reasonable, in all the circumstances of the case, for it to take in order to:

s 21 applied by s 21A(6)

 change the practice, policy or procedure so that it no longer places disabled people at a substantial disadvantage;

A vocational guidance organisation has a policy of not allowing dogs in its building. It would be reasonable to waive this policy so that disabled people with assistance dogs can enter the building.

• overcome the physical feature;

A building from which an employment agency operates has no colour contrast on the entrance steps, making it unreasonably difficult for blind and partially sighted people to use them. It is likely to be reasonable to apply colour contrast to the steps.

• provide the auxiliary aid or service.

A woman who has dyslexia finds it difficult to fill in an employment agency's registration form. An employee of the agency helps her to fill it in.



11.18 In contrast to the position under Part 2 of the Act, a service provider's duty to make reasonable adjustments under Part 3 is a duty owed to disabled people at large. It is not simply a duty that is weighed up in relation to each individual disabled person who wants to access a service provider's services. Providers of employment services should therefore consider the need for reasonable adjustments in advance of being

approached by a disabled person. Carrying out an access audit of their premises and services (including websites) is likely to assist service providers in this regard. More information about website accessibility is given in Appendix C.

A recruitment agency advertises job vacancies on its website. The agency has its website checked for accessibility, and makes changes to enable disabled people using a variety of access software to use it.

11.19 More detailed guidance on the application of the duty to service providers is set out in Chapters 4 and 5 of the Part 3 Code. Chapter 6 of that Code gives guidance on how leases, the building regulations and other statutory requirements affect a service provider's duty to make reasonable adjustments to premises. The issues are similar to (but not exactly the same as) those explained in the following chapter of this Code.

Can acts which are potentially discriminatory ever be justified?

11.20 Most conduct which is potentially unlawful under the Act is incapable of being justified. Treatment of a disabled person by a provider of employment services can never be justified if it amounts to direct discrimination (see paragraph 4.5), and the same applies to a failure to comply with a duty to make reasonable adjustments to a practice, policy or procedure. However, as noted at paragraphs 11.12 and 11.14, a provider of such s 20(3A) applied by s 21A(5) services may in limited circumstances be permitted to justify certain other failures to make a reasonable adjustment as well as treatment which would otherwise amount to disability-related discrimination (see paragraph 11.21).

s 20 applied by s 21A(5)

- 11.21 Where a claim of discrimination against a provider of employment services is based on a failure to comply with a duty to make reasonable adjustments in relation to a physical feature or a duty to provide an auxiliary aid or service, or on less favourable treatment which does not amount to direct discrimination, that failure or treatment (as the case may be) may be justified if:
 - the service provider holds the opinion that one or more of the conditions listed in section 20 of the Act are satisfied; and
 - it is reasonable in all the circumstances of the case for it to hold that opinion.
- 11.22 The conditions listed in section 20 of the Act relate to:
 - health or safety;
 - the disabled person being incapable of entering into a contract;
 - the service provider being otherwise unable to provide the service to the public;
 - enabling the service provider to provide the service to the disabled person or other members of the public; and
 - the greater cost of providing a tailor-made service.

These conditions are explained in more detail in Chapter 7 of the Code of Practice: Rights of Access - Goods, Facilities and Services and Premises.

Bringing claims for discrimination relating to employment services

11.23 The Act says that if a disabled person believes that s/he has been unlawfully discriminated against in respect of employment services, s/he may make an application to an industrial tribunal. Discrimination includes victimisation, and an industrial tribunal also deals with claims of harassment in relation to employment services. This is an exception to the usual practice for claims brought under Part 3, which are otherwise brought to the county court. Claims of discrimination in respect of employment services are therefore heard in the same venue as other employment-related discrimination claims.

s 25(8)

Making reasonable adjustments to premises - legal considerations

Introduction

- 12.1 In Chapter 5 it was explained that one of the situations in which there is a duty to make reasonable adjustments arises where a physical feature of premises occupied by an employer places a disabled person at a substantial disadvantage compared with people who are not disabled. In such circumstances the employer must consider whether any reasonable steps can be taken to overcome that disadvantage. Making adjustments to premises may be a reasonable step for an employer to have to take. This applies equally to people or bodies other than employers who have duties under Part 2 of the Act. This chapter addresses the issues of how leases, Building Regulations and other statutory requirements affect the duty to make reasonable adjustments to premises.
- 12.2 The issues dealt with in this chapter largely concern the need to obtain consent to the making of reasonable adjustments where an employer occupies premises under a lease or other binding obligation. However, employers should remember that even where consent is not given for altering a physical feature, they still have a duty to make reasonable adjustments, which will involve them considering taking other steps to overcome the disadvantage which the feature causes in respect of the disabled person.

What about the need to obtain statutory consent for some building changes?

- **s 59** 12.3 An employer might have to obtain statutory consent before making adjustments involving changes to premises. Such consents include planning permission, building regulations approval, listed building consent, scheduled monument consent and fire regulations approval. The Act does not override the need to obtain such consents.
 - 12.4 Employers should plan for and anticipate the need to obtain consent to make a particular adjustment. It might take time to obtain such consent, but it could be reasonable to make an interim or other adjustment - one that does not require consent - in the meantime.

An employee who uses a wheelchair requires a ramp in order to access the building. The employer provides a temporary ramp, pending the outcome of an application for consent to install a permanent ramp. This is likely to be a reasonable step to take.

12.5 Where consent has been refused, there is likely to be a means of appeal. Whether or not the employer's duty to take such steps as it is reasonable to take includes pursuing an appeal will depend on the circumstances of the case.

Building Regulations and building design

- 12.6 In general, the design and construction of a new building, and the extension and/or alteration of existing buildings, must comply with Building Regulations (currently the Building Regulations (Northern Ireland) 2000 ('the Building Regulations')). Part R of these Regulations ensures that reasonable provision is made for people to gain access to and use buildings. The Building Regulations require a reasonable standard of accessibility to be provided but are not intended to address the specific requirements of individual employees.
- 12.7 Technical Booklet R sets out the methods and standards by which Part R of the Building Regulations are deemed to be satisfied. The scope of the Technical Booklet R however does not extend to certain access features (for example, the density of doors, door handles, lighting, colour contrasting, etc.)
- 12.8 It should be noted that under the Disability Discrimination (Providers of Services) (Adjustment of Premises) Regulations (Northern Ireland) 2003, there is a partial exemption from the duty to remove or alter an existing physical feature i.e. where the physical feature satisfies "the relevant design standard" which is where it (the physical feature) accords with the relevant provisions of Technical Booklet R: 1994 or Technical Booklet R: 2000. This exemption however, only applies to service providers under Part 3 of the Act. It does not apply to employers under Part 2 of the Act.
- 12.9 The fact that the design and construction of a building (or a physical feature of a building) which an employer occupies meets the requirements of the Building

S.R.2000 No.389

S.R.2003 No.109

Technical Booklet R

Regulations does not diminish the employer's duty to make reasonable adjustments in respect of the building's physical features. In the context of determining what is reasonable for an employer to have to do, it is unlikely to be reasonable for an employer to have to make an adjustment to a physical feature of a building which it occupies if that feature accords with the relevant provisions of Technical Booklet R.

- 12.10 It is good practice for employers to carry out an assessment of the access requirements of each disabled employee, and to consider what alterations can be made to the features of its buildings in order to meet those requirements. It is also good practice to anticipate the requirements of disabled people when planning building or refurbishment works.
- 12.11 When assessing the access requirements of disabled people, it is also likely to be helpful for employers to refer to British Standard 8300:2001, 'Design of buildings and their approaches to meet the needs of disabled people Code of Practice'. Indeed, in the context of determining what is reasonable for an employer to have to do, it is unlikely to be reasonable for an employer to have to a physical feature of a building which it occupies if the design and construction of the physical features of the building is in accordance with BS8300. Further information about BS8300 can be found in Appendix C.
- 12.12 Any work carried out which may be a "structural alteration" will also have to meet the requirements of the Building Regulations (see paragraph 12.6).

12.13 Financial assistance may be available from Access to Work to help meet the cost of making reasonable adjustments to the physical features of a building which an employer occupies (see paragraph 8.19).

What if a binding obligation other than a lease prevents a building being altered?

12.14 The employer may be bound by the terms of an agreement or other legally binding obligation (for example, a mortgage, charge or restrictive covenant) under which it cannot alter the premises without someone else's consent. In these circumstances, the Act provides that it is always reasonable for the employer to have to request that consent, but that it is never reasonable for the employer to have to make an alteration before having obtained that consent.

A retailer builds his shop with the assistance of a bank loan. The loan is secured by way of a charge on the shop under which the bank's consent is required for any changes. It is reasonable for the retailer to seek the bank's consent for changes (such as the installation of a lift) but it is not reasonable for the retailer to have to make any alterations if the bank does not give its consent.

What happens if a lease says that certain changes to premises cannot be made?

12.15 Special provisions apply where an employer occupies premises under a lease, the terms of which prevent it from making an alteration to the premises. In such circumstances, if the alteration is one which the s 18B(3)

12

s 18A(2)

employer proposes to make in order to comply with a duty to make reasonable adjustments, the Act overrides the terms of the lease so as to entitle the employer to make the alteration with the consent of its landlord ('the lessor'). In such a case the employer must first write to the lessor asking for consent to make the alteration. The lessor cannot unreasonably withhold consent but may attach reasonable conditions to the consent.

Sch 4, Part 1 para 1 12.16 If the employer fails to make a written application to the lessor for consent to the alteration, the employer will not be able to rely upon the fact that the lease has a term preventing it from making alterations to the premises to defend its failure to make an alteration. In these circumstances, anything in the lease which prevents that alteration being made must be ignored in deciding whether it was reasonable for the employer to have made the alteration.

An employer occupies premises under a lease, a term of which says that the employer cannot make alterations to a staircase. When deciding whether or not it was reasonable for the employer to make an alteration to the staircase to overcome a disadvantage experienced by a disabled employee, a tribunal will ignore the term of the lease unless the employer has written to the lessor to ask for permission to make the alteration.

What happens if the lessor has a 'superior lessor'?

12.17 The employer's lessor may itself hold a lease the terms of which prevent it from consenting to the alteration without the consent of its landlord ('the superior lessor'). In such circumstances the effect of the superior lease is modified so as to require the lessee of that lease to apply in writing to its lessor (the 'superior lessor' in this context) if it wishes to consent to the alteration. As with the employer's lessor, the superior lessor must not withhold such consent unreasonably but may attach reasonable conditions to the consent.

A bank occupies its premises under a lease, the terms of which prevent it from making alterations without the consent of the landlord. The landlord holds the premises under a lease which has a similar term. The landlord receives an application from the bank for consent to alter the premises. The landlord is entitled to consent to the application if it receives the consent of its landlord. The bank's landlord writes to the superior lessor asking for this consent. The superior lessor cannot unreasonably refuse to give consent but may consent subject to reasonable conditions.

12.18 Where a superior lessor receives an application from its lessee, the provisions described in paragraphs 12.19 to 12.32 apply as if its lessee were the employer.

How do arrangements for gaining consent work?

12.19 Regulations made under the Act govern the procedure for obtaining consent. These Regulations (the Disability Discrimination (Employment Field) (Leasehold Premises) (Northern Ireland) Regulations 2004) are referred to in this Chapter as the 'Leasehold Premises Regulations'. S.R.2004 No.374

- **reg 4** 12.20 The Leasehold Premises Regulations state that, once the application has been made, the lessor has 21 days, beginning with the day on which it receives the application, to reply in writing to the employer (or the person who made the application on its behalf). If it fails to do so it is taken to have unreasonably withheld its consent to the alteration. However, where it is reasonable to do so, the lessor is permitted to take more than 21 days to reply to the request.
 - 12.21 If the lessor replies consenting to the application subject to obtaining the consent of another person (required under a superior lease or because of a binding obligation), but fails to seek the consent of the other person within 21 days of receiving the application (or such longer period as may be reasonable), it will also be taken to have withheld its consent.
 - 12.22 The Leasehold Premises Regulations provide that a lessor will be treated as not having sought the consent of another person unless the lessor has applied in writing to the other person indicating that the occupier has asked for consent for an alteration in order to comply with a duty to make reasonable adjustments, and that the lessor has given its consent conditionally upon obtaining the other person's consent.
 - 12.23 If the lessor replies refusing consent to the alteration, it is recommended that the employer inform the disabled person of this.

When is it unreasonable for a lessor to withhold consent?

12.24 Whether withholding consent will be reasonable or not will depend on the specific circumstances. For example, if a particular adjustment is likely to result in a substantial permanent reduction in the value of the lessor's interest in the premises, the lessor is likely to be acting reasonably in withholding consent. The lessor is also likely to be acting reasonably if it withholds consent because an adjustment would cause significant disruption or inconvenience to other tenants (for example, where the premises consist of multiple adjoining units).

A particular adjustment helps make a public building more accessible generally and is therefore likely to benefit the landlord. It is likely to be unreasonable for consent to be withheld in these circumstances.

A particular adjustment is likely to result in a substantial permanent reduction in the value of the landlord's interest in the premises. The landlord is likely to be acting reasonably in withholding consent.

A particular adjustment would cause significant disruption or major inconvenience to other tenants (for example, where the premises consist of multiple adjoining units). The landlord is likely to be acting reasonably in withholding consent.

	12.25	A trivial or arbitrary reason for withholding consent would almost certainly be unreasonable. Many reasonable adjustments to premises will not harm the lessor's interests and so it would generally be unreasonable to withhold consent for them.
reg 5	12.26	The Leasehold Premises Regulations say that, provided the consent has been sought in the way required by the lease, it is unreasonable for a lessor to withhold consent in circumstances where the lease says that consent will be given to alterations of the kind for which consent has been sought.
reg 6	12.27	The Leasehold Premises Regulations provide that withholding consent will be reasonable where:
		 there is a binding obligation requiring the consent of any person to the alteration;
		• the lessor has taken steps to seek consent; and
		 consent has not been given or has been given subject to a condition making it reasonable for the lessor to withhold its consent.
		It will also be reasonable for a lessor to withhold consent where it is bound by an agreement under which it would have to make a payment in order to give the consent, but which prevents it from recovering the cost from the employer.
		What conditions would it be reasonable for a lessor to make when giving consent?
	12.28	The Leasehold Premises Regulations set out some conditions which it is reasonable for a lessor to make. Depending on the circumstances of the case there may

be other conditions which it would also be reasonable for a lessor to require the employer to meet. Where a lessor imposes other conditions, their reasonableness may be challenged in the course of subsequent industrial tribunal proceedings (see paragraph 12.30).

- 12.29 The conditions set out in the Leasehold Premises Regulations as ones which a lessor may reasonably require an employer to meet are that it:
 - obtains any necessary planning permission and other statutory consents;
 - submits plans and specifications for the lessor's approval (provided that such approval will not be unreasonably withheld) and thereafter carries out the work in accordance with them;
 - allows the lessor a reasonable opportunity to inspect the work after it is completed; or
 - reimburses the lessor's reasonable costs incurred in connection with the giving of consent.

In addition, in a case where it would be reasonable for the lessor to withhold consent, the lessor may give such consent subject to a condition that the premises are reinstated to their original condition at the end of the lease.

What happens if the lessor refuses consent or attaches conditions to consent?

12.30 Where a disabled person brings legal proceedings against his employer under Part 2 of the Act - and those proceedings involve a failure to make an alteration to premises - s/he may ask the industrial tribunal hearing the case to bring in the lessor as an

209

Sch 4,

Part 1.

para 2

reg 7

additional party to the proceedings. The employer may also make such a request. The tribunal will grant that request if it is made before the hearing of the case begins. It may refuse the request if it is made after the hearing of the claim begins. The request will not be granted if it is made after the tribunal has determined the claim.

- 12.31 Where the lessor has been made a party to the proceedings, the industrial tribunal may determine whether the lessor has unreasonably refused consent to the alteration or has consented subject to unreasonable conditions. In either case, the tribunal can:
 - make an appropriate declaration;
 - make an order authorising the employer to make a specified alteration;
 - order the lessor to pay compensation to the disabled person.
- 12.32 The tribunal may require the employer to comply with any conditions specified in the order. If the tribunal orders the lessor to pay compensation, it cannot also order the employer to do so.

Comparison with the procedure for obtaining consent under Part 3

12.33 There are similar provisions which govern the procedure by which a service provider may obtain consent to an alteration which it proposes to make in order to comply with a duty to make reasonable adjustments under Part 3 of the Act. These provisions - which are explained in Chapter 6 of the Code of Practice: Rights of Access -

Goods, Facilities, Services and Premises - apply where a provider of employment services needs to obtain consent in order to make a reasonable adjustment (see paragraphs 11.16 and 11.17). However, it should be noted that the procedures for obtaining consent under Parts 2 and 3 of the Act differ in certain ways. In particular:

- The periods within which the lessor must respond to an application for consent are not the same under Part 3 the relevant period is 42 days beginning with the day on which the application is received.
- Under Part 3 the lessor may require plans and specifications to be submitted **before** it decides whether to give consent.
- Under Part 3 it is possible to make a free-standing reference to the court if the lessor has either refused consent or attached conditions to it. Under Part 2, the question of consent to alterations can only be considered by an industrial tribunal in the course of a complaint of discrimination.

Other relevant provisions

13.1 Additional provisions of the Act (and provisions of other legislation) are relevant to understanding the protection from discrimination afforded to disabled people in respect of employment. This chapter describes those provisions, and focuses in particular on the way in which disputes should be resolved.

Resolving disputes

When is it necessary to try to resolve disputes within the workplace?

- 13.2 It is good practice to attempt to resolve disputes within the workplace and without resorting to legal proceedings. In addition, the Employment (Northern Ireland) Order 2003 (Dispute Resolution) Regulations (Northern Ireland) 2004 ('the 2004 Regulations') **require** employers and employees to try to do this in certain circumstances. The requirement may apply where an employer has dismissed an employee, or is contemplating dismissing or taking disciplinary action against him or her or where an employee has a grievance against his or her employer. In broad terms, the statutory procedures - which are set out in the Employment (Northern Ireland) Order 2003 ('the 2003 Order') - require:
 - the grounds for the employer's action, or details of the employee's grievance, to be set out in writing and sent to the other party;
 - a meeting to take place between the employer and employee in order to discuss the matter, and for the employer to inform the employee of its decision afterwards; and

S.R. 2004 No. 521

• an internal appeal against that decision to take place if the employee is not satisfied with it.

Full details on the implications of the dispute resolution legislation for employers and employees are available from the Department for Employment and Learning website: www.delni.gov.uk/resolvingdisputes. You can also get help from the Labour Relations Agency or your local Citizens Advice Bureau.

- 13.3 Employers must make reasonable adjustments in respect of the way in which the statutory procedures are implemented to prevent a disabled person from being placed at a substantial disadvantage in comparison with people who are not disabled. Failure to do so will itself amount to a breach of the Act.
- 13.4 So, for example, it is likely to be a reasonable adjustment for an employer to allow a disabled employee who has communication difficulties some assistance to make a written statement of a grievance s/he wishes to make (such as by providing him or her with assistance via an impartial party). Depending on the circumstances, it may be reasonable to allow a disabled person with learning disabilities to be accompanied to a meeting by a family member or friend, or to send written communications to a blind or visually impaired person in a format which is accessible to him or her.
- 13.5 The effect of what the 2003 Order says about dispute resolution is that, where an employee has a grievance against his or her employer (including an allegation that the employer has breached Part 2 of the Act) subject to the exceptions outlined below, s/he may not be entitled to commence industrial tribunal proceedings without first giving the employer a written statement of the reasons for the grievance and allowing at least 28 days to pass.

13.6 In addition, in any industrial tribunal proceedings where it appears to the tribunal that the statutory procedures apply, any award which the tribunal makes will be either reduced or increased if the procedures were not fully complied with before the proceedings were begun. The award will be reduced if the tribunal concludes that the failure to comply is wholly or mainly attributable to the employee. It will be increased if the tribunal concludes that the employer is wholly or mainly at fault. Save in exceptional circumstances, the amount by which any award will be reduced or increased will be between 10% and 50%.

Will the legislation require parties to attempt to resolve disputes internally in every case?

- 13.7 It is good practice to try to resolve disputes within the workplace wherever possible. There are occasions, however, where internal dispute resolution will not be practical or appropriate. The legislation recognises this and the 2004 Regulations specify exceptional circumstances where parties will not be required to complete the statutory procedures mentioned in paragraph 13.2.
- 13.8 Compliance with these procedures is only **required** in respect of disputes involving employers and employees who work (or have worked) under a contract of service or apprenticeship. People who fall within the wider definition of 'employment' set out at paragraph 3.8 do not need to use the statutory procedures before bringing a claim in the industrial tribunal. Neither do these procedures apply to disputes under the Act involving partners in firms, barristers, or office holders, for example. Equally, they do not apply in respect of disputes between employees and pension scheme trustees or managers, or to disputes involving providers of employment services.

13.9	As indicated above, however, there are also cases where, because of the circumstances surrounding the dispute, it may be inappropriate for the parties to be required to seek a resolution within the workplace. The 2004 Regulations provide that the statutory procedures do not need to be followed if:
	 one of the parties to the dispute has reasonable grounds to believe that compliance with the procedure would result in a significant threat to himself or herself, his or her property, to another person, or to another person's property; or
	 one of the parties has been subjected to harassment and has reasonable grounds to believe that complying with the procedure would result in his or her being subjected to further harassment; or
	 it is not practicable to comply with the procedure within a reasonable period.
13.10	In addition, the 2004 Regulations say that, where an employee's grievance is that disciplinary action taken against himself or herself amounts to discrimination by the employer, the parties are not required to meet to discuss the matter. However, an employee must still send the employer written details of his or her grievance before commencing industrial tribunal proceedings. More specific information about when the statutory procedures apply and when they do not have to be followed is available from the Department for Employment and Learning website: www.delni.gov.uk/resolvingdisputes.

What if a dispute cannot be resolved by using an employer's grievance procedure?

- 13.11 The Act says that a person, who believes that someone has unlawfully discriminated against him or her (which includes victimising him or her or failing to make a reasonable adjustment) or has subjected him or her to harassment, may make an application to an industrial tribunal. Such an application must normally be made within **three months** of the date when the incident complained about occurred. Before making an application to a tribunal, however, it is necessary to ensure that any requirement under the 2003 Order relating to internal dispute resolution procedures has been complied with (see paragraphs 13.2 to 13.10).
- 13.12 In cases where such a procedure applies, the 2004 Regulations say that the usual time limit is generally extended by three months (so that the time limit for making a claim becomes six months in total) in order to allow time for the completion of the appropriate procedure. However, this only applies to those people who are **required** to use such a procedure. It does not apply to people who do not have to do so (see paragraph 13.8).
- 13.13 Before making an application to an industrial tribunal (or within 28 days of lodging it), a disabled person can request information relevant to his or her claim from the person against whom the claim is made. This is known as the 'questionnaire procedure'. There is a standard form of questionnaire and accompanying booklet which explains how the procedure works (see Appendix C for details).

s 17A(1) s 55

Sch 3, para 3

reg 15

s 17A(2)	13.14	 When an application to an industrial tribunal has been made, the Labour Relations Agency (LRA) will try to promote settlement of the dispute without a tribunal hearing. However, if a hearing becomes necessary - and if the application is upheld - the tribunal may: declare the rights of the disabled person (the applicant), and the other person (the respondent) in relation to the application; order the respondent to pay the applicant compensation; and recommend that, within a specified time, the respondent takes reasonable action to prevent or
		reduce the adverse effect in question.
s 17A(4)	13.15	The Act allows compensation for injury to feelings to be awarded whether or not other compensation is awarded.
s 17A(5)	13.16	The Act also says that if a respondent fails, without reasonable justification, to comply with an industrial tribunal's recommendation, the tribunal may:
		• increase the amount of compensation to be paid; or
		 order the respondent to pay compensation if it did not make such an order earlier.
s 4J 3	13.17	It should be noted that in relation to certain claims of discrimination concerning occupational pension schemes, the Act modifies the remedies available (see paragraphs 10.13 and 10.14).
	13.18	Sources of information about how to make an application to an industrial tribunal are listed in Appendix C.
218		

Other provisions

Anti-avoidance provisions

- 13.19 A disabled person cannot usually waive his or her rights (or an employer's duties) under the Act. The Act says that any term in a contract of employment or other agreement is 'void' (i.e. not valid) where:
 - making the contract is unlawful under Part 2 of the Act because of the inclusion of the term;
 - the term is included in furtherance of an act which is itself unlawful under Part 2; or
 - the term provides for the doing of an act which is unlawful under Part 2.
- 13.20 An employer should not include in an agreement any provision intended to avoid obligations under the Act, or to prevent someone from fulfilling obligations. An agreement should not, therefore, be used to try to justify less favourable treatment or deem an adjustment unreasonable. Even parts of agreements which unintentionally have such an effect are unenforceable if they would restrict the working of the Act's provisions on employment and occupation. However, as explained in Chapter 12, special arrangements cover leases and other agreements which might restrict the making of adjustments to premises.

Compromise agreements

13.21 The effect of the Act's provisions is also to make a contract term unenforceable if it would prevent anyone from making an application to an industrial tribunal under Part 2, or would force them to discontinue an application (see paragraph 13.11). There is a limited

Sch 3A, Part 1

Sch 3A, Part 1

		exception to this principle relating to settlement agreements which have either been brokered by the LRA, or which are made in circumstances where the following conditions are satisfied:
		 the disabled person has received advice from a relevant independent adviser about the terms and effects of the agreement, particularly its effect on his or her ability to apply to a tribunal;
		 the adviser has a contract of insurance or an indemnity provided for members of a profession or professional body; and
		 the agreement is in writing, relates to the application, identifies the adviser and says that these conditions are satisfied.
Sch 3A, Part 1		The Act defines the circumstances in which a person is a 'relevant independent adviser' for this purpose.
		Variation of contracts
Sch 3A, para 3	13.22	A disabled person interested in a contract of employment or other agreement which contains a term of the kind mentioned in paragraph 13.19 may apply to a county court for an order removing or modifying that term.
2		Collective agreements and rules of undertakings
Sch 3A, Part 2	13.23	There are also anti-avoidance provisions in the Act relating to the terms of collective agreements, and to rules made by employers in relation to working practices or recruitment. The Act says that any such term or rule is void where:
220		

- making the collective agreement is unlawful under Part 2 of the Act because of the inclusion of the term;
- the term or rule is included in furtherance of an act which is itself unlawful under Part 2; or
- the term or rule provides for the doing of an act which is unlawful under Part 2.
- 13.24 It does not matter whether the collective agreement was entered into, or the rule was made, before or after these provisions became law - the term or rule in question can still be challenged under the Act. In addition, where these provisions apply, certain disabled people may ask an industrial tribunal to make a declaration that a discriminatory term or rule is void if they believe that it may affect them in the future. The Act specifies which disabled people may make such an application.

Provision for certain charities

- 13.25 The Act states that some charities (and governmentfunded supported employment schemes such as Employment Support) are allowed to treat some groups of disabled people more favourably than others. They can do this only if two conditions are met:
 - first, the group being treated more favourably must be connected with the charitable purposes of the charity;
 - second, the more favourable treatment of that group must be in pursuance of those charitable purposes.

In the case of supported employment, those treated more favourably must be disabled people whom the programme aims to help. 13

s18C

The Equality Commission for Northern Ireland

General functions

13.26 The Equality Commission has statutory powers to work towards the elimination of discrimination and to promote the equalisation of opportunities for disabled people. In particular, the Equality Commission:

- keeps the Act under review;
- provides assistance and support to disabled litigants under the Act;
- provides information and advice to anyone with rights or obligations under the Act;
- carries out formal investigations;
- prepares new or revised Codes of Practice; and
- has various powers and duties in relation to the statutory equality duty under Section 75 of the Northern Ireland Act 1998.

Enforcement of certain provisions under Part 2

- 13.27 In addition, the Equality Commission has a direct involvement in the enforcement of the provisions of Part 2 relating to:
 - instructing or pressurising other people to act unlawfully (see paragraph 3.22); and
 - discriminatory advertisements (see paragraphs 7.12 to 7.14).

Northern Ireland Act 1998, section 74.

The Equality (Disability, etc.) (NI) Order 2000

- 13.28 Only the Equality Commission may bring proceedings in respect of these matters. Where it does so, the Equality Commission may seek:
 - a declaration from an industrial tribunal that a • contravention has occurred; and
 - an injunction from a county court restraining further • contraventions.
- 13.29 The Equality Commission may only apply for an injunction or order if it has first obtained a declaration from an industrial tribunal that an unlawful act has occurred, and then only if it appears to the Equality Commission that a further unlawful act is likely to occur unless the person concerned is restrained.

Further information

13.30 The Equality Commission may be contacted at:

The Equality Commission for Northern Ireland Equality House 7-9, Shaftesbury Square Belfast BT2 7DP Telephone: 028 90 500 600

- Textphone: 028 90 500 589
- Fax: 028 90 248 687
- information@equalityni.org E-mail:
- Website: www.equalityni.org

Typetalk

Appendix A: Changes to the Act

The table below summarises the main changes to the Act's provisions on employment and occupation which took effect on 1 October 2004. It does not include all the changes occurring on that date, and is not a full explanation of the law.

	Position before 1 October 2004	
Scope	 The Act covered employers with 15 or more employees. Some occupations (e.g. police & firefighters) were not covered. 	
Types of Discrimination	 Three kinds of discrimination: Less favourable treatment. Failure to make reasonable adjustments. Victimisation. 	
When is Justification relevant?	 Justification was of relevance in cases about: Less favourable treatment. Failure to make reasonable adjustments. 	
Harassment	Covered, but no separate provisions on this.	
Claims	Most claims covered by the Code were brought in the industrial tribunal apart from those involving trustees and managers of occupational pension schemes and claims about employment services.	

Po	sition after 1 October 2004
PO	sition after 1 October 2004
•	All employers are covered by the Act except for the Armed Forces.
•	New occupations such as police and partners in firms are covered.
•	Practical work experience, whether paid or unpaid, is covered.
•	There are new provisions on discriminatory advertisements.
•	Employment services are covered.
Fo	our kinds of discrimination:
•	Direct discrimination.
•	Failure to make reasonable adjustments.
•	'Disability-related discrimination'.
•	Victimisation.
Ju	stification is NOT relevant in cases about:
•	Direct discrimination.
•	Failure to make reasonable adjustments.
Ju	stification is relevant in cases about:
•	Disability-related discrimination.
Ne	ew provisions on harassment.
	l claims covered by this Code are brought in the industrial bunal.

Appendix B: The Meaning of Disability

This appendix is included to aid understanding about who is covered by the Act and should provide sufficient information on the definition of disability to cover the large majority of cases. The definition of disability in the Act is designed to cover only people who would generally be considered to be disabled. A Government publication 'Guidance on matters to be taken into account in determining questions relating to the definition of disability', is also available.

When is a person disabled?

A person has a disability if s/he has a physical or mental impairment, which has a substantial and long-term adverse effect on his or her ability to carry out normal day-to-day activities.

What about people who have recovered from a disability?

People who have had a disability within the definition are protected from discrimination even if they have since recovered.

What does 'impairment' cover?

It covers physical or mental impairments; this includes sensory impairments, such as those affecting sight or hearing.

Are all mental impairments covered?

The term 'mental impairment' is intended to cover a wide range of impairments relating to mental functioning, including what are often known as learning disabilities. The Act says that a mental illness must be a clinically well-recognised illness in order to amount to a mental impairment. A clinically wellrecognised illness is one that is recognised by a respected body of medical opinion.

What is a 'substantial' adverse effect?

A substantial adverse effect is something which is more than a minor or trivial effect. The requirement that an effect must be substantial reflects the general understanding of disability as a limitation going beyond the normal differences in ability which might exist among people.

What is a 'long-term' effect?

A long-term effect of an impairment is one:

- which has lasted at least 12 months; or
- where the total period for which it lasts is likely to be at least 12 months; or
- which is likely to last for the rest of the life of the person affected.

Effects which are not long-term would therefore include loss of mobility due to a broken limb which is likely to heal within 12 months and the effects of temporary infections, from which a person would be likely to recover within 12 months.

What if the effects come and go over a period of time?

If an impairment has had a substantial adverse effect on normal day-to-day activities but that effect ceases, the substantial effect is treated as continuing if it is likely to recur; that is if it is more probable than not that the effect will recur.

What are 'normal day-to-day activities'?

They are activities which are carried out by most people on a fairly regular and frequent basis. The term is not intended to include activities which are normal only for a particular person or group of people, such as playing a musical instrument, or a sport, to a professional standard or performing a skilled or specialised task at work. However, someone who is affected in such a specialised way but is also affected in normal day-today activities would be covered by this part of the definition. The test of whether an impairment affects normal day-to-day activities is whether it affects one of the broad categories of capacity listed in Schedule 1 to the Act. They are:

- mobility;
- manual dexterity;
- physical co-ordination;
- continence;
- ability to lift, carry or otherwise move everyday objects;
- speech, hearing or eyesight;
- memory or ability to concentrate, learn or understand; or
- perception of the risk of physical danger.

What about treatment?

Someone with an impairment may be receiving medical or other treatment which alleviates or removes the effects (though not the impairment). In such cases, the treatment is ignored and the impairment is taken to have the effect it would have had without such treatment. This does not apply if substantial adverse effects are not likely to recur even if the treatment stops (i.e. the impairment has been cured).

Does this include people who wear spectacles?

No. The sole exception to the rule about ignoring the effects of treatment is the wearing of spectacles or contact lenses. In this case, the effect while the person is wearing spectacles or contact lenses should be considered.

Are people who have disfigurements covered?

People with severe disfigurements are covered by the Act. They do not need to demonstrate that the impairment has a substantial adverse effect on their ability to carry out normal day-to-day activities.

What about people who know their condition is going to get worse over time?

Progressive conditions are conditions which are likely to change and develop over time. Examples given in the Act are cancer, multiple sclerosis, muscular dystrophy and HIV infection. Where a person has a progressive condition s/he will be covered by the Act from the moment the condition leads to an impairment which has some effect on ability to carry out normal day-to-day activities, even though not a substantial effect, if that impairment is likely eventually to have a substantial adverse effect on such ability.

What about people who are registered disabled?

Those registered as disabled under the Disabled Persons (Employment) Act (Northern Ireland) 1945 both on 12 January 1995 and 2 December 1996 were treated as being disabled under the Disability Discrimination Act 1995 for three years from the latter date. At all times from 2 December 1996 onwards they are covered by the Act as people who have had a disability. This does not preclude them from being covered as having a current disability any time after the three year period has finished. Whether they are or not will depend on whether they, like anyone else, meet the definition of disability in the Act.

Are people with genetic conditions covered?

If a genetic condition has no effect on ability to carry out normal day-to-day activities, the person is not covered. Diagnosis does not in itself bring someone within the definition. If the condition is progressive, then the rule about progressive conditions applies.

Are any conditions specifically excluded from the coverage of the Act?

Yes. Certain conditions are to be regarded as not amounting to impairments for the purposes of the Act. These are:

- addiction to or dependency on alcohol, nicotine, or any other substance (other than as a result of the substance being medically prescribed);
- seasonal allergic rhinitis (e.g. hayfever), except where it aggravates the effect of another condition;
- tendency to set fires;
- tendency to steal;
- tendency to physical or sexual abuse of other persons;
- exhibitionism;
- voyeurism.

Also, disfigurements which consist of a tattoo (which has not been removed), non-medical body piercing, or something attached through such piercing, are to be treated as not having a substantial adverse effect on the person's ability to carry out normal day-to-day activities.

Appendix C: Further information

Leaflets about the Disability Discrimination Act 1995 ("the Act")

A range of leaflets about the Act is available from the Equality Commission

The Equality Commission for Northern Ireland Equality House 7-9 Shaftesbury Square Belfast BT2 7DP

Telephone: 028 90 500 600

Textphone: 028 90 500 589

Fax: 028 90 248 687

E-mail: information@equalityni.org

Typetalk

On-line information

Information about the Act, including practical guidance about employment policies and procedures is also available on the Equality Commission's website: **www.equalityni.org.**

Codes of Practice

Codes of Practice and accompanying guidance for Part 2 (this Code as well as the Disability Code of Practice for Trade Organisations and Qualifications Bodies), and Part 3 (the Code of Practice: Rights of Access - Goods, Facilities, Services and Premises) are available from the Equality Commission using the contact information given above or through the Equality Commission's website.

Making a complaint

Information about making a complaint of disability discrimination to an industrial tribunal or using the Questionnaire Procedure is available from the Equality Commission or from the Office of the Industrial Tribunals and Fair Employment Tribunal (see below for contact details). A form and accompanying guidance can also be obtained from the Equality Commission, as well as JobCentres and Jobs and Benefits Offices and Citizens Advice Bureaux.

Practical guidance

In addition to the Equality Commission, there is a wide range of practical help and advice to assist employers in the recruitment and employment of disabled people available from the Disablement Advisory Service through JobCentres and Jobs and Benefits Offices. Addresses and telephone numbers of JobCentres and Jobs and Benefits Offices can be found in local telephone directories or through the Department for Employment and Learning website: www.delni.gov.uk.

Access to Work

The Disablement Advisory Service through its Disablement Employment Advisers, based in JobCentres and Jobs and Benefits Offices, provides information and advice to disabled people in work or looking for work. It is also the contact point for people wishing to get help from the Access to Work scheme. For further information on the Access to Work scheme, disabled people should contact their local Disablement Employment Adviser or the Department for Employment and Learning website: www.delni.gov.uk.

Access to Work may be able to offer advice and help on the following:

Adaptations to Premises and Equipment

Modification of an employer's premises or equipment, to enable them to employ or retain a disabled employee. Employers will be expected to contribute if adaptations bring general benefits to the business, firm, other employees or customers.

Communication support at interview

Help with the costs of employing an interpreter or communicator to accompany a hearing impaired person, where there might be communication difficulties at a job interview with an employer.

Miscellaneous assistance

Provision under this element is largely 'one off' items of support that do not fit elsewhere, such as a grant towards the costs of deaf awareness training for close colleagues of a deaf person.

Special aids and equipment

Provision of aids and equipment to a disabled person which a non-disabled person doing the same job would not need. Leasing equipment can also be considered when it offers the most cost-effective solution. Employers will be expected to contribute if aids bring general benefits to the business, firm, other employees or customers.

Support workers

Help with the costs of employing personal support for a person with a disability either at work or getting to and from work. (This includes personal reader support for a person with a visual impairment).

Travel to work

Support when a disabled person incurs extra costs in travelling to and from work because of their disability. Beneficiaries are expected to contribute the usual costs of travelling to work.

Other sources of information

Labour Relations Agency

The Labour Relations Agency (LRA) can help employers and individuals with information on legislation and on industrial relations practices and procedures. The LRA's contact details are:

Head Office:	4-8, Gordon Street Belfast BT1 2LG		
	Telephone:	028 9032 1442	
	Fax:	028 9033 0827	
	Email:	info@lra.org.uk	
Regional Office:	1-3, Guildhall Street Londonderry BT48 6BJ		
	Telephone:	028 7126 9639	
	Fax:	028 7126 7729	
	Email:	info@lra.org.uk	
	Website:	www.lra.org.uk	

Office of the Industrial Tribunals and Fair Employment Tribunal

Longbridge House 20 - 24 Waring Street Belfast BT1 2EB

Tel: 028 9032 7666 Fax: 028 9023 0184

Website: www.industrialfairemploymenttribunalsni.gov.uk

Guidance on building design

Copies of British Standard 8300:2001 "Design of buildings and their approaches to meet the needs of disabled people - Code of Practice" can be obtained from the British Standards Institute:

Telephone: 020 8996 9002 Fax: 020 8996 7001 Website: www.bsi-global.com

Making websites accessible

Disabled people use a wide range of specialist hardware and software to access computers. It is important that websites are designed to be compatible with this. Websites can also have 'access features' built into their design, such as a choice of font sizes or colour schemes.

RNIB's online Web Access Centre can provide more information on designing and evaluating websites.

Health and Safety Executive for Northern Ireland

The Health and Safety Executive for Northern Ireland is the regional authority for health and safety at work in Northern Ireland and, along with the District Councils, is responsible for the regulation of the risks to health and safety arising from work activity in Northern Ireland.

Telephone:08000320121 (Free phone Helpline)Fax:02890235383Textphone:02890546896Email:hseni@detini.gov.ukWebsite:www.hseni.gov.uk

Other sources of information

The Information Commissioner's Office provides information and guidance about the Data Protection Act and the Codes of Practice which relate to it.

Telephone: 01625 545 745 Fax: 01625 524510 Email: mail@ico.gsi.gov.uk Website: www.informationcommissioner.gov.uk

OPAS, The Pensions Advisory Service, is an independent organisation that provides information and guidance on the whole spectrum of pensions, including State, company, personal and stakeholder schemes. It can help any member of the public who has a problem, complaint or dispute with their occupational or private pension provider which they have been unable to resolve with the Scheme Manager.

Telephone: 0845 6012923 Fax: 020 7233 8016 Email: enquiries@opas.org.uk Website: www.opas.org.uk **The Pensions Ombudsman** investigates any complaint alleging injustice as a result of maladministration of a pension scheme.

Telephone:02078349144Fax:02078210065Email:enquiries@pensions-ombudsman.org.ukWebsite:www.pensions-ombudsman.org.uk

Index

access audits 2.7, 11.18 Access to Work scheme 5.21, 5.38, 5.39, 7.22, 8.6, 8.19-8.20, 12.11 and Appendix C advertisements 2.15, 2.47, 9.20, 9.29, 9.36, 11.10, 13.27 and Appendix A legal action in respect of discriminatory advertisements 7.14, 9.42 recruitment 2.48, 7.11-7.15 advising fellow employees of employee's disability 8.21-8.22 aiding an unlawful act 3.24-3.25, 6.1 annual appraisal interviews 2.26, 5.12, 8.6 anti-avoidance provisions 13.19-13.20 collective agreements 13.23-13.24 rules of undertakings 13.23-13.24 variation of contracts 13.22 anti-discriminatory policies and practices, implementation 2.8-2.25 application forms and information 7.16-7.18 aptitude tests 7.25-7.26 Armed Forces 1.2, 3.9, 5.40 and Appendix A assumptions about disabled people, avoiding 2.4, 4.8, 6.7, 6.9 auditing policies and procedures 2.26 avoiding discrimination - action to take 2.1-2.50 barristers 3.11, 9.32-9.37, 13.8 benefits provided by employers 8.9-8.12

BS8300 2.7, 12.9, 12.11 and Appendix C

building design, guidance, see Appendix C Building Regulations 2.7, 12.1, 12.3, 12.6-12.13 Technical Booklet R 12.7-12.9 car parking 5.8, 8.10 careers guidance services 3.13, 11.3-11.23 complaint, making, see Appendix C collective agreements 13.23-13.24 complaints of discrimination dealing with complaints 2.19-2.22 harassment procedure 2.12-2.16 monitoring complaints 2.44 pension complaints 10.11-10.15 compromise agreements 13.21 conditions excluded from Act, 3.9 and Appendix B confidential information 8.21-8.23 constructive dismissal 8.27 consultation with disabled employees 2.36-2.42 contract workers 3.11, 9.3-9.13 duty to make reasonable adjustments 9.7-9.12

customers, disabled 2.7, 2.39

Data Protection Code of Practice (Part 1) on Recruitment and Selection 7.27 and Appendix C direct discrimination 3.17, 4.1-4.23, 4.26, 4.30, 4.31, 4.32, 4.37, 6.1, 6.7, 6.9, 6.10, 6.12, 6.14, 7.8, 7.32, 8.5, 8.8, 8.13, 8.24, 10.4, 11.20 and Appendix A comparators 4.13-4.22 compared to disability-related discrimination 4.29 definition 4.5-4.6 evidence required to prove 4.43 justification 4.23 disability, meaning of, see Appendix B, 3.2, 3.5, 3.6 disability awareness training 2.24 and Appendix C disability in the past 3.5, 3.7, 9.6 disability-related discrimination 3.17, 4.1, 4.2, 4.3, 4.14, 4.19, 4.27-4.32 compared to direct discrimination 4.29 definition 4.27-4.29 evidence required to prove 4.43 disabled customers 2.39 disabled person, definition, see Appendix B, 3.2-3.4 Disabled Students Allowance 9.44 disciplinary process 8.26 disciplinary rules 2.23 disclosure of information about disability 8.23 disfigurement, see Appendix B dismissal, duty to make reasonable adjustments 5.5, 8.24, 8.26 dispute resolution 2.49-2.50, 10.15, 13.2-13.18 pensions 10.15 reasonable adjustments 13.3-13.4 statutory procedures 13.2-13.6 when cannot be resolved 13.11-13.18 diversity policies see anti-discriminatory policies and practices emergency evacuation procedures 2.26 employee becomes disabled or disability worsens, reasonable adjustments 8.16-8.17 employees, discrimination against 8.1-8.32 benefits provided by employers 8.9-8.12 induction, training and development 8.7-8.8 managing disability or ill health 8.15-8.23

promotion and transfer 8.13-8.14 termination of employment 8.24-8.27 terms and conditions of service 8.4-8.6 employees, former 5.6, 8.11, 8.28-8.32 employers

covered by Act 1.2, 3.8-3.10

use of term in Code 1.17

employment, definition 3.8

employment agencies 3.13, 11.4

see also employment services

employment service providers, disputes with 13.8

employment services

discrimination in 11.1-11.23

duty to make reasonable adjustments 11.15-11.19, 11.20

harassment of disabled person 11.8, 11.23

services included 11.3-11.4

victimisation of person 11.9, 11.23

Employment Support Scheme 9.13, 13.25

enforcement of rights under the Act 3.26

equal opportunities policy 2.11-2.14, 2.24

Equality Commission for Northern Ireland 1.4-1.5, 1.20-1.21, 2.11-2.12, 2.16, 2.32, 3.22, 7.14, 9.20, 9.29, 9.36, 9.41, 9.42, 10.20, 13.26,13.27-13.30

contact details 1.21

enforcement of certain provisions of Part 2 13.27-13.30

functions 13.26

equipment owned by disabled employee, use of 5.39, see also examples in 4.22

evidence required to prove discrimination or harassment 4.41-4.45

exit interviews 2.45

expert advice 2.5, 8.16

former employees 5.6, 8.11,8.28-8.32

government departments and agencies 3.12 grievance procedures 2.19 -2.22, 2.49, 2.50, 5.20 harassment procedure 2.12-2.16 monitoring of grievances 2.44 group insurance services 3.12, 5.17, 8.3, 8.9, 10.16-10.21 rights of disabled people to complain about discrimination to industrial tribunal 10.21 services covered 10.18 guaranteed interview scheme 7.20-7.21

harassment 3.21, 3.25, 4.4, 4.38-4.39, 6.1, 8.28, 9.4, 9.18, 9.23, 9.26, 9.34, 9.39, 10.7, 11.8, 11.23, 13.9 and Appendix A, see also complaints of discrimination

definition 4.38 dispute resolution 13.9, 13.11 evidence required to prove 4.41-4.45 examples 4.39 policy and procedure 2.12-2.14 health and safety 5.26, 6.7-6.13 and Appendix C risk assessment 6.8-6.13

image of organisation, promoting positive 2.48 induction procedures 8.7

industrial tribunals 3.26, 4.41-4.42, 7.15, 13.8, 13.10, 13.11-13.18, 13.21, 13.24, 13.28, 13.29, Appendix A and Appendix C awards 13.6, 13.14-13.17 claims for unlawful discrimination, employment services 11.23 compensation 13.14-13.17 group insurance services 10.21

lessor party to proceedings 12.28-12.30 occupational pension schemes 10.11-10.15 procedure 13.11-13.18 questionnaire procedure 13.13 time limit 13.11-13.12 use of Labour Relations Agency 13.14 Information Commissioner's Office, contacting, see Appendix C instructions and pressure to discriminate 3.22, 6.1, 11.10 insurance services see group insurance services insurers 3.12, 10.16-10.21 interview arrangements, see recruitment - selection, assessment and interview arrangements job advertisements 2.15, 7.11-7.15 job application forms and information 7.16-7.18 job description/specification 7.7-7.10 justification for discrimination 4.3, 4.31-4.32, 6.1-6.16, 7.8, 7.10, 7.31, 13.16 and Appendix A conduct which cannot be justified 6.1 health and safety concerns 6.7-6.13 medical information 6.14-6.16 when permitted 6.2-6.6 Labour Relations Agency 13.14, 13.21 contact details, see Appendix C landlords of premises 3.12 Leasehold Premises Regulations 12.19-12.29 leasehold property see reasonable adjustments to premises, lease prevents building being altered without consent leave policy 2.25

limited liability partnerships 9.27

managing disability or ill health 8.15-8.23
retention of disabled employees 8.15-8.18
medical examinations, job applicants 7.31
medical information, consideration of 6.14-6.16
medical questionnaires 7.27-7.28
Ministers of the Crown 3.12
monitoring disabled employees and applicants 2.27-2.45, 7.27, 7.30
exit interviews 2.45

grievances 2.44 quantitative 2.31-2.35 qualitative 2.36-2.42

nature of job changes, reasonable adjustments 8.18 needs of disabled people, ascertaining, 2.31-2.35

occupation, discrimination in 9.1-9.46 barristers and advocates 9.32-9.37 contract workers 9.3-9.13 office holders 9.14-9.22 partners in firms 9.25-9.31 police officers 9.23-9.24 work experience 9.38-9.46 occupational pension schemes 10.2-10.15, Appendix A and Appendix C industrial tribunals 10.11-10.14, 13.17 dispute resolution 10.15 duty to make reasonable adjustments 10.9 trustees and managers 3.12, 10.5-10.6, 10.11-10.13, 10.15 and Appendix A offers of employment 7.32 office holders, discrimination against 3.11, 9.14-9.22

appointment to post 9.15-9.16, 9.21

duty to make reasonable adjustments 9.21

examples of office holders 9.14

overseas employment 3.9

overview of Act's provisions 3.1-3.26

partners in firms, discrimination against 3.11, 9.25-9.31, 13.8

duty to make reasonable adjustments 9.30-9.31 pension schemes see occupational pension schemes Pensions Advisory Service 10.15 and Appendix C Pensions Ombudsman 10.15 and Appendix C performance-related pay 2.26, 8.5-8.6 persons with rights under the Act 3.2-3.7, 3.11 physical features of premises, description 5.9-5.10 planning permission 12.3, 12.29 police officers, discrimination against 3.11, 9.23-9.24 positive discrimination 3.16 premises, reasonable adjustments see reasonable adjustments to premises private households, duty to make reasonable adjustments, consideration of reasonableness 5.41 procurement policy 2.26 progressive conditions, see Appendix B promotion 3.19, 5.8, 7.13, 8.1-8.2, 8.13-8.14, 9.17 public authorities 1.22-1.23 monitoring by, 2.28

qualifications bodies 3.14 and Appendix C

reasonable adjustments

advice and assistance to make adjustments 5.21

barristers 9.37 co-operation of other employees 5.22 contract workers 9.7-9.13 duty to make 5.1-5.44 effectiveness of adjustment 5.28-5.29 employee becomes disabled or disability worsens 8.15-8.17 examples of adjustments (steps) 5.18-5.23 extent of reasonable adjustment duty 5.6 failure to make 3.17, 4.24-4.26, 4.43, 5.43-5.44 financial assistance 5.31-5.39 financial considerations 5.31-5.39 health and safety concerns 5.26, 6.13 knowledge of disability 5.12-5.16 nature of employer's activities 5.40 nature of job changes 8.18 office holders 9.21 partners in firms 9.30-9.31 physical features 5.9-5.11 practicability of adjustment 5.30 private households 5.41 provisions, criteria and practices 5.8 recruitment interview 7.22-7.30 size of undertaking 5.40 work experience 9.43-9.46 when reasonable adjustment duty arises 5.3-5.17 when 'reasonable' for employer to have to make adjustments 5.24-5.42 reasonable adjustments to premises 12.1-12.33 lease prevents building being altered without consent 12.15-12.32

statutory consent 12.3-12.5 recruitment 2.46-2.48, 3.18, 7.1-7.32 advertisements 2.15, 2.46-2.48, 7.11-7.15 application forms and information 7.16-7.18 aptitude and other tests 7.25-7.26 attracting disabled applicants 2.46-2.48 blanket exclusions 7.8 disability-related questions 7.27-7.30 duty to make reasonable adjustments 5.3-5.4 guaranteed interview scheme 7.20-7.21 interviews 7.22-7.24 job description/specification 7.7-7.10 medical examinations 7.31 monitoring job applicants 2.31-2.35 offers of employment 7.32 selection, assessment and interview arrangements 7.19-7.31 short-listing for interview 7.20-7.21 redundancy, selection criteria 8.25 responsibility for the acts of others 3.23 retention of disabled employees 3.19, 8.15-8.18 rules of undertakings 13.23-13.24 self-employed 3.11 sickness absence policy, disability-related sickness 2.26 size of employer 1.15 size of undertaking, and consideration of reasonableness 5.40 social club 8.10 statutory obligations 4.40

Technical Booklet R 12.7-12.9 termination of employment 8.28-8.32

constructive dismissal 8.27 redundancy selection criteria 2.26, 8.25, 5.20 terms and conditions of service/employment 7.32, 8.4-8.6 tests in recruitment process 7.25-7.26 trade organisations 3.14 training 2.14, 2.20, 8.8 training of employees 2.24 disability awareness 2.24 understanding employer's equal opportunities policies 2.24 training services see employment services transfer of employees 8.13-8.14

variation of contracts, anti-avoidance provisions 13.22
victimisation 3.7, 3.17, 4.33-4.36, 11.9, 11.23 and Appendix A after termination of employment 8.29 dispute resolution 13.11
vocational guidance see employment services
vocational training 3.11, 9.40, 11.3

website design, Appendix C, 2.6, 2.7, 7.16, 11.18 work experience 3.11, 9.38-9.46 and Appendix A definition 9.40 reasonable adjustments 9.43-9.46



The Equality Commission for Northern Ireland can give advice and information on the Disability Discrimination Act 1995 through training, telephone and textphone advice, booklets and leaflets or we can meet with you.

For further information, please contact us at: Promotion and Education Department Equality Commission for Northern Ireland Equality House 7-9 Shaftesbury Square Belfast BT2 7DP



Telephone: 028 90 500 600

Textphone: 028 90 500 589



Fax: 028 90 248 687



Email: information@equalityni.org



Website: www.equalityni.org

ISBN 1-903941-76-8 June 2005